### FORAGING STRATEGIES AND EFFICIENCIES OF LACTATING NORTHERN AND ANTARCTIC FUR SEALS: IMPLICATIONS FOR REPRODUCTIVE SUCCESS

by

### **TIPHAINE JEANNIARD-DU-DOT**

DEUG Sciences de la Vie et de la Terre, Université de Tours, France, 1999 Licence Biologie des Organismes, Université Bordeaux 1, France, 2000 Maîtrise Biologie des Populations et Ecosystèmes, Université Bordeaux 1, France, 2001 DEA de Toxicologie Environnementale, Université de Paris V, France, 2002 M.Sc., The University of British Columbia, Canada, 2007

### A DISSERTATION SUBMITTED IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

in

### THE FACULTY OF GRADUATE AND POSTDOCTORAL STUDIES

(Zoology)

### THE UNIVERSITY OF BRITISH COLUMBIA

(Vancouver)

September 2015

© Tiphaine Jeanniard-du-Dot, 2015

### Abstract

Efficient extraction of energy from the environment is key to the survival and reproductive success of wild animals. Understanding the ratio of energy gained to energy spent of different foraging strategies (i.e., foraging efficiency) can shed light on how animals cope with environmental changes and how it affects population trajectories. I investigated how female foraging strategies during the breeding season impact the foraging efficiencies and reproductive successes of two fur seal species-one declining (NFS-northern fur seals, St. Paul Island, Alaska) and one increasing (AFS-Antarctic fur seals, Kerguelen Island, Southern Ocean). I also sought to develop new accelerometry-based methods to easily determine fine-scale energy expenditure at sea (VeDBA and flipper stroke metrics). Twenty lactating females of each species were captured and equipped with biologging tags to record GPS locations, depth and tri-axial acceleration. Energy expenditure for each foraging trip was measured using the doubly-labeled water method, and energy gained while foraging was determined from 1) diet composition (scat hard-parts and DNA) and blood stable isotope ratios; and 2) numbers of prey capture attempts (from head acceleration). Maternal investment in pups was determined from pup growth rates or from energy content of milk samples. Results showed acceleration metrics were only accurate at predicting energy expended by fur seals when time-activity budgets were taken into account (i.e., time spent performing different types of activity). Foraging strategies of AFS females resulted in efficiencies of ~3.4, with more efficient females producing bigger pups at weaning that had greater chances of survival. NFS females employed two foraging strategies with very different efficiencies (~1.4 vs ~3.0) that were associated with different foraging habitats and diet qualities. However, NFS with the more efficient strategy (3.0) undertook longer foraging trips than the other NFS (1.4) or AFS (3.4), and thus fed their pups ~20% less frequently. As a consequence, the declining NFS (unlike the increasing AFS) had to compromise between the rate of energy acquisition and the pup feeding frequency. Such reductions in energy intake and time allocated to nursing pups can ultimately lower juvenile survival, and may explain the population decline of NFS in Alaska.

## Preface

I designed the research protocols and conducted the field experiments described in Chapters 3 to 6, and performed all of the data analyses in all of the Chapters. I also prepared and wrote Chapters 2 to 6 as stand-alone manuscripts for publication. I benefited from comments and edits by co-authors Andrew Trites and Christophe Guinet (Chapters 2 to 6), John Arnould (Chapters 2, 3, 5 & 6), and Yves Cherel and Austen Thomas (Chapter 4). Laboratory measurements of oxygen and hydrogen stable isotope for the doubly-labeled water analyses (Chapters 2, 3, 5 and 6) were done at John Speakman's Isotope Laboratory in Energetics at the University of Aberdeen in Scotland. DNA-metabarcoding measures of northern fur seal scats were performed in collaboration with Austen Thomas (Chapter 4). Identification of scat hard part remains were performed by Pacific ID Inc., Victoria, BC, Canada for northern fur seal scats (Chapters 4, 5, & 6). Northern fur seal milk samples were analysed at the SGS Canada Laboratories in Burnaby, BC, Canada (Chapter 6). All the work was approved under the US NMFS permit # 14329-01, the UBC animal care permit # A10-0364 and the ethical regulations approval from the French Polar Institute (IPEV).

## **Table of contents**

Abstract	.ii
Preface	iii
Table of contents	iv
List of tables	• X
List of figuresx	iii
Lists of abbreviationsxv	vii
Acknowledgementsx	ix
Dedicationx	xi
Chapter 1: General introduction	.1
1.1 Study species and sites	. 2
1.2 Methodology	. 8
1.3 Research objectives and hypotheses	11
1.4 Thesis outline	13
Chapter 2: Accelerometers can measure total and activity-specific energy	gy
expenditure in free-ranging marine mammals only if linked to time-activi	ty
budgets	14
2.1 Summary	14
2.2 Introduction	15
2.3 Material and methods	17
2.3.1 Data collection	17
2.3.2 Daily energy expenditure – doubly-labeled water method	19
2.3.3 Diving and foraging behaviour parameters	20
2.3.4 Time-activity budget	21

2.3.5 Dynamic body acceleration	23
2.3.6 Statistical analyses	24
2.4 Results	26
2.4.1 Relationship between total energy expenditure and total DBA	27
2.4.2 Time-activity budget and activity-specific energy expenditure	29
2.4.3 Activity-specific DBA	32
2.4.4 Predicting energy expenditure at sea from activity-specific DBA and behavior	ırs33
2.5 Discussion	37
2.5.1 Dynamic body acceleration	37
2.5.2 Time-activity budgets and activity-specific energy expenditures	41
2.5.3 Conclusions	45
Chanter 3. Do flipper stroke rates and amplitudes accurately predict en	ergy
chapter 5. Do import stroke rates and amplitudes accurately predict en	
expenditure in foraging northern fur seals and Antarctic fur seals?	47
expenditure in foraging northern fur seals and Antarctic fur seals?	<b>47</b> 47
expenditure in foraging northern fur seals and Antarctic fur seals?	47 47 48
<ul> <li>expenditure in foraging northern fur seals and Antarctic fur seals?</li></ul>	47 47 48 49
<ul> <li>expenditure in foraging northern fur seals and Antarctic fur seals?</li></ul>	47 47 48 49 49
<ul> <li>expenditure in foraging northern fur seals and Antarctic fur seals?</li></ul>	47 47 48 49 49 s. 50
<ul> <li>expenditure in foraging northern fur seals and Antarctic fur seals?</li></ul>	47 47 48 49 49 s. 50 50
<ul> <li>expenditure in foraging northern fur seals and Antarctic fur seals?</li></ul>	47 47 48 49 49 s . 50 50 53
<ul> <li>expenditure in foraging northern fur seals and Antarctic fur seals?</li></ul>	47 47 48 49 49 s. 50 50 53 54
<ul> <li>expenditure in foraging northern fur seals and Antarctic fur seals?</li></ul>	47 47 48 49 s. 50 50 53 54 59
<ul> <li>expenditure in foraging northern fur seals and Antarctic fur seals?</li></ul>	47 47 48 49 s. 50 50 53 54 59 60

3.5.3 Flipper stroke cost versus cost of transportation	64
3.5.4 Comparison of flipper stroke vs VeDBA as an index of total energy spen	nt at sea 64
3.5.5 Conclusion	65
Chapter 4: Combined hard-parts and DNA analyses of scats, and stab analysis of blood from lactating females reveal two distinct diets and	le isotope foraging
strategies for northern fur seals breeding in the Bering Sea	
4.1 Summary	
4.2 Introduction	67
4.3 Material and methods	69
4.3.1 Data collection	69
4.3.2 Morphological identification from hard-parts in scats and prey size	
4.3.3 Prey species identification from DNA sequences analyses	
4.3.4 Combined results and prey associations	71
4.3.5 Stable isotopes	
4.4 Results	
4.4.1 Foraging location and behaviour	
4.4.2 Scats hard-part remains	75
4.4.3 Scat DNA analyses	75
4.4.4 Combined methods	77
4.4.5 Prey associations	
4.4.6 Stable isotopes	
4.5 Discussion	
4.5.1 DNA versus hard-parts techniques	
4.5.2 Stable isotopes and prey associations reveal two foraging strategies	

4.5.3 Conclusions	88
Chapter 5: Foraging efficiency is related to the reproductive fitnes	s of lactating
Antarctic fur seal through increased prey catch success and reduce	d time at sea
	89
5.1 Summary	89
5.2 Introduction	90
5.3 Material and methods	
5.3.1 Data collection	
5.3.2 Energy expenditure	
5.3.3 Foraging behaviours	
5.3.4 Prey capture attempts	
5.3.5 Diet estimates	97
5.3.6 Foraging efficiency	
5.3.7 Pup growth	100
5.3.8 Linking foraging efficiency of mothers and growth of pups	100
5.4 Results	101
5.4.1 Foraging behaviours	101
5.4.2 Energetic cost of foraging	101
5.4.3 Prey capture attempts	103
5.4.4 Diet and prey energy contents	104
5.4.5 Foraging efficiency and pup growth	
5.5 Discussion	108
5.5.1 Estimation of foraging efficiency (FE)	109
5.5.2 Foraging efficiency and pup growth	111

5.5.3 Effect of time-activity budget	
5.5.4 Effect of phenotypic traits	
5.5.5 Conclusions	
Chapter 6: Lactating females in a declining population	of northern fur seals
compromise either on foraging efficiency or pup feed	ing rates during the
breeding season	
6.1 Summary	
6.2 Introduction	
6.3 Material and methods	
6.3.1 Data collection	
6.3.2 Energy expenditure, foraging behaviours and prey captur	e attempts 121
6.3.3 Diet estimates	
6.3.4 Foraging efficiency	
6.3.5 Milk energy content	
6.3.6 Pup growth	
6.4 Results	
6.4.1 Foraging behaviours	
6.4.2 Energetic cost of foraging	
6.4.3 Prey capture attempts	
6.4.4 Diet and prey energy contents	
6.4.5 Foraging efficiency	
6.4.6 Milk energy content	
6.4.7 Pup growth	
6.5 Discussion	

6.5.1 Foraging efficiency varies with foraging strategy	137
6.5.2 Conflict between time and energy currencies	140
6.5.3 Maternal investment	142
6.5.4 Conclusions	145
Chapter 7: General conclusion	146
7.1 Summary of findings	147
7.2 Evaluation of research hypotheses	149
7.3 Caveats and limitations	152
7.4 Future research	156
References	160
Appendices	195

## List of tables

Table 2.1: Energy expenditure estimated for 16 lactating northern (NFS) and 17 Antarctic fur
seals (AFS) during a single loraging trip
Table 2.2: Percentage of total time at sea and average VeDBA spent in 4 types of activity for
16 lactating northern fur seal and 17 lactating Antarctic fur seal during a single foraging trip.
Table 2.3: Parameters for general linear models describing the relationship between energy
expended by fur seals per day to complete a single foraging trip (in MJ) as well as the cost of
different foraging activities (time spent foraging, transiting, performing surface movements /
grooming or resting)
Table 2.4: Best predictive models for energy expenditure (mass-corrected or not) by lactating
northern and Antarctic fur seals
Table 3.1: Morphometric, metabolic and foraging parameters for northern and Antarctic fur
seals during a single foraging trip at sea
Table 3.2: Best models explaining at-sea energy expenditure in MJ/kg of northern (n=12)
and/or Antarctic fur seals (n=13) as a function of flipper stroke count or of swimming effort
Table 4.1: Frequency of occurrence (FO) and split sampling frequency of occurrence (SSFO)
of prey in scats collected on Reef rookery, St. Paul Island during the breeding season 2011
measured from hard-part remains and from DNA analyses in scats
Table 5.1: Relative proportion (%), average prey mass (in g), prey energy density (ED in
kJ/g), energy content (in kJ) of prey groups in diets of female Antarctic fur seals breeding to
Pointe Suzanne on Kerguelen Island
Table 5.2: Measured and corrected energy expenditure (EE in MJ), measured and corrected
number or prey capture attempts (PrCA) during a foraging trip at sea for female Antarctic fur
seals (n=14)

Table 5.3: Standardized regression coefficients (SRC), minimum and maximum 95% confidence interval, bias and standard error (SE) of the sensitivity analysis on the calculated

Table A.2: Details of the best models predicting energy expenditure (in MJ or in MJ/kg). 196

Table A.5 Relative proportion (%), average prey mass (in g), prey en	ergy density (ED in
kJ/g), energy content (in kJ) of prey species in the diets of female north	ern fur seal breeding
on St. Paul, and of females breeding on Bogoslof Island	

## List of figures

Figure 2.1: Example of depth profile, acceleration signals over the 3 axes, and combined running variance for 35 min for a female northern fur seal foraging in the Bering Sea. ...... 22

Figure 3.3: Relationships between number of flipper strokes during diving or while transiting at the surface (speed  $\geq 1$  m/s) and activity-specific energy expenditure (in MJ) for northern fur seals (orange triangles, n = 16) and Antarctic fur seals (green squares n = 16). 56

Figure 4.6: Stable carbon isotope values in the blood of 20 adult northern fur seal females breeding on St. Paul Island from samples collected before versus after their foraging trip... 80

Figure 5.1. Prey capture attempts (PrCA) detected from back acceleration compared to PrCAs detected from head acceleration signals
Figure 5.2: Example of a dive profile of a female Antarctic fur seal during a foraging trip (left panel) and over 5 min (right panel)
Figure 5.3: Foraging locations of the 20 Antarctic fur seal females tracked on the Kerguelen plateau. Red dots show where the animals attempted to capture prey within the range of their foraging location
Figure 5.4: Growth of 36 individual Antarctic fur seal pups ( $n = 23$ for males and $n = 13$ for females) from birth to weaning on Pointe Suzanne colony, Kerguelen Island in the breeding season 2012
Figure 5.5: Deviation of individual Antarctic fur seal pup mass at weaning from average pup mass at weaning as a function of the deviation of foraging efficiency of individual pup's mothers from the average mothers foraging efficiency over one foraging trip during the 2012 breeding season on Kerguelen Island
Figure 5.6: Relationship between relative mass of pups at weaning (calculated as deviation from mass-specific average pup mass at weaning) and time mothers spent diving during their foraging trip at sea in h (graph A), and energy gain per min spent diving (Graph B). Time at sea was also linked to pup mass at weaning
Figure 6.1: Foraging location, dive profiles over a full foraging trip, and over 5-min of diving of lactating northern fur seals from Reef rookery on St. Paul Island foraging off the shelf ( $n = 8$ , A, C, E) or on the shelf ( $n = 12$ , B, D, F) in the Bering Sea
Figure 6.2: Time spent at sea by females during a foraging trip as a function of their body condition (mass/length in kg/cm) prior to the trip at sea (in days, $n = 19$ )
Figure 6.3: Frequency of estimated walleye pollock length (in cm) consumed by female fur seals in the Bering Sea estimated from otolith size ( $n = 173$ ) found in 98 scats collected on Reef Rookery in August 2011
Figure 6.4: Foraging trip duration of females as a function of the milk lipid content after the trip (A, $n = 5$ ) or prior to it (B, $n = 5$ )

## Lists of abbreviations

AFS	Antarctic fur seal
AIC	Akaike Information Critera
AICc	Second-order Akaike Information Critera
BM	Body Mass (kg)
BMR	Basal Metabolic Rate (MJ/d)
CI	Confidence Intervals
СОТ	Cost of Transport (J/m)
DBA	Dynamic Body Acceleration (m/s <sup>2</sup> )
DD	Daily-Diary tag
DLW	Doubly-Labeled Water
DNA	Deoxyribonucleic Acid
EC	Energy Content (kJ)
ED	Energy Density (kJ/g)
EDTA	Ethylenediaminetetraacetic Acid
EE	Energy Expenditure (MJ)
ENSO	El Nino Southern Oscillation
ESS	Error Sum of Squares
FE	Foraging Efficiency
FL	Fish Length (cm)
FO	Frequency of Occurrence
GCDC	Gulf Coast Data Concept
gls	Generalized Least Square
GPS	Global Positioning System
hclust	Hierarchical Cluster Analysis
ID	Identity
IPEV	Institut Paul Emile Victor
LIENSs	Laboratoire du Littoral Environnement et Sociétés
lm	Linear Model
lme	Linear Mixed-Effects model

LRL	Lower beak Rostrum Length
MR	Metabolic Rate (MJ/d)
nlme	Non Linear Mixed-Effects models
nls	Non-Linear Least Squares
NFS	Northern fur seal
ODBA	Overall Dynamic Body Acceleration (m/s <sup>2</sup> )
OL	Otolith Length (mm)
PrCA	Prey Capture Attempt
PCR	Polymerase Chain Reaction
PFZ	Polar Front Zone
RMA	Ranged Major Axis
RQ	Respiratory Quotient
RSS	Regression Sum of Squares
SD	Standard Deviation
SE	Standard Error
SRC	Standardized Regression Coefficients
SSFO	Split-Sample Frequency of Occurrence
UBC	University of British Columbia
US NMFS	US National Marine Fisheries Services
VeDBA	Vectorial Dynamic Body Acceleration (m/s <sup>2</sup> )
X axis	Surge
$X_{dyn}$ , $Y_{dyn}$ and $Z_{dyn}$	Dynamic acceleration on axes X, Y and Z $(\text{m/s}^2)$
Y axis	Sway
Z axis	Heave

### Acknowledgements

I am very thankful to my supervisors Andrew Trites and Christophe Guinet for their guidance and support during the course of this PhD. I have learned how to become a skilled and independent researcher from these amazing professionals, and will always be grateful for the opportunity they provided to work on these 2 very special species of fur seals. I am also grateful to Bill Milsom and Scott Hinch, my research committee members, for their ever so helpful comments and suggestions on my research, and for providing a different 'non-marine mammal' perspective on the topic. My many thanks are extended to John Arnould, who has been so kind as to share both his great field and academic expertise with me, to welcome me in his lab for 6 months and with whom I shared many heated and highly entertaining scientific debates over the years. I am also grateful to Yves Cherel for sharing his extended knowledge with me and for providing the opportunity to measure stable isotopes in my animals and reveal a key piece in the fur seal puzzle.

My research involved months in the field in remote locations, and I could have never obtained such a dataset without the help of many field mates: John Arnould, Michelle Barbieri, Alistair Baylis and Rachel Orben all incredibly skilled in Alaska; and Nory El-Ksaby, Pacha of Kerguelen and field master extraordinaire, Jade Vacquie-Garcia, Laetitia Pons (who went out of her medical comfort zone to contribute to this project both in the field and in the lab!). Samantha Patrick, Aurelien Prudor, Pascal Grijol, Miguel and Tony 'Garage' among many other volunteers of the Mission 62 on Kerguelen Island. They were all amazing adventurers and dedicated individuals who spend months/years of their lives in hostile yet stunning territories to study wildlife in extreme environments. Special thanks to the Tafelzout tribe of Pointe Suzanne of course.

Invaluable logistic help was provided by Bill Briggs and NMFS on St. Paul and by Yann Le Meur and Nina Marchand's team from IPEV on Kerguelen Island. Financial support from NSERC, IPEV, CNRS, NPRB, UBC and the Zoology Department allowed me to take on this great project to the fullest. John Arnould and Yves Cherel provided additional support. Thanks to Agnes Lacombe for showing me the ropes of teaching at the university level and sharing her endless and communicative enthusiasm to infuse scientific knowledge into young minds.

My PhD experience over the years was greatly enriched from interacting with numerous amazing and inspirational people, scientists and students on 3 continents, all of whom I am fortunate to now and forever call my dear friends. Thanks to Chiara Piroddi my stellar twin, Rowenna Gryba, Maud Berlincourt, Susana Cardenas, Frances Robertson, Pamela Allen, and Samantha Patrick my pillars, and all the other amazing women who will undoubtedly shape the future of the research in marine biology for the better; and to the equally inspiring colleagues and friends Austen Thomas, Andrew Hoskins, Guillaume Bessigneul, Hassen Allegue for their help, support, presence and friendship throughout this experience. I am not forgetting the many friends Franck, Maxime, Fabrice, Nathan, Johan, Anne, Gabbie, Alejandro, Mariana, Sarah, Hassen and more with whom I shared gems of moments that made my life exciting and filled it with brunches, treks, yoga challenges, volleyball leagues, beach barbecues, camping trips, Palme d'Or ceremonies, holidays and much more. Obviously I want to send big thanks to my far-away family. Despite their almost exclusively literary and artistic background, they embraced my project and my way of life and got excited by fur seals (with the ears!) almost as much as I am. They supported me from the far, and became invested into this 'foreign' and 'exotic' field for them. I could not have done it without any of them and their unconditional emotional and intellectual support throughout the ups and downs, and for reminding me what is important in life when I lost sight of it.

To Erell and Vladik, the next generation

### **Chapter 1: General introduction**

The effects of natural and anthropogenic environmental changes occurring in marine ecosystems are usually most noticeable on species occupying the top trophic levels, such as marine mammals and seabirds. These top predators are considered sentinels of the health of ecosystems and are often used as environmental indicators within the context of international marine conservation treaties (Boyd et al., 2006). Environmental changes linked to global warming have been occurring at a greater rate at high latitudes, affecting Arctic and Antarctic ecosystems more drastically than temperate or tropical ones (Watson et al., 1998; IPCC, 2007). Rapid population decline of marine mammals and seabirds as well as lower survival rates of new-borns and juveniles, and frequent reproductive failures have been observed in the Bering Sea and Arctic regions, concomitantly to changes in prey distribution and availability (Kock, 1992; Kitaysky and Golubova, 2000; Trites and Donnelly, 2003; Weimerskirch et al., 2003b; Jodice et al., 2006; Kitaysky et al., 2006). In the Southern hemisphere, changes in population dynamics of marine birds or mammals are likely linked to temperatures, sea-ice coverage or position of the Polar Frontal Zone (PFZ) that have varied in recent years (Barbraud and Weimerskirch, 2001; Siniff et al., 2008). Documenting such changes is important for understanding the processes that underlie the changes, and for developing sound and efficient management strategies for populations of animals in these marine ecosystems.

Survival and reproductive success of individuals, and thus the health of their populations, depends on the efficiency with which individuals extract energy from their environment per unit of time (MacArthur and Pianka, 1966). The more efficient they are, the more energy and time they can allocate between the competing needs of growth, reproduction, and body maintenance (Boggs, 1992). Animals have some flexibility to accommodate changes in environmental conditions. However, if energy intakes are ultimately sub-optimal, decisions regarding how energy is allocated will likely be modified. In long-lived animals, self-maintenance has priority over reproduction because fitness and lifetime reproductive success depend on longevity (Bouten et al., 1994). Hence, fluctuating energy resources are expected to first affect breeding performances before the survival of adults is compromised, but the extent to which reproduction is impaired by food availability

partly depends on the flexibility with which animals can acquire energy (foraging efficiency) and allocate it (physiological plasticity).

Understanding the responses of populations to environmental changes requires linking ecological factors with fitness, survival, and reproduction. One promising approach to study this is through bioenergetics that considers time budgets and energy balances of animals within an environmental context. This approach is ideal because it integrates biotic and abiotic influences, as well as external and internal factors that are easily translated from individual animals into population or community outcomes (Kooijman, 2000; Kooijman et al., 2004; Fischer et al., 2009; Sousa et al., 2010). In addition, individual animals are the unit of adaptability to changes in habitat conditions and are the drivers of population changes. Consequently, understanding the bioenergetic dynamics of individuals is needed to infer the potential success of an individual and ultimately the collective success of the population.

### **1.1 Study species and sites**

As top predators breeding in high-latitude marine ecosystems with similar features, northern fur seals (*Callorhinus ursinus*) and Antarctic fur seals (*Arctocephalus gazella*) are ideal candidates to monitor and investigate the mechanisms and linkages between the environment, individual behaviours, energetics, fitness, and population dynamics. These are two closely related species of otariids that live at 50-60° latitude north in the Bering Sea for the northern fur seals and at 50-60° latitude south in the Southern Ocean for the Antarctic fur seals. They share physical, behavioural and ecological features, and have similar natural histories, maternal behaviours, and reproductive strategies (Gentry and Kooyman, 1986). They are the two northernmost and southernmost otariid species. As such, they experience the extreme seasonal variations in weather conditions that occur at high latitudes—which may explain why they are the only two species of otariids that have a short 4-month lactation period during the Austral and Boreal summers. Pups of both species wean abruptly, and are migratory and pelagic for the next 20–32 months of their lives.

The reproductive strategies of northern fur seals and Antarctic fur seals are energetically expensive, and rely exclusively on the mother's capacity to care for her pup (Costa, 1993). Females nurse their offspring on land with alternate trips at sea to replenish their reserves. Lactation is one of the most demanding activities in their life cycles with energy demands increasing by 30% compared to non-breeders (Gentry and Kooyman, 1986; Arnould, 1997). Consequently, lactating females are likely to be more affected than other adults by sub-optimal foraging conditions. Both species are opportunistic feeders and can adapt to some extent to environmental changes (Guinet et al., 2001; Lea et al., 2002b; Reid et al., 2006; Zeppelin and Ream, 2006; Lea et al., 2008; Zeppelin and Orr, 2010). However, lactating females are limited in how far and for how long they can search for prey due to the fasting constraints of their young.



Figure 1.1: Picture of Northern fur seal male, females and pup during the breeding season 2011 on St. Paul Island (indicated by the yellow star and detailed on the close up map), part of the Pribilof Islands located in the Bering Sea, Alaska.

Pups wait on the rookeries for their mother's return, and rely on milk to grow and accumulate energy reserves. At weaning, pups of both species leave the breeding grounds and return several years later. Survival of pups and young of the year are linked to their growth rates and mass at weaning (Baker and Fowler, 1992; Boltnev et al., 1998; Hall et al., 2001; Lea et al., 2009). Weaned fur seals heavily rely on the reserves acquired during lactation to support their physiological needs at sea when they are inexperienced and ontogenetically immature—and can lose up to half their weaning body mass during their first year at sea (Trites and Bigg, 1996). Mortality rates are up to 60% in the first 2 years of life for northern fur seals (Trites, 1989; Gentry, 2002), but are hypothesized to have been lower prior to the population decline (Trites, 1989). In contrast, early mortality rates of Antarctic fur seals are around 30% (although it has never been precisely calculated from birth to first return on land, Boyd et al., 1995a; Wickens and York, 1997). Mortality rates of both species drop rapidly after the first 2 years at sea—the most critical period in their life histories (York, 1985). The nursing period is thus critical for mothers (via increased energy requirements) and pups (for their future survival).

Northern fur seals and Antarctic fur seals were heavily hunted in the 18<sup>th</sup> and 19<sup>th</sup> centuries, to the point that Antarctic fur seals were virtually extinct by the early 20<sup>th</sup> century. However, the northern fur seals breeding on the Pribilof Islands recovered once protected, and there were ~ 1.5 M animals from mid-1980s to mid-1990s when the decline resumed. Since then, pup production on St. Paul Island (the largest of the Pribilof Islands) has declined at ~ 6 % per year (Towell et al., 2006). They were listed as depleted under the Marine Mammal Protection Act in 1988 because population levels had declined to less than 50% of levels observed in the late 1950s, with no compelling evidence that carrying capacity had changed (NMFS, 1993). Although pre-sealing numbers are unknown, the Antarctic fur seal population has recovered rapidly since the end of their exploitation in the 1960s increasing by 6 to 14 % per year at South Georgia since 1990 (Arnould, 2002; Aurioles and Trillmich, 2008). Recovery rate has not been systematically assessed on Kerguelen Island, but pup production at the biggest colony is currently estimated at 6500 per year and increasing (Figure 2). Both species are subject to protection and conservation plans (Convention for the Conservation of Antarctic Seals, 1972; Pribilof Islands Area Habitat Conservation Zone, NMFS, 1993).



Figure 1.2: Picture of Antarctic fur seal male, females and pup during the breeding season 2011-2012 on Kerguelen Island (indicated by the yellow star and detailed on the close up map) located in the Sub-Antarctic Southern Ocean.

I studied two populations of fur seals located on St. Paul Island (northern fur seals, Figure 1.1) and Kerguelen Island (Antarctic fur seals, Figure 1.2). Both islands are located at similar high latitudes in the Arctic and Antarctic regions (49°S for Kerguelen Island and 56°N for St. Paul Island). The Pribilof Islands have an Arctic maritime climate with constant wind speed averaging 25–50 km/h year round. In contrast, the Kerguelen Archipelago is located in the "Furious Fifties" with winds blowing from the west at a remarkable average speed of 35 km/h and regularly reach 150 km/h. St. Paul and the Kerguelen Islands are both close to oceanographic discontinuities (*i.e.*, surrounded by relatively shallow waters with oceanic currents bringing up high nutrient concentrations and productivity in the close

vicinity) that are preferred foraging zones for top predators (Boyd and Arnbom, 1991; Georges et al., 2000).

The Kerguelen Archipelago is located on the Kerguelen plateau, with depths from 500 to 2000 m. The state of the sea is rough with wave heights of 12 - 15m being common, but the sea around Kerguelen is ice free. The Antarctic Polar Front where Antarctic surface waters moving northward sink below sub-Antarctic waters passes in the vicinity of the Kerguelen Plateau where it is responsible for mesoscale meandering, eddies, and ring formations. Upwelling from the Antarctic mixes with the warmer waters of the Indian Ocean stimulating oceanic primary production.

Like the Kerguelen Islands, St. Paul is located on the outer domain of the shallow continental shelf with depths < 200 m, 90 km North West of the Bering Sea shelf-break beyond which water depths are > 1000 m. The waters surrounding the island consist of discrete domains divided by oceanographic fronts. Coastal waters are generally ice free year-round around St. Paul, and marine currents present a steady clockwise flow, stronger south of the island and weaker to the east (Kowalik and Stabeno, 1999). These currents stir the relatively shallow, nutrient-rich waters that form the basis for the high productivity and biodiversity present. As around the Kerguelen Islands, the waters surrounding St. Paul Island have elevated productivity.

Major environmental changes have occurred in the Bering Sea and over the Kerguelen plateau in the last three decades. In the Eastern Bering Sea, water temperatures have risen by 2 - 3°C, and the ice-associated, cold-water bodies have retreated north. This has resulted in lower nutrient input in the water columns, and changes in primary production and dynamics. Other changes associated with the rise in water temperatures are species distributions, both geographically and vertically to the water column, prey-predator trophic interactions, and large scale ecosystem dynamics (Anderson and Piatt, 1999; Napp et al., 2000; Overland and Stabeno, 2004). Major threats to the recovery of northern fur seals are due to heavy fishing of adult walleye pollock in the Bering Sea, or to potential impacts of climate change on the dynamics of the ecosystem and of the pelagic prey (Mace and Balmford, 2000). Major environmental changes starting in the mid-1970s altered the distribution and relative abundances of different species of fish in the Bering Sea (Anderson

et al., 1997; Anderson and Piatt, 1999) resulting in a switch from mainly high-energy prey to low-energy prey (Trites et al., 2007; Sinclair et al., 2008). Although fur seals in the central Bering Sea are declining, a small colony recently established on Bogoslof Island in the southern Bering Sea within the deep basin of the Bering Sea is increasing (Towell and Ream, 2011). This difference in dynamics and habitats suggests that local foraging conditions likely lie at the base of these changes (Trites et al., 2015). Reef rookery on St. Paul Island is one of the largest breeding sites for northern fur seals, and individual females are known to forage in different locations both on and off the Bering Sea shelf (Robson et al., 2004; Call et al., 2008), which might affect their foraging behaviours and efficiencies.

Although the Antarctic fur seal population on Kerguelen Island is currently healthy, they might face similar threats in the near future, as fisheries increase around the breeding areas and water temperatures increase due to climate change. These changes might result in the Antarctic Polar Front retreating southward, which would affect the availability of the main prey source for Antarctic fur seals (Hanchet et al., 2003; Lea et al., 2006; Learmonth et al., 2006). Significantly higher rates of reproductive failure have occurred in Antarctic fur seal colonies during warmer years (Lea et al., 2006). Antarctic fur seals are also thought to have lost considerable genetic diversity due to the historical population bottleneck (Wynen et al., 2000) and may now be at increased risk from disease outbreaks and environmental change. Under both scenarios, animals might have to employ new strategies to balance the compromise between the potential to gain energy against the actual time and energy spent foraging.

Differences in population trajectories provide an opportunity to compare how two closely related, and biologically and ecologically similar species respond and adapt to environmental changes, despite the inherent challenges of two-species comparisons. It also provides an opportunity to understand the mechanisms that underlie these different population responses.

#### 1.2 Methodology

The rate with which climate is changing in the northern and southern hemisphere is unlikely to slow anytime soon. However, there is little understanding about how climate change will affect the physiology, behaviour and energetics of fur seals and other species—and how this will ultimately affect the dynamics of their populations. Animals modulate their behaviours over varying time scales, which need to be quantified and related to energetics to better understand the choices they will have to make. Consequently, understanding how animals respond to environmental changes requires information about their behaviours at sea and how it affects their foraging efficiency (*i.e.*, the energetic gains versus the energetic costs of such foraging strategies).

Traditionally, radio and satellite telemetry methods have been used to determine movement patterns and at-sea locations of fur seals. However, the resolution of the data has generally been too coarse to study fine scale behaviours of animals. Recently, the development of new remote sensors such accelerometers and magnetometers provide new insights into tri-dimensional body movements and behaviours. The resolution of the data is unprecedented and provides considerable information on animal behaviours at a fine scale never attained before. Combined with more traditional loggers such as time-depth recorders and satellite loggers, it is now possible to reconstruct the route of animals in 3 dimensions, measure distances and swimming speed, as well as allow segregating foraging trip into different types of activities such as transiting, grooming, resting and diving/feeding behaviours (*i.e.*, in time-activity budgets).

Types of behaviours displayed while foraging are associated with different energy costs, so defining and quantifying the behaviours that make up activity budgets is an important step in understanding the energetics of free-ranging marine mammals. Methods to measure energy expended by animals while foraging include heart rate monitors and doubly-labeled water (Lifson and McClintock, 1966; Butler et al., 1992; Butler, 1993; Speakman, 1997; Froget et al., 2004; Wilson et al., 2006; Young et al., 2011). However, these techniques can be invasive, very costly, and difficult to obtain in large free-ranging marine animals—especially with heart rate monitors (Nagy, 1980; Thorarensen et al., 1996; Ward et

al., 2002; Butler et al., 2004; Dalton et al., 2014b). In addition, the DLW method only provides an estimate of energy expenditure over an entire feeding trip. Consequently, there is a need to develop methods to determine energy expenditure at the activity level, which remains to be achieved in marine mammals.

Overall Dynamic Body Acceleration (ODBA) and Vectorial Dynamic Body Acceleration (VeDBA) are 3D body acceleration metrics that have been hypothesized to correlate with energy expenditure in various animal taxa (Wilson et al., 2006; Halsey et al., 2009a; Halsey et al., 2009b; Qasem et al., 2012). This method is based on the assumption that most of the energy spent while foraging is due to the cost of locomotion, and that there is a direct link between the energy needed to contract muscles (how muscular tissue converts metabolic energy to mechanical work) and the resulting movement of the body during locomotion from acceleration (the rate at which work is performed) (Gleiss et al., 2011). As acceleration can be recorded at a sub-second rate, it has the potential to resolve energy expenditure with fine temporal resolution, making it the only method able to determine the cost of short-lived behaviours. However relationships between acceleration and energy expenditure vary by species and by type of activity, and need to be individually calibrated for each case, and has seldom been tested on free-ranging animals (Halsey et al., 2008; Elliott et al., 2013; Wright et al., 2014).

In addition to affecting energy expenditure, foraging strategies impact foraging success of wild animals (*i.e.*, their energetic gains while foraging). Foraging success is inherently difficult to assess in elusive animals. It has traditionally been assessed from body mass or composition change during foraging trips. However, this metric does not consider that animals can lose weight even at satiation (ref) depending on types of prey they consume, or differences in energy allocation between conflicting physiological functions. Understanding energetic gains from foraging requires an understanding on both the diet composition of individuals and their success at capturing prey.

Assessing diets is thus important for understanding the balance between energy input from food, and the energy spent foraging. Diets of free-ranging animals are traditionally estimated by identifying the undigested hard-part remains of prey found in stomach contents, feces (scats) or regurgitates (Perez and Bigg, 1981; Perez and Bigg, 1986; Antonelis et al.,

1990). Taxon, size and relative number of prey ingested can be determined from hard-parts (Tollit et al., 2003), but the method is biased by differential digestion or retention rates of prey parts depending on species and sizes of prey consumed, and on species, sex, activity level, GI tract morphology and meal size of predator that lead to under- or over-estimation of prey occurrence (or size of prey consumed) in the diets (Bigg and Fawcett, 1985; Jobling and Breiby, 1986; Harvey and Antonelis, 1994; Tollit et al., 1997; Bowen, 2000; Arim and Naya, 2003; Gudmundson et al., 2006). More recently, molecular-based tools, such as DNA meta-barcoding, have been developed to overcome some of the limitations of hard-part analyses (Deagle et al., 2005; Casper et al., 2007a; Deagle and Tollit, 2007; Deagle et al., 2009; Deagle et al., 2010; Cristescu, 2014; Thomas et al., 2014), and allow species of organisms to be identified within a scat sample at a finer resolution than hard parts. However, this new DNA approach has its own inherent technical and interpretation biases (Casper et al., 2007a; Deagle et al., 2010; Thomas et al., 2014).

Both the hard-part and DNA-based methods provide diet composition only at the population level and only represent what animals consumed in the last few days (~ 48h, Orr and Harvey, 2001; Staniland, 2002; Deagle et al., 2005; Casper et al., 2007a). Carbon and nitrogen stable isotope concentrations in blood, hair and bone synthesize dietary have the advantages of providing information of the trophic level of the consumer and characteristics of its foraging habitat over longer timescales and at the individual level (Wada et al., 1991; Hobson and Welch, 1992). Consequently, scat-based methods and stable isotopes provide different but complementary insights into the diet and foraging habits of animals. Using a multi-methodological approach by pairing an individual's foraging strategy information from telemetry data with respective stable isotope analyses, and general scat based approaches can provide a more complete understanding of foraging strategies and diets, and disentangle spatial, trophic and energetic differences between individuals (Newsome et al., 2010).

Quantitative estimates of foraging success, or energetic gain, also requires determining numbers of prey captured and consumed by individuals. This was previously assessed from changes in body water pool (Costa, 1993), or with stomach temperature sensors (Grémillet and Plös, 1994; Kuhn and Costa, 2006), both of which present methodological and technical challenges in wild marine mammals. More recently, however,

high resolution tri-axial accelerometers have been used to measure numbers of prey capture attempts in free-ranging marine predators (Skinner et al., 2009; Suzuki et al., 2009; Viviant et al., 2010; Watanabe and Takahashi, 2013; Ydesen et al., 2014) from detection of head strikes during a capture attempt. Quantitative estimates of foraging energetic gain (from diet composition and number of prey capture attempts) paired with foraging costs and fine scale behaviours and movement at sea can thus provide an indication of how foraging strategies affect foraging efficiencies of animals within a specific habitat, which ultimately impact the energetic balance and fitness as individuals.

### **1.3** Research objectives and hypotheses

Given the on-going environmental changes occurring at a large scale in marine environments, there is a need to understand how animals will behaviourally and energetically respond to rapid shifts in their habitats, and how this will affect their capacities to survive and reproduce. Fortunately, recent technological and methodological advancements now allow multi-disciplinary approaches to be applied to individual animals to understand how collective responses can impact population trajectories. The main goal of my study was consequently to investigate how foraging strategies affect foraging efficiency of individual female fur seals during their breeding season, whether changes in their foraging efficiencies affected their maternal investment and reproductive success, and relate this back to environmental conditions. I also sought to develop new methods to easily determine finescale energy expenditure of fur seals at sea based on accelerometry.

My general hypothesis was that, despite the capacity to buffer environmental changes around their breeding grounds, female fur seals from the declining northern population would be expending greater foraging effort for a lower foraging success (due to lower prey accessibility and/or energetic quality) compared to females from the increasing population. They would therefore have less energy and time to allocate to reproduction, and would produce pups with lower growth rates. As mass at weaning is critical for survival during their first year at sea, this would lead to lower recruitment in the declining population. I also hypothesized that acceleration would be useful to determine energy expenditure when animals are under water, but not when they are at the surface given the conditions of the marine environment and the limitations of accelerometry for uncontrolled studies in the wild (Halsey et al., 2011b).

To test these hypotheses, I investigated fine scale foraging behaviour of individuals using state-of-the-art biologging devices. I developed and tested methods and tools to effectively determine energy expenditure at sea and activity-specific metabolic rates that can be used in future energetic studies. I also investigated new methods to estimate diet composition of individuals and the rate with which fur seals attempt to capture prey, which together provide measures of energy gained while foraging. Finally, I combined morphometric, metabolic and isotopic data with location, depth, acceleration and dietary data to determine the linkages between foraging strategies, foraging efficiencies, and the capacity to invest in reproduction within and between fur seal species. This was complemented by data collected on pups during the same breeding season.

Overall, I investigated the foraging ecology of northern and Antarctic fur seals from an ecological-energetics perspective. Using such robust methods based on energetic-flows and thermodynamic laws to compare two similar species with opposite population trajectories provided an opportunity to study the mechanistic aspect of individual plasticity and their responses to their environment. It also adds an essential and yet often underinvestigated individual-based bioenergetic piece to the puzzle by linking individual fitness to environmental conditions, and ultimately the response of populations to environmental changes. Multi-disciplinary studies integrating metabolic, energetic, biologging and modelling methods also provide greater clarity for explaining observed variations in foraging efficiency within and between species in a wider physiological and ecological context. My study was thus designed to yield results that could be easily incorporated within individualbased models and extended to a broad array of questions related to the impacts of environmental changes on the fitness and demography of animals, as well as the impacts of fur seals on prey stocks (and *vice versa*).

#### **1.4** Thesis outline

My thesis is structured as follows. Chapter 1 provides a general introduction to this study, background information, research questions and objectives, and working hypotheses. The subsequent five chapters consist of studies independently written for publication in peerreviewed journals. Chapters 2 and 3 investigate whether acceleration data can be used as a reliable indicator of energy expenditure or metabolic rates of marine mammals at sea, with Chapter 2 focusing on Dynamic Body Acceleration, and Chapter 3 on flipper stroke metrics. Chapter 4 combines scat-based methods, DNA-based methods and isotopic methods to determine the diet of northern fur seals, which is needed to quantitatively measure energy gained while foraging as a function of foraging strategy (diet and stable isotopes of Antarctic fur seals were less thoroughly investigated and are mentioned as a sidenote). Chapters 5 and 6 build on the results of the previous chapters and explore how differences in foraging strategies affect foraging efficiencies and individual fitness of mothers via maternal investment in reproduction. Chapter 5 focuses on Antarctic fur seals and empirically tests the relationship between individual foraging behaviours, foraging efficiencies and pup growth, while Chapter 6 focuses on the different strategies observed within the northern fur seal population, and how this affects foraging efficiencies and indirectly reproductive success of individuals through trade-offs between the gain and expenditure of time and energy. Chapter 6 also compares the foraging strategies of northern and Antarctic fur seals, and discusses whether they provide clues into why their reproductive success and population trajectories differ. Finally, Chapter 7 provides a summary of the major findings and associated biases, and discusses future research directions.

# Chapter 2: Accelerometers can measure total and activityspecific energy expenditure in free-ranging marine mammals only if linked to time-activity budgets

### 2.1 Summary

Energy expenditure is an important component of foraging ecology, but is extremely difficult to estimate for free-ranging animals, especially marine mammals. I investigated whether vectorial dynamic body acceleration (VeDBA) could accurately predict the energy expended by northern fur seals (*Callorhinus ursinus*) and Antarctic fur seals (*Arctocephalus gazella*) during single foraging trips or whether other dive or time-distance foraging parameters were better predictors. I also determined how well VeDBA could predict the energetic cost of the different foraging activities that made up each foraging trip (*i.e.*, diving, transiting, surface activity, and resting). To do so, I equipped 20 lactating females of each species with GPS, dive-behaviour data loggers and tri-axial accelerometers and simultaneously obtained estimates of field metabolic rates for each of 13 Antarctic fur seal and 12 northern fur seal females using the doubly-labeled water method. VeDBA was derived from tri-axial acceleration, and at-sea activities were inferred using dive depth, tri-axial acceleration and traveling speed to differentiate diving, transiting, resting and surface activities. Activityspecific metabolic rates were estimated using linear regression with individual time-activity budgets as independent variables and DLW measurements as the response variable. I found that VeDBA did not accurately predict the total energy expended by the fur seals to complete their full foraging trips ( $R^2 = 0.36$ ). However, the accuracy of VeDBA increased significantly when I partitioned their foraging trips by activity, and used activity-specific VeDBA to estimate activity-specific energy expenditures ( $R^2 > 0.85$ ). I was also able to accurately determine total energy expenditure at sea from activity-specific VeDBA paired with time activity budgets ( $R^2 = 0.70$ ). My study confirms that acceleration, in combination with other behavioural knowledge, is a promising way to estimate energy expenditures of marine mammals, but shows that it needs to be based on the different activities that make up foraging trips and the time allocated to each of them rather than being derived as a single measure of VeDBA applied to entire foraging trips. This activity-based method provides a cost-effective means to accurately calculate energy expenditures of fur seals using acceleration and time-activity budgets.

#### 2.2 Introduction

Predators constantly make decisions on where to hunt, what to hunt, and for how long to hunt that collectively affects the efficiency with which they obtain energy and minimize foraging costs. It is this foraging efficiency, or the energy gain/cost ratio of foraging, that drives many aspects of the physiology, biology, and ecology of wild animals—which in turn affects their health, reproduction and survival. It is important to accurately estimate foraging costs and benefits to understand and predict survival and reproductive success at the individual and population levels (Boyd, 2002), or to calculate food requirements and understand predator-prey interactions (Lavigne et al., 1982; Winship et al., 2002; Halsey and White, 2010).

Heart rate monitors, accelerometers, and doubly labelled water have all been used to measure energy expenditure in vertebrates (Lifson and McClintock, 1966; Butler et al., 1992; Butler, 1993; Speakman, 1997; Froget et al., 2004; Wilson et al., 2006; Young et al., 2011). However, heart rates and DLW measurements can be invasive, very costly, have their own biological limitations and are often times impractical for large wild animals (Nagy, 1980; Thorarensen et al., 1996; Ward et al., 2002; Butler et al., 2004; Dalton et al., 2014b). They are difficult also to obtain for large sample sizes and across different temporal scales that are required in ecological studies if we are to better understand links between animals and changes in their environment. Accelerometry has the potential to provide valuable information over days, weeks or months at sea, and so simple measures of body movement (e.g., acceleration) are increasingly being used to estimate energy expended by animals. They however need to be truly tested in the field under natural meaningful conditions.

Overall Dynamic Body Acceleration (ODBA) and Vectorial Dynamic Body Acceleration (VeDBA) are 2 very similar metrics based on tri-axial acceleration of the body that can be linked to energy expenditure (Wilson et al., 2006; Halsey et al., 2009a; Halsey et
al., 2009b; Qasem et al., 2012). ODBA is the sum of the absolute values of the dynamic acceleration of the 3 body axis while VeDBA is the vectorial product of these same data. These measures of acceleration are based on the assumption that most of the energy spent while foraging is due to the cost of locomotion, and that there is a direct link between the energy needed to contract muscles and the resulting movement of the body during locomotion (Gleiss et al., 2011). ODBA and VeDBA have been tested and calibrated on various taxa, both marine and terrestrial, endotherm and ectotherm, during different activities (walking, flying, swimming etc., Fahlman et al., 2008; Halsey et al., 2008; Gleiss et al., 2010; Gomez-Laich et al., 2011; Halsey et al., 2011a). They appear to have acceptable accuracy for determining energy expenditure, but relationships between acceleration and energy expenditure vary by species and by type of activity, and need to be individually calibrated for each case (Halsey et al., 2008; Elliott et al., 2013; Wright et al., 2014).

Establishing the relationship between ODBA and energy expenditure is particularly difficult for air-breathing divers due to a possible uncoupling of ODBA and gas exchange (Halsey et al., 2011b). Differences in resistance between air and water may also create different relationships between acceleration and energy expenditure and there may be effects of winds/waves at surface as well as water currents on acceleration that are not reflected in energy expenditure (Halsey et al., 2011b). Similarly, the effects of buoyancy, gliding, and other physiological functions (*i.e.*,thermoregulation, digestion etc.) that contribute to energy expenditure, but not to acceleration, may uncouple potential relationships between acceleration and energy expenditure (Gleiss et al., 2011; Halsey et al., 2011b).

Most validation and calibration studies of ODBA and VeDBA have been conducted in controlled environments over short periods, which might buffer the above limitations. For example, ODBA correlates with energy expenditure of semi-captive Steller sea lions (*Eumetopias jubatus*) trained to dive at sea (although with an  $R^2$  of 0.47, Fahlman et al., 2008), but does not correlate with the daily metabolic rate and partial dynamic body acceleration of captive northern fur seals over a 5-day period (Dalton et al. (2014b). This suggests that the predictive power of ODBA may decrease with recording time over days and weeks increases due to animals engaging in a wider range of behaviours or experiencing greater variability in environmental conditions. This may mean that ODBA and VeDBA are best applied to individual activities, rather than to the sum of all activities over extended periods.

Defining and quantifying the behaviours that make up activity budgets appear to be an important step in understanding the energetics of free-ranging marine mammals. Studies have attempted to determine time-activity budgets using a mix of acceleration, geolocation, altitude and depth data to visually discriminate behaviours (Yoda et al., 2001; Gomez-Laich et al., 2008; Insley, 2008), or have used supervised or unsupervised classification techniques such as Kmean clustering techniques (Sakamoto et al., 2009), K-nearest neighbour algorithms (Bidder et al., 2014) or decision-tree classifications (Nathan et al., 2012). Activities can be linked to specific energy expenditures within a global framework (Elliott et al., 2013; Gomez-Laich et al., 2013; Wright et al., 2014), but appear to be highly species-, environment- and activity-specific.

My primary goal was to determine whether behavioural or acceleration-based parameters could accurately predict the energy expended by 2 species of fur seals (northern and Antarctic fur seals) during individual foraging trips. Therefore, I tested whether ODBA or VeDBA alone yielded indices of energy expenditure that were comparable to simultaneous doubly-labelled water measurements of field metabolism for each species of fur seal. A secondary goal was to see whether better estimates of energy expenditure could be obtained by dividing the foraging trips into behavioural components (*i.e.*,diving, transiting, surface activity and resting). I also determined how well the indices performed at estimating the energy expenditure of the different behavioral activities that made up each foraging trip. Finally, I attempted to derive better predictions of energy expenditure from various acceleration- and behaviour-based parameters using multi-factorial models.

# 2.3 Material and methods

#### 2.3.1 Data collection

Data were collected from 20 lactating northern fur seals at the Reef rookery on St. Paul Island (Bering Sea, 57°6'N - 170°17'W) during the breeding season from Aug-Sep 2011, and from 20 lactating Antarctic fur seal females at Pointe Suzanne, Kerguelen Island (Southern Ocean, 49°26'S - 70°26'E) during the breeding season from Jan-Feb 2012. Data were collected under the US NMFS permit # 14329-01, the UBC animal care permit # A10-0364 and the ethical regulations approval from the French Polar Institute (IPEV).

All females were captured using a hoop net and were mature adults with a confirmed suckling pup. The females were carried in the hoop nets a short distance to a restraint board where they were anaesthetized with isoflurane gas. Standard morphometric measurements of length and axial girth were made to the nearest 0.5 cm, and mass was recorded using a scale at  $\pm 0.2$  kg.

Measurements of daily energy expenditure (EE, kJ/day) were obtained using the Doubly-Labeled Water (DLW) method (Lifson and McClintock, 1966; Butler et al., 2004). This method has been previously validated by comparison to indirect calorimetry in a range of small mammals (Speakman and Krol, 2005), as well as on seals (Sparling et al., 2008). I took a first blood sample (3 to 8 ml) by veinipuncture of an inter-digital vein in a hind-flipper to determine baseline levels of <sup>2</sup>H and <sup>18</sup>O for each individual (Speakman and Racey, 1987 – method D). A known mass pulse-dose of DLW (622272 ppm <sup>18</sup>O, 384645 ppm <sup>2</sup>H) was then administered intravenously via a catheter on the other hind-flipper (0.3-0.6 g/kg body mass) and flushed with saline solution to ensure full injection into the blood stream. Syringes were weighed before and after administration ( $\pm$  0.0001 g, Sartorius balance) to calculate the mass of DLW injected. The labelled isotopes were allowed to equilibrate with the body water pool for 2 h during which the seals were either kept under very light anaesthesia or kept in a quiet closed environment. Equilibration times have previously been determined by serial blood sampling to be less than 2 hours on fur seals (Costa, 1987; Arnould, 1995). At the end of this period, a second blood sample was taken.

During the DLW equilibration period, data loggers were glued to the dorsal mid-line fur using a 2-part Devcon 5-min epoxy glue. Daily Diary tags (DD, Wildlife Computers) recording tri-axial acceleration and tri-axial magnetic field at 16 Hz, and depth, light level, and water temperature at 1Hz were glued as close as possible to the projection of the center of mass on the back of the animal (roughly between the scapula). Fastloc GPS MK10 loggers (Wildlife Computers) were glued lower down the back from the DD tags. They recorded GPS coordinates along the track of the animal at sea, as well as depth and water temperature at 1 Hz. Once the devices were securely attached and the second blood sample was taken at the end of the equilibration period, the females were released upon full recovery from the anaesthesia and allowed to rejoin the colony.

Individuals were recaptured after a single foraging trip and anaesthetized as previously described. A final blood sample was taken to determine isotope levels of <sup>3</sup>H and <sup>18</sup>O at the end of the foraging trip, and all the data loggers were removed by cutting the fur beneath them. A second set of morphometric measurements were also taken at this time.

# 2.3.2 Daily energy expenditure – doubly-labeled water method

All blood samples were collected in Monovette syringes (Sarstedt) coated with Li-Heparin and containing a plasma-red blood cell separator. Plasma was isolated from red blood cells either by centrifugation at 1000 g for 10 min (northern fur seals), or by natural gravity separation for 4 h when no electricity was available (Antarctic fur seals). Plasma samples were then flame sealed into 2 x 100  $\mu$ L glass capillary tubes, and stored at room temperature until isotopic analyses were performed.

For isotopic analyses, plasma samples were vacuum distilled (Nagy, 1983) and the resulting distillate was used to produce CO<sub>2</sub> and H<sub>2</sub> (methods in Speakman, 2005 for CO<sub>2</sub>; and Speakman and Krol, 2005 for H<sub>2</sub>). The isotope ratios <sup>18</sup>O: <sup>16</sup>O and <sup>2</sup>H: <sup>1</sup>H were analysed using gas source isotope ratio mass spectrometry (Optima, Micromass IRMS and Isochrom  $\mu$ G, Manchester, UK). Samples were run alongside three lab standards for each isotope (calibrated to International standards) to correct delta values to ppm. Isotope enrichments were converted to CO<sub>2</sub> production for each individual using a two-pool model (*i.e.*,considering respective individual dilution spaces for <sup>18</sup>O and <sup>2</sup>H), best suited for larger animals including pinnipeds (Speakman, 1987; Schoeller, 1988; Sparling et al., 2008). Initial isotope dilution spaces were calculated using the plateau method (Halliday and Miller, 1977).

Several approaches can be taken to account for evaporative water loss when calculating metabolic rates from DLW concentrations (Visser and Schekkerman, 1999). I used 4 equations for my calculations: the equation 7.43 in Speakman (1997) that assumed evaporation of 25% of the water flux which minimizes error in a range of conditions and

were deemed most appropriate in a validation study on seals (Sparling et al., 2008); and the equations from Speakman (1993), Speakman et al. (1993) and Coward et al. (1985) that were recently shown to be the most accurate on captive northern fur seals in summer and fall when compared to metabolic rate measurements via respirometry (Dalton et al., 2014b). Finally, I converted  $CO_2$  production rates into daily energy expenditure using a respiratory quotient (RQ =  $O_2$  consumption /  $CO_2$  production) of 0.80 as an estimate based on measurements performed in spring and summer (0.80) and fall (0.77) on northern fur seals (Dalton et al., 2014b) and on grey seals on a typical fish diet (0.76, Sparling et al., 2008).

The study individuals spent time on land after the post-equilibration sample and upon return to the colony before recapture and the final blood sample was collected. Energy spent during this 'non-foraging' time was part of the DLW measurement so energy expenditure at sea was calculated by subtracting on-land expenditure from the total estimate using previously determined values for lactating females in northern (4.67 W/kg, chapter 5 in Gentry and Kooyman, 1986) and Antarctic fur seals (4.56 W/kg in Costa and Trillmich, 1988) while on land.

## 2.3.3 Diving and foraging behaviour parameters

I used depth data recorded by the DD tags to determine diving behaviours, or depth data recorded by the fastloc MK10 when the Daily Diary tags malfunctioned. Any drift in the pressure sensors or error spikes were corrected prior to analyses using a zero-offset correction. Diving behaviours were quantified using a custom-made R program previously developed on Antarctic fur seals. Dives were defined as periods of time that animals spent under water below a minimum depth of 3 m and for a minimum of 4 s until they went back to the surface and I derived dive duration and maximum dive depth for each of them. The end of the descent phase was defined as the point at which the ascent rate was below 0.4 m/s and the start of the ascent phase was defined as the point at which the ascent rate exceeded 0.4 m/s. Bottom duration was the difference between the end of the descent and the beginning of the ascent.

Distances traveled on the surface of the ocean (horizontal distances) were calculated by measuring the linear distance between two successive GPS locations taking into account the curvature of the Earth using the Haversine formula (Sinnott, 1984). GPS locations have a high spatial and temporal resolution (they were set to record a location every 5 min), so GPS tracks did not require interpolation or filtering (Tremblay et al., 2006). Part of the distance traveled under water while diving is inherently taken into account in the measured horizontal distance traveled. I approximated vertical distance traveled while diving by doubling the maximum dive depth of each dive (underestimate).

#### 2.3.4 *Time-activity budget*

Fur seal behaviours were separated into 4 categories to determine time-activity budgets: 1) Diving for foraging; 2) resting and sleeping; 3) surface activity, grooming and slow travel; and 4) fast transiting at the surface. These 4 behaviours were identified using a custom-made classification-tree algorithm in R, parameterized as follows:

- 1. *Diving* and foraging time was defined as the period when animals were actively diving (see *Diving and foraging behaviour* parameters above) and included the postdive intervals. Post-dive intervals were defined as the interval between two successive dives during which time the animal needed to return to the surface to replenish its oxygen stores and were estimated using the Bout-Ending Criterion (BEC) calculated with the maximum likelihood estimation method using the package diveMove in R (S. Luque). This validated method is considered to accurately measure the post-dive intervals of diving fur seals (Luque and Guinet, 2007).
- 2. *Resting* time was calculated by first applying a running variance over 3 s on the raw acceleration of each of the 3 axes. It was then defined as the time when the resulting acceleration variance signal was less than 2.5 m/s<sup>2</sup> for all 3 axes for more than 5 min (Figure 2.1). This acceleration threshold was determined visually and was similar for all animals. Very short spikes in variance (of maximum 5 s) in the middle of long periods of variances (30 min or more) below the threshold of 2.5 m/s<sup>2</sup> were considered to be due to the animal changing position while laying at surface and body orientation in the water (verified with the change in static acceleration) and were still considered in the resting time.
- 3. *Transiting* time was the period during which the animals were neither diving nor resting, and were moving at the surface at or faster than 1m/sec. Surface speed was

calculated from the distance between 2 successive GPS points and dividing it by the interval of time between them.

4. *Surface activities, grooming and slow travel* time occurred when the animals were neither diving nor resting, and were moving at the surface at a speed < 1m/s. Surface speed was calculated by determining the distance between two successive GPS points (see *Diving and foraging behaviour* parameters above) and dividing it by the interval of time between them.

Gaps in acceleration due to daily diary tag malfunction for northern fur seals were also quantified as a percent of the foraging trip duration (from 0.3 to 11.5 % of the datasets) that I accounted for in the calculation of ODBA and VeDBA by substituting average overall acceleration to the times when no data were recorded. Accuracy of the classification-tree model was visually verified over the entire foraging trip for all animals.



Figure 2.1: Example of depth profile, acceleration signals over the 3 axes, and combined running variance for 35 min for a female northern fur seal foraging in the Bering Sea.Variance was calculated over the 3 acceleration channels, variance thresholds showing the level below which the animal was considered to be sleeping or/resting (when the signal stayed below the threshold for more than 5 min at a time).

#### 2.3.5 Dynamic body acceleration

Overall Dynamic Body Acceleration (ODBA, Wilson et al., 2006) and Vectorial Dynamic Body Acceleration (VeDBA, Gleiss et al., 2011) were calculated using the tri-axial acceleration data collected at 16Hz by the DD tag on the back of the animals. I performed the same analyses on both ODBA and VeDBA metrics, but I refer to these metrics as DBA for simplicity. Care was taken to standardize the positioning of the devices as close as possible to the center of mass of the seals to ensure accuracy of measurements between and within animals. The three axes X (surge, roll), Y (sway, pitch) and Z (heave, yaw), were first individually normalized using static data collected on all azimuths while the tags were still on a hard surface. The normalized signal was then filtered using a running mean of 2 s (Shepard et al., 2008; Fahlman et al., 2013; Dalton et al., 2014b) to dissociate the static acceleration (due to the positioning of the animal in space in respect to gravity) from the dynamic acceleration ( $X_{dyn}$ ,  $Y_{dyn}$  and  $Z_{dyn}$ , due to the movement of the animal). I then calculated ODBA and VedBA using the following equations:

$$ODBA = |X_{dyn}| + |Y_{dyn}| + |Z_{dyn}|$$
$$VeDBA = \sqrt{X_{dyn}^2 + Y_{dyn}^2 + Z_{dyn}^2}$$

ODBA or VeDBA have been traditionally calibrated as indices of energy expenditure by averaging their values over the period of homogenous activities during which reference metabolic rates are measured (such as with respirometry or heart rate monitoring) (Fahlman et al., 2008; Clark et al., 2010; Halsey et al., 2011c). However, my estimation of metabolic rate from the doubly-labelled water method provided a single measurement for the foraging trip at sea (see below) of different durations, during which the seals displayed a wide range of behaviours. In the case of averaging, the mean values of DBA for animals engaged in similar behaviours (*i.e.*, with similar acceleration signals) should be similar regardless of whether the trip occurred over 2 days or 8 days. However, the longer trip would have higher total energy expenditure. Thus, measures of DBA were normalized for time spent at sea.

I normalized acceleration for time spent at sea by first integrating DBA over periods of 1 s using a linear interpolation (composite trapezoid rule; 'auc' function from the R package MESS, Author C. Ekstrom). Areas under the curves were then summed over the entire foraging trip or over the period of specific activity types previously defined. Another normalized metric was to multiply the average DBA over their foraging trip by the time the animals spent at sea. Finally, calculated the 'noise–free' DBA defined as the dynamic body acceleration from which the residual acceleration caused by waves and wind at the surface was removed. I also normalized this metrics by multiplying the average over the foraging trip by the time the foraging trip lasted for each seal individually.

I assumed that the acceleration recorded when animals were resting and sleeping on the surface of the water was mostly caused by the movement of water (waves etc.) and was not produced by the seals. I consequently subtracted this 'residual' DBA from transiting and surface movement DBA to obtain 'noise-free' DBA (DBA<sub>NoiseFree</sub> in  $m/s^{2}*min$ ) as follows:

$$DBA_{NoiseFree} = [T_{Transit.} \times (DBA_{Transit.} - DBA_{Rest.})] + [T_{Surf.} \times (DBA_{Surf.} - DBA_{Rest.})] + (DBA_{Dive} \times T_{Dive.})$$

where DBA<sub>I</sub> are the average DBA (m/s<sup>2</sup>) displayed during each type of activities *i* mentioned in section Time-activity budget and  $T_i$  are the time spent performing each activity *i*.

#### 2.3.6 Statistical analyses

*Foraging parameters* – Statistical differences between 2 groups (for example between species, or between 2 activity types) were tested with two-sample *t*-tests ( $\alpha = 0.05$ ) or Mann-Whitney tests depending on normality. Averages for dive parameters, such as for dive depths and dive durations, are nested within animals and were calculated using linear mixed effect models with no fixed effects (only the intercept is calculated) and with individual as a random effect to take into account that each animal performed a different number of dives.

*DBA versus energy expenditure over full foraging trip* – I tested whether VeDBA or ODBA could reliably predict total energy expenditure at sea in fur seals using general linear models (lm, 'stats' package, R 3.0.3) or general linear model using generalized least square that allows for unequal variances (gls, 'nlme' package, R 3.0.3) after verifying model assumptions. Metabolic rate and ODBA or VeDBA were mass-corrected for each animal by

dividing the values by individual body mass, as both of these parameters are known to depend on the mass of the animals (Kleiber, 1947; Gleiss et al., 2011). Relationships with energy expenditure for different measures of ODBA and VeDBA over the full foraging trip were investigated, including 1) areas under the curve over the entire foraging trip, 2) average ODBA/VeDBA normalized by time each animal spent at sea, and 3) total 'noise-free' DBA to account for differences in foraging duration normalized by time each animal spent at sea (see *Time-activity budget*).

*Activity-specific metabolic rate* – Energy expenditure per type of activity was calculated by fitting the following model with general linear models (lm, 'stats' package, R 3.0.3):

Eq. 2.1 
$$EE = C_{Transit} \times T_{Transit} + C_{Dive} \times T_{Dive} + C_{Surf} \times T_{Surf} + C_{Rest} \times T_{Rest}$$

where EE is the total energy expenditure in MJ, and the parameters  $T_i$ , are the time in d spent per activity (*i*), *i.e.*,diving (*Dive*), transiting (*Transit*), performing slow surface activities (*Surf*) and resting (*Rest*). This means that the parameter estimates  $C_i$  for each type of activity are in MJ/d and correspond to the rate of energy expenditure for diving, transiting, surface activity, and resting respectively. I consequently estimated the activity-specific metabolic rates by fitting Eq. 2.1 for all my seals to obtain the parameter estimates  $C_i$ . If EE was estimated for the total DLW time (including time on land between 2 captures), the term  $C_{land}$  $\times T_{land}$  was added to Eq. 2.1.

I forced the intercept through 0 because no energy is spent if no time passes. As  $R^2$  values get overinflated in models without intercepts, I computed the  $R^2$  as 1 - (RSS / ESS), where RSS is the regression sum of squares and ESS the error sum of squares of the models, as an additional mean of comparison of the variance between models.

Normality, homogeneity of variances, and correlation between different explanatory variables were verified to ensure accordance with the model assumptions, even though the parameters  $T_i$  are not theoretically independent from one another. The model (Eq. 2.1) was compared to the NULL model which considered only time at sea as the independent variable. In models with more than two parameters, the sample size per species became too small (n =

12 for NFS and n = 13 for AFS) and the power too low to obtain relevant statistics, so I pooled both species together.

*Multivariate predictive models* – I first tested whether all the activity-specific DBAs together were a good predictor of energy spent at sea using the same statistical methods described for full foraging trip. Measured energy expenditure and activity-specific DBA were all mass corrected, and were multiplied by respective time animals spent performing different activities. Second, I tested which other parameters would best predict energy expenditure at sea over a full foraging trip time-scale. These included foraging parameters (total distance traveled, travel rate, vertical distance traveled, dive rate, dive number), DBA parameters (DBA as area under the curve for total trip or for time spent displaying each type of activity), time-activity budget parameters (total time at sea or time displaying each type of activity), and morphometric parameters (mass and species).

All combinations of parameters were tested, and best models were selected based on the AICc (second-order information criterion or AIC adjusted for small sample size compared to the number of estimated parameters). I also computed AICc weights to obtain information about the model probability given the data and a set of models where one of the models is considered the best model for the situation at hand (log-likelihood of models normalized to be a set of positive AICc weights adding to 1, Burnham and Anderson, 2002). I also built models with energy expenditure corrected for mass of the animals (MJ/kg). In this case, I mass-corrected all DBA parameters, and did not include the mass of the animals in the list of tested explanatory variables. Finally, I compared estimated total energy expenditure from the best models to my DLW measurements to determine the accuracy of my method. All results are means ± SE.

#### 2.4 Results

Three DD tags failed to record any data and 4 stopped recording before the end of the foraging trip. Seven females also came back on land with blood H and O isotopic levels too close to initial background levels to yield accurate metabolic rate measurements and were removed from further analyses. Consequently, sample size for analyses that only required

acceleration data or that only required energy expenditure data was n = 16 for northern fur seals and n = 17 for Antarctic fur seals. However, females missing acceleration data were usually not the ones also missing metabolic rate measurements. Consequently, sample size for analyses in which energy expenditure and acceleration data were combined was n = 12for northern and n = 13 for Antarctic fur seals

Northern fur seal females weighed  $37.9 \pm 1.3$  kg prior to departure (range 30.8 - 55.6 kg), and gained  $1.1 \pm 0.7$  kg after a single foraging trip (which corresponds to  $3.5 \pm 1.8$  % of their initial body mass). Female Antarctic fur seal were slightly smaller than northern fur seals with an average mass of  $31.0 \pm 0.8$  kg (range 24.0 - 39.0 kg), and gained  $0.6 \pm 0.6$  kg during a foraging trip (which corresponded to  $2.2 \pm 1.8$  % of their initial body mass). Foraging trips lasted  $7.96 \pm 2.17$  d (range 4.26 - 12.03 d) for northern fur seals, and  $7.65 \pm 3.88$  d (range 2.34 - 15.47 d) for Antarctic fur seals (no inter-species difference p = 0.750). Both species travelled similar distances during their foraging trip (p = 0.221):  $750 \pm 50$  km (range 391 - 1200 km) for northern fur seals and  $635 \pm 77$  km (range 225 - 1295 km) for Antarctic fur seals.

#### 2.4.1 Relationship between total energy expenditure and total DBA

Energy expenditures estimated from 4 different dilution equations are summarized in Table 2.1. Northern and Antarctic fur seals spent the same amount of energy while foraging at sea (p > 0.09). In addition, rate of energy expenditure per day was slightly, but not significantly greater for northern fur seals whether for the total DLW time or for the at-sea time (all 0.09 < p < 0.06, and 0.08 < p < 0.06). Rate of energy expenditure at sea was slightly higher than when time on land was taken into account for both species (although not significantly, all p > 0.47).

VeDBA performed slightly, but not significantly, better than ODBA in all tested models. Consequently, I only present results fitted with VeDBA and refer to this measure henceforth as DBA. Averaging dynamic body acceleration over the entire foraging trip (and for each type of activity, see below and in Table 2.2) showed that total average DBA was greater for Antarctic fur seals (0.411  $\pm$  0.02 m/s<sup>2</sup>) than for northern fur seals (0.312  $\pm$  0.014 m/s<sup>2</sup>, p = 0.0004).

Table 2.1: Energy expenditure estimated for 16 lactating northern (NFS) and 17 Antarctic fur seals (AFS) during a single foraging trip. Total time refers to the energy spent between two measurements of isotopic concentrations, and at-sea time refers to only the time when animal were at sea (see Appendices section). Values are mean  $\pm$  SE.

		Speakm 2-pool r	an 1997 method	Speakman et al. 1993		Coward et al. 1985		Speakman 1993	
Energy expenditure		NFS	AFS	NFS	AFS	NFS	AFS	NFS	AFS
Total time	MJ/d	22.67	19.19	20.02	17.02	19.57	16.15	20.50	17.41
		± 1.38	± 1.18	± 1.27	± 1.08	± 1.35	± 0.99	± 1.28	± 1.09
At- sea time	MI/d	24.22	20.39	20.93	17.72	20.39	16.66	21.53	18.20
	ivij/u	± 1.58	± 1.21	± 1.47	± 1.15	± 1.65	± 1.11	± 1.48	± 1.15
	MJ	179.93	138.93	155.10	121.41	150.78	114.13	159.68	124.53
		± 14.50	± 18.88	± 13.01	± 17.06	± 14.26	± 16.30	± 13.26	± 17.35

Average DBA for the entire foraging trip (DBA<sub>M</sub>) only explained ~ 36% of variability in energy expenditure at sea (R<sup>2</sup> = 0.36) using measures of energy expenditure from Speakman et al. (1993) equation. This relationship between DBA and energy expenditure improved somewhat when DBA was controlled for time animals spent at sea either by integrating DBA (DBA<sub>I</sub>) by normalizing average DBA by the time at sea (DBA<sub>MT</sub> average DBA), or by calculating 'noise-free' DBA over the entire foraging trip (DBA<sub>Noise-Free</sub> multiplied by time at sea). Under these scenarios, DBA<sub>I</sub> or DBA<sub>MT</sub> was able to explain ~ 60% of the variance in energy expenditure (Figure 2.2 B&C).

When fitted individually by species, the relationships between EE and DBA were significant for Antarctic fur seals, but not for northern fur seals. For Antarctic fur seals, EE  $(MJ/kg) = (-0.19 \pm 0.68) + (3.10^{-6} \pm 4.10^{-7}) \times DBA_I (m/s/kg) (R^2 = 0.80, AIC = 45.5, slope p = 3.10^{-5})$ , while for northern fur seals, EE  $(MJ/kg) = (3.00 \pm 1.15) + (1.10^{-6} \pm 1.10^{-6}) DBA_I (m/s/kg) (R^2 = 0.08, AIC = 43.6, slope p = 0.363)$ . The same general relationships were found using DBA<sub>MT</sub> or DBA<sub>Noise-Free</sub>. Rate of energy expenditure (in MJ/d) was never accurately predicted by acceleration, whether with averaged or integrated DBA (Figure 2.2 D, E, F), or over the entire foraging trip or by activity type (all R<sup>2</sup> < 0.15).



Figure 2.2: Relationships between energy expenditure (A, B and C) or rate of energy expenditure (Panels D, E and F) and the average dynamic body acceleration (A, D), the area under the curve of the dynamic body acceleration (B, E) over the entire foraging trip or the average DBA normalized by time spent at sea (C, F) for northern fur seals (yellow triangles, n = 12) and Antarctic fur seals (green squares, n = 13) during one single trip at sea. Each data point represents a single animal and was mass corrected. There were significant species-specific relationships between DBA and energy expenditure (general linear regressions with 95% CI dotted lines in A, B and C), but not between the DBA and the rate of energy expenditure (D, E and F). Equations were: Graph A: EE =  $(-1.17 \pm 1.49) + (360.29 \pm 103.51) \times DBA_M$  (+ 2.11 ± 0.86 for NFS only), R<sup>2</sup> = 0.36, AIC = 102.1, slope p = 0.002, species p = 0.02. Graph B: EE =  $(0.21 \pm 0.68) + (2.10^{-6} \pm 4.10^{-7}) \times DBA_I$  (+ 1.17±0.53 for NFS only), R<sup>2</sup> = 0.63, AIC = 88.2, slope  $p = 3.10^{-6}$ , species p = 0.039. Graph C: EE =  $(0.65 \pm 0.65) + (0.027 \pm 0.005) \times DBA_{NoiseFree}$  (+ 1.31 ± 0.53 for NFS only), R<sup>2</sup> = 0.56, AIC = 84.7, slope  $p = 5.10^{-5}$ , species p = 0.023.

# 2.4.2 Time-activity budget and activity-specific energy expenditure

Time-activity budgets revealed that the proportions of time allocated to transiting, diving, and surface activities / grooming were relatively evenly distributed between 25 and 35% average of the seals time at sea – range 20 - 45% between animals – for both species (Table 2.2). Conversely, fur seals spent only a small part of their time resting and sleeping at

the surface, (~ 6 - 10%). Northern and Antarctic fur seals allocated similar time to foraging (p = 0.328) and transiting to and from their foraging grounds (p = 0.063). However, Antarctic fur seals spent more time grooming and moving slowly at the surface (p = 0.013) and less time resting and sleeping (p = 0.040) than northern fur seals.

Models from Eq. 2.1 were fitted for each individual energy expenditure measures from the DLW measurement depending on the equation used (see Table 2.3), but I only report results from the best model with the lowest AIC and BIC values (Table 2.3, energy expenditure calculated with the equation from Speakman et al., 1993). Models involving time-activity budgets were better predictors of energy expenditure than models with only total time at sea or total time of DLW measurement (Table 2.3). As explanatory variables are in days at sea, and the response variable is in MJ, parameter estimates for each variable indicate the rate of energy expenditure (in MJ/d) for each type of activity.

The two major drivers of energy expenditure in fur seals were diving and transiting (lowest p values, Table 2.3). The seals used on average  $\sim 30 - 31$  MJ/d diving, which was greater than the total rate of energy expenditure measured over the entire foraging trip (19.47  $\pm$  1.05 MJ/d over all the seals). I also estimate that transiting at the surface required less energy than diving ( $\sim 18 - 21$  MJ/d), but required more energy than performing surface activities ( $\sim 14 - 15$  MJ/d). Outputs from Eq. 2.1 provided a negative parameter estimate for resting activity. However, standard errors for the term were more than 10-fold greater than the estimate itself, and values were not significantly different from 0. In addition, the p values associated with resting were > 0.05. I consequently excluded resting/sleeping time from Eq. 2.1. The simpler model did not fit the data better than the full model (Eq. 2.1; F-test p > 0.922) but had a lower AIC value (Table 2.3) and was thus considered the most parsimonious model. It was thus possible to calculate energy spent while performing each activity type (in MJ) by multiplying the parameter estimates from Eq. 2.1 (in MJ/d, see Table 2.3) by individual seal time-activity budgets (in d).

The relationship between time-activity budget and energy spent during the period between the two DLW measurements (*i.e.*,including energy spent during the time females were on land before and after their foraging trip in between capture and recapture) showed similar trends and parameter estimates compared to time-at-sea alone. Rate of energy spent

Table 2.2: Percentage of total time at sea and average VeDBA spent in 4 types of activity for 16 lactating northern fur seal and 17 lactating Antarctic fur seal during a single foraging trip. Activities included active foraging (diving + post dive surfacing), resting at the surface, transiting at a speed greater than 1 m/s, and slow surface movements (< 1 m/s) / grooming. Gap refers to the proportion of time when data were missing and could not be allocated to either of the 4 activity types. Values are means  $\pm$  SE and asterisks show the values significantly different between species.

Activity	Proportion of at activit	-sea time in each ty (%)	Average VeDBA during each type of activity (m/s <sup>2</sup> )			
	NFS	AFS	NFS	AFS		
Foraging/diving	$28.6 \pm 2.0$	$29.0 \pm 0.7$	$0.297 \pm 0.013$	$0.310 \pm 0.018$		
Transiting	$30.5 \pm 1.8$	$26.4 \pm 1.6$	$0.414 \pm 0.013^*$	$0.556 \pm 0.026*$		
Surf mov.	$28.8 \pm 1.4^{*}$	$36.3 \pm 2.0*$	$0.456 \pm 0.22*$	$0.605 \pm 0.017*$		
Resting	$10.9 \pm 1.3$	$8.2 \pm 1.7$	$0.125 \pm 0.008*$	$0.155 \pm 0.004*$		
Gap	$1.1 \pm 0.26$	NA	NA	NA		

Table 2.3: Parameters for general linear models describing the relationship between energy expended by fur seals per day to complete a single foraging trip (in MJ) as well as the cost of different foraging activities (time spent foraging, transiting, performing surface movements / grooming or resting). Note that "Total time" includes time at sea and time on shore tending to pups (*i.e.*, one foraging cycle), while "Time at sea" only accounts for the time away from land. Thus models were fitted with or without resting time.

Dependant variable	Parameter (d)	Estimate (MJ/d)	SE	р	$\mathbf{R}^2$	AICc	Estimate	SE	р	$\mathbf{R}^2$	AICc
EE at sea (MJ)	Time at sea	19.47	1.05	<10 <sup>-15</sup>	0.64	264.6					
i	Diving	30.92	5.77	<2.10 <sup>-5</sup>	0.70	265.7	30.84	5.62	<10 <sup>-4</sup>	0.70	263.7
EE at sea	Transiting	18.66	6.80	0.012			18.50	6.48	0.009		
(MJ)	Surf. Mov.	14.95	7.93	0.073			14.47	6.18	0.028		
	Resting	-3.04	30.67	0.922							
EE total time (MJ)	Total time	18.58	0.88	<10 <sup>-15</sup>	0.65	265.4			-	-	_
. <u> </u>	Diving	29.96	6.02	<10-4	0.64	268.1	29.93	5.87	<10-4	0.70	266.1
	Transiting	21.55	7.49	0.004			21.49	7.27	0.007		
time (MJ)	Surf. Mov.	13.85	8.24	0.105			13.49	6.36	0.045		
	Land	13.11	7.02	0.068			12.98	6.63	0.063		
	Resting	-2.34	32.69	0.942							

for time resting on land equalled the rate of energy spent at sea engaged in grooming and slow surface movements (13 - 14 MJ/d, Table 2.3).

# 2.4.3 Activity-specific DBA

When split by activity, DBA was the greatest when the animals were either transiting  $(0.414 \pm 0.013 \text{ m/s}^2 \text{ for NFS} \text{ and } 0.556 \pm 0.026 \text{ m/s}^2 \text{ for AFS})$  or active at the surface of the water  $(0.456 \pm 0.22 \text{ m/s}^2 \text{ for NFS} \text{ and } 0.605 \pm 0.017 \text{ m/s}^2 \text{ for AFS})$ . DBA while diving was significantly lower than any surface activity  $(0.297 \pm 0.013 \text{ m/s}^2 \text{ for NFS} \text{ and } 0.310 \pm 0.018 \text{ m/s}^2 \text{ for AFS}, p < 10^{-6}$ , no difference between species p > 0.05).

When animals were resting and sleeping at the surface, DBA should equal 0 as animals do not move but I found resting/sleeping DBA still significantly greater than 0 for both species ( $p < 10^{-16}$ ), which means there was significant residual dynamic acceleration due to external factors (waves etc..) when the seals were lying on the water surface. I subtracted resting and sleeping DBA value from the transiting and surface movements DBA and found 'noise-free' DBA averages at the surface did not differ significantly from diving DBA for northern fur seals (0.278 ± 0.023 m/s<sup>2</sup> for transiting, and 0.321 ± 0.035 m/s<sup>2</sup> for surface movements, both p > 0.316), and for transiting for Antarctic fur seals (0.394 ± 0.035 m/s<sup>2</sup>, p = 0.08) although surface-activity/grooming DBA was still slightly greater than diving DBA for AFS (and 0.428 ± 0.021 m/s<sup>2</sup>,  $p < 10^{-3}$ ).

Energy spent while performing each type of activity (MJ/kg) was significantly related to activity-specific DBA when standardized for time spent performing activities (acceleration multiplied by days spent performing activities Figure 2.3). DBA/EE relationships improved greatly when split by type of activity rather than over the full foraging trip (all  $R^2 > 0.90$ ). Both species had similar mechanic-to-energy efficiencies while diving, but differences in slopes indicate that they differed while transiting or during surface activity. Regression slopes are lower during transiting and surface activity than while diving, and correcting for surface acceleration 'noise' decreased the mechanic-to-energy efficiency of seals. Specific equations for diving time, transiting and surface activity from Figure 2.3 included:

Eq. 2.2 
$$EE_{Dive} \sim (0.10 \pm 0.10) + (91.99 \pm 4.42) \times DBA_{Dive} + (0.14 \pm 0.08 \text{ for NFS only})$$
  
 $R^2 = 0.94$ , slope  $p < 2.10^{-16}$ 

Eq. 2.3 
$$\text{EE}_{\text{Transit}} \sim (0.14 \pm 0.05) + (27.62 \pm 1.11) \times \text{DBA}_{\text{Transit}} + [(0.06 \pm 0.08) + (10.19 \pm 2.54) \times \text{DBA}_{\text{Transit}} \text{ for NFS only}]$$
  
 $R^2 = 0.96$ , slope  $p < 2.10^{-16}$ 

Eq. 2.4 
$$EE_{Surf} \sim (0.06 \pm 0.07) + (23.40 \pm 1.48) \times DBA_{Surf} + [(0.22 \pm 0.06) \times DBA_{Surf} \text{ for NFS only}]$$
  
 $R^2 = 0.90$ , slope  $p < 2.10^{-15}$ 

# 2.4.4 Predicting energy expenditure at sea from activity-specific DBA and behaviours

Total energy expenditure predicted by activity-specific DBA and time-activity budgets ( $R^2 = 0.71$ , AICc=91.1) was:

Eq. 2.5 Total EE ~ 
$$(0.09 \pm 0.65) + (99.68 \pm 46.28 \text{ x DBA}_{Dive} \times \text{Time}_{Dive})$$
  
+  $(3.54 \pm 19.80 \times \text{DBA}_{\text{Transit}} \times \text{Time}_{\text{Transit}})$   
+  $(35.15 \pm 14.86 \times \text{DBA}_{\text{Surf.}} \times \text{Time}_{\text{Surf.}}) + [0.31 \pm 0.55 \text{ for NFS only}].$ 

Where total EE is in MJ/kg, activity-specific DBA in m/s<sup>2</sup> and activity-specific time is in days ( $R^2 = 0.70$  and AICc = 90.8). This model with no resting DBA was the most parsimonious model, but did not differ significantly than the model that took Resting into account (ANOVA, p = 0.25). Total energy expenditure estimated from activity-specific DBA and time-activity budgets correlated well with measured energy expenditure from the DLW method ( $R^2 = 0.70$ , Figure 2.4). Slope of the linear regression (1.00 ± 0.14) was not different from 1 ( $p < 4.10^{-7}$ ) and the intercept (1.10<sup>-15</sup> ± 0.56) was also not different from 0. This means that there were no systematic differences between observed and simulated values, and that my model yielded appropriate estimates of total energy expenditure.



VeDBA per type of activity (m/s²/kg for days spent per activity)

Figure 2.3: Relationships between activity-specific VeDBA standardized for individual timeactivity budgets and body mass and activity-specific energy expenditure in MJ/kg for lactating northern fur seals (triangle symbols, n = 16) and Antarctic fur seals (square symbols n = 16) during a single foraging trip at sea. Filled symbols in middle and bottom graphs represents VeDBA at the surface while open symbols represent 'noise-free' VeDBA values corrected for environmental residual dynamic acceleration. Plain lines show the results of linear models that included species as an independent variable, and dotted lines are the 95% CI on the predicted values. Linear model for Diving activity:  $EE_{Dive} = (0.10 \pm 0.10) + (91.99 \pm 4.42) \times DBA_{Dive} + (0.14 \pm 0.08 \text{ for NFS only})$  ( $R^2 = 0.94$ , slope  $p < 2.10^{-16}$ ), for Surface Transiting is  $EE_{Transit} = (0.14 \pm 0.05) + (27.62 \pm 1.11) \times DBA_{Transit} + [(0.06 \pm 0.08) + (10.19 \pm 2.54) \times DBA_{Transit}$  for NFS only],  $R^2 = 0.96$ , slope  $p < 2.10^{-16}$ ; and for Surface activity  $EE_{Surf} = (0.06 \pm 0.07) + (23.40 \pm 1.48) \times DBA_{Surf} + [(0.22 \pm 0.06) \times DBA_{Surf}$  for NFS only];  $R^2 = 0.90$ , slope  $p < 2.10^{-15}$ . The model above is however not the best model I found. Best multivariate models (with a  $\Delta AICc < 2$ , i.e. models that have approximately equal weight in the data) are shown in Table 2.4 (parameter estimates of all models from Table 2.4 are detailed in Table A.2 in the Appendices section). They all included indices related to transiting (area under the DBA curve while transiting (m/s) and/or time spent transiting in min) and diving (number of dives per trip (Dive.nb), dive rate per min (Dive.rate), and/or time spent diving in min). Best predictive models for energy expenditure at sea were:

Eq. 2.6 
$$EE(MJ) \sim (1.616 \pm 1.417) - (3.10-7 \pm 7.10-8) \times Int.DBA_{Transit} - (0.041 \pm 0.010) \times Dive.nb + (0.026 \pm 0.006) \times Time_{Dive} - (2.10-6 \pm 8.10-7) \times Int.DBA_{Rest.} + (0.021 \pm 0.003) \times Total.Time$$

Eq. 2.7  $EE(MJ/kg) \sim (2.314 \pm 0.629) - (3.10^{-7} \pm 3.10^{-8}) \times Int.DBA_{Transit}$  (/kg) - (0.178 ± 0.044) × Dive.rate + (9.10<sup>-4</sup> ± 1.10<sup>-4</sup>) × Time\_{Dive} + (2.10^{-6} \pm 6.10^{-7}) \times Int.DBA\_{Total} (/kg).



Figure 2.4: Comparison between measured energy spent at sea by lactating northern and Antarctic fur seals using the DLW method, and the predicted energy expenditure estimated using Eq. 2.5. The regression line has an intercept of  $1.4 \times 10^{-15} \pm 0.56$ , not different from 0, and a slope of  $1.00 \pm 0.13$ , not different from 1,  $R^2 = 0.70$ .

Table 2.4: Best predictive models for energy expenditure (mass-corrected or not) by lactating northern and Antarctic fur seals.Models were fit using general linear models and were selected using AICc. Int.DBA<sub>Transit</sub>, Int.DBA<sub>Rest</sub> and Int.DBA<sub>Total</sub> mean area under the curve of the dynamic body acceleration (integral) during transit time, resting/sleeping time or total time at sea. Dive.NB is the number of dives performed during the foraging trip; Dive.rate is the rate per min that animal dove over the foraging trip; and Time<sub>Dive</sub>, Time<sub>Rest</sub>, Time<sub>Transit</sub>, Time<sub>Surf.mov</sub> and Total.Time indicate time spent in ech type of activity and total time spent foraging at sea. Only models with AIC less than 2 units above best model are shown. As a comparison, model from Eq. 2.5 for total energy expenditure (in MJ) as a function of activity-specific DBAs (EE ~ DBA<sub>Dive</sub>×Time<sub>Dive</sub> + DBA<sub>Transit</sub>×Time<sub>Transit</sub> + DBA<sub>Surf.</sub>×Time<sub>Surf.</sub> + Species) has a R<sup>2</sup> of 0.71 and an AIC of 91.1.

Eq.		Model	$\mathbf{R}^2$	ΔAICc	AICc weight
2.8	$EE(MJ) \sim$	$Int.DBA_{Transit} + Dive.NB + Time_{Dive} + Int.DBA_{Rest.} + Total.Time$	0.89	0.00	0.260
2.9		Int.DBA <sub>Transit</sub> + Dive.NB + Time <sub>Dive</sub> + Time <sub>Transit</sub> + Time <sub>Surf.mov</sub> .	0.89	0.72	0.181
2.10		Int.DBA <sub>Transit</sub> + Dive.NB + Time <sub>Dive</sub> + Time <sub>Rest.</sub> + Total.Time	0.88	1.50	0.123
2.11		$Int.DBA_{Transit} + Dive.rate + Time_{Dive} + Int.DBA_{Rest} + Total.Time$	0.88	1.80	0.105
2.12	$EE(MJ/kg) \sim$	Int.DBA <sub>Transit</sub> (/kg) + Dive.rate + Time <sub>Dive</sub> + Int.DBA <sub>Total</sub> (/kg)	0.86	0.00	0.327



Figure 2.5: Relationship between estimated energy expenditure from best predictive models, and measured energy expenditure in MJ (graph A) or in MJ/kg (graph B) for lactating northern and Antarctic fur seals. Black squares show results from the two best predictive models: Eq. 6 (graph A) and Eq. 2.10 (graph B) from Table 2.4. Regression lines from best models (squares) are shown on both graphs ( $R^2 = 0.89$  for A, and  $R^2 = 0.86$  for B).

#### 2.5 Discussion

I tested whether dynamic body acceleration and other behavioural parameters could accurately predict energy expended by fur seals, and found that DBA calculated independently of foraging behaviors or time activity budgets did a poor job of estimating energy expenditure of long foraging trips. However, activity-specific acceleration metrics could accurately predict energy used during specific types of behaviours at sea, and could be summed by types of activity (*i.e.*, diving, transiting, surface activity and resting) to accurately estimate energy expenditure of complete foraging trips. Even better estimates of energy expenditure were obtained by using multivariate models that include indices of duration and amplitude of effort (whether acceleration-related or not) of diving and transiting, as well as total foraging time.

My results confirm that acceleration has the potential to estimate energy expenditure in free-ranging animals but that it is only the case if the animals spend performing different types of activities—or time-activity budgets—are taken into account. DBA over a full trip is not an accurate predictor otherwise explaining only 36 % of the variation of EE (Figure 2.2A), an  $R^2$  much lower than values reported for other species of vertebrates (0.47 for diving Steller sea lions, Fahlman et al., 2008; 0.84 in birds, Halsey et al., 2009a; 0.60 for swimming sharks, Gleiss et al., 2010; 0.56 for turtles, Halsey et al., 2011a)

#### 2.5.1 Dynamic body acceleration

DBA (ODBA or VeDBA) is usuallyf study and comparing it to an independent measure of energy expenditure for a number of terrestrial and aquatic taxa (Halsey et al., 2009b). It has been argued that VeDBA estimates should be better than ODBA estimates if logger alignment cannot be standardized between individuals (Qasem et al., 2012). My models with VeDBA had slightly but not significantly lower AIC values (same as for European sea bass with implanted loggers, Wright et al., 2014), which indicates negligible differences in logger positions on the back of seals that cannot explain why DBA failed to accurately predict the total energy expended in my study.

Another explanation for the differences between my validation of DBA as a measure of energy expenditures with fur seals and studies of other species lies in the assumptions behind DBA as a proxy for energy expenditure, and sources of unmeasured error associated with free-ranging environmental conditions (Halsey et al., 2011b). Most studies have validated DBA in captive or semi-captive settings where conditions could be controlled (Wilson et al., 2006; Gleiss et al., 2010; Halsey et al., 2011a; Fahlman et al., 2013). However, free-ranging marine animals live in a dense medium under fluid environmental conditions. In addition, marine mammals such as fur seals spend a significant portion of their time at the water surface, where wind-related motions, mainly through wave action can interfere both with energy expenditure and the dynamic body acceleration signal. Considerable surface acceleration 'noise' occurs in the acceleration signal at the interface between the two mediums of different densities (air and water), which largely disappears when the animals dive.

I attempted to crudely quantify and account for environmental surface motion by averaging the dynamic acceleration at times when the animals were resting and sleeping at the surface. This 'acceleration noise' was slightly greater in the Southern Ocean for Antarctic fur seals  $(0.15 \pm 0.004 \text{ m/s}^2)$  than in the Bering Sea for northern fur seals  $(0.12 \pm 0.009 \text{ m/s}^2)$ . However, environmental dynamic acceleration does not appear to explain much of the uncertainty in my global model given that the models that incorporated average 'noise-free' DBA did not perform significantly better than those with non-corrected DBA. Thus, it appears that surface noise augments body acceleration but that the influence is consistent between seals over their foraging trips (as shown by the very small standard errors mentioned above) and should thus not affect relationships. An interesting application of this would be to use acceleration signals during these time when seals are sleeping/resting as indices of ocean conditions in a given time and location.

Another difference between other validation studies and mine is the variation in the duration of measurements taken (*i.e.*,hours vs. days or weeks). In my case, the northern and Antarctic fur seals undertook foraging trips that averaged 7 - 8 days (range 2.5 - 15 days). A fur seal that makes a long foraging trip is likely to allocate energy differently compared to an animal that makes a short trip—yet averaging DBA over time does not account for such

differences. I suspect this is why none of my analyses that used metabolic rate as my reference measurement (in MJ/d, Figure 2.2D, E & F) yielded significant relationships with DBA, unlike the relationships reported for other studies.

I tried to account for foraging trip duration in two different ways. The first was to integrate the DBA signal (Figure 2.2B), and the second was to standardize the averaged DBA (or 'noise-free' DBA) by multiplying it by foraging time (Figure 2.2C). Both procedures improved the DBA/EE relationship between DBA and energy expenditure. Most notably, the  $R^2$  increased to 0.63 in both cases, suggesting that difference in foraging trip durations might explain ~ 30% of the uncertainty in the initial model. It is equally noteworthy that time spent at sea ( $R^2 = 0.64$ ) proved to be a much better predictor of energy expenditure over a foraging trip than was DBA (Table 2.3). However, it is debatable whether the improved predictions of energy expenditure that come by combining DBA and trip duration are sufficient given that the predictive power of such a combined model can only explain ~ 60% of the variation in the data.

Energy expended by fur seals could have been affected by a number of sources of errors that had no effect on DBA—or vice versa. For example, body condition of seals affects buoyancy, which in turn affects mechanical power and cost of transportation (through changes in buoyancy and gliding — Williams et al., 2000; Wilson et al., 2010), but would not affect DBA. In addition, fur seals typically transit by porpoising at the surface, which means that they switch from moving through air to moving through water in less than a second. Such changes in movement between mediums with highly different densities likely affect DBA in different ways, as would differences in gaits between swimming and porpoising movements. In addition, effects of 'barrel-wheel' turns while diving or surface grooming create centrifuge forces that remain part of the dynamic acceleration after low-pass filtering. This relationship between non-rotative acceleration and energy expenditure. However, the only way I could have estimated whether centrifugal force had a significant impact on the dynamic acceleration was with a gyroscope (Noda et al., 2012), which my data loggers did not contain.

In addition to the physical effects of moving through air and water, physiological functions can affect energy expenditure and DBA in different ways. For example, thermoregulation costs may increase while seals rest at the surface, or animals will incur costs of digestion, growth, and gestation that are independent of DBA. It has been suggested that the energy budgets of fur seals and other pinnipeds are largely a balance between basal metabolic costs, locomotion costs and feeding costs (Costa and Williams, 1999). Out of these expenses, digestion costs are non-negligible, and can increase the metabolic rate of seals by ~ 45% in Weddell seals, or ~ 55% in Steller sea lions (Rosen and Trites, 1997). A study with domestic chickens showed a clear increase in metabolic rate during digestion while DBA remained stable (Green et al., 2009). No differences have been reported in the relationship between DBA and energy expenditure for fasted and non-fasted Steller sea lions (Fahlman et al., 2013), but increase in metabolic rate due to feeding depends on the quantity of fish ingested (usually small in captive trials to keep animals working) and are usually deferred to resting time few hours apart from feeding (Sparling et al., 2007). It could thus be significant in free-ranging seals. Collectively, such studies point to an uncoupling between a significant portion of metabolic rate and the acceleration which could potentially contribute to the uncertainty in the DBA and energy expenditure relationship (Halsey et al., 2011b).

In addition to the inherent uncertainties associated with DBA measures of energy expenditure in free-ranging fur seals, there are also uncertainties with using the doubly-labeled water method (DLW) as my reference measure of energy expenditure. Some studies of specialist marine carnivores have suggested that the DLW method has high accuracy, but low precision (Speakman, 1993). For example, the DLW method applied to grey seals (*Halichoerus grypus*) subjected to simulated foraging conditions over 5-day periods yielded estimates of energy expenditure for groups or population averages that were similar to estimates derived from respirometry (Sparling et al., 2008). However, precision was quite low at the individual level (Sparling et al., 2008). Group error was 0.5% (based on the 2-pool equation from Speakman (1997) or the original calculations from Lifson and McClintock (1966)) when tested against 32 respirometry measurements on 9 individuals over 3 years ). However, individual error was  $\sim \pm 40\%$ . Similarly, a study of captive northern fur seals showed that the average error of the DLW method compared to respirometry measurements could be as low as  $\sim 0.8\%$  but as high as  $\sim 27\%$  depending on the calculation method used

and the time of year (lowest in the fall and highest in the summer, Dalton et al., 2014b). Consequently, the error associated with my reference measurement of energy expenditure is likely significant since I compared DLW to DBA measurements at the individual level (one DLW and one DBA point per animal).

I attempted to decrease the DLW-associated error by using estimates derived from the best equation provided by Sparling et al. (2008) as well as the 3 best equations derived by Dalton et al. (2014b), and chose the best fit for my data out of the 4 equations. I recognize that using DLW measurements as a reference measurement of energy expenditure comes with associated caveats, but was the only option available to us to study energy expenditure at sea for free-ranging fur seals. Either way, however, there seems to be no escaping the fact that DBA metrics are not an appropriate means to predict the total energy expended over the duration of a full foraging trip regardless of what means I might have used to establish the reference measurement of energy expenditure.

#### 2.5.2 *Time-activity budgets and activity-specific energy expenditures*

DBA is not a good proxy for energy expenditure by fur seals over long periods of time and in field settings, and appears to also perform poorly for other species that engage in different behaviors that have markedly different energetic costs (Green et al., 2009; Halsey et al., 2011b; Dalton et al., 2014b). A study of free-ranging imperial cormorants (*Phalacrocorax atriceps*), for example, found that activity-specific DBAs were better predictors of energy expenditure during a foraging trip than overall DBA, especially if one activity type had a greater energetic cost than others (in this case flying, Elliott et al., 2013). Animals are known to incur different energetic costs for different activities, and different relationships are known to exist between DBA and energy expenditure for seemingly inconsequential activities as different gaits in humans (Halsey et al., 2008), different intensities of swimming in sharks (Gleiss et al., 2009) and different types of muscles involved in the movement of birds (Gomez-Laich et al., 2008). All told, this suggests that the poor ability of DBA to predict the energetic cost of the full foraging trips undertaken by my fur seals was due to differences in time-activity budgets (*i.e.*, how the animals partitioned their time at sea between diving, transiting, resting and surface activities).

Time budgets are the currency that defines foraging strategies and ultimately reflects the foraging efficiencies of animals when combined with energetics. In my study, both species of fur seals had surprisingly similar time-activity budgets despite inhabiting significantly different habitats. The lactating northern and Antarctic fur seals spent ~ 29% of their time diving, ~ 26 - 30% of their time transiting, and ~ 8 - 10% of their time resting and sleeping at the surface (Table 2.2). These activity time-budgets are consistent with previous findings showing northern fur seals spent 22-26% of their foraging trips diving, and Antarctic fur seals spent 35% of their time engaged in diving (Costa and Gentry, 1986; Insley, 2008). Similarly, northern fur seals rested for 17% of their time at sea, while Antarctic fur seals spent just 5% of their time resting (Costa and Gentry, 1986). However, this consistency between the activity time budgets of the 2 species of fur seals masks considerable variability between individual seals which differed by as much as a factor 2 - 2.5 times for time spent diving or foraging trips.

Overall, the average daily metabolic rate at sea (MR<sub>sea</sub>,Table 2.1) for both species of fur seals was  $4.7 \pm 0.3 \times$  basal metabolic rate (BMR in kJ/d) calculated from Kleiber's equation (Kleiber, 1975). This calculated metabolic rate is consistent with values previously reported for Antarctic fur seals ( $4.59 \pm 0.32 \times$  BMR from Arnould et al., 1996) and California sea lions *Zalophus californianus* ( $5.2 \times$  BMR from Boyd et al., 1995b;  $4.8 \times$  BMR from Ponganis et al., 1997). Consequently, total energy spent at sea over a foraging trip ranged from 38 to 259 MJ depending upon the duration of foraging trips and differences in the proportion of time the seals engaged in diving, transiting and resting.

The most expensive activity the fur seals engaged in was diving, which was  $1.6 \pm 0.1 \times MR_{sea}$  (Table 2.3) — or roughly  $7.5 \times BMR$ ,  $(7.0 \pm 0.2$  for NFS and  $8.1 \pm 0.2$  for AFS). This high metabolism seems counter-intuitive given that diving in pinnipeds usually involves hypometabolism to preserve oxygen stores and extend dive duration (Sparling and Fedak, 2004). However, my values are close to previous measures of at-sea metabolic rates for northern fur seals ( $6.16 \times BMR$  from Costa and Gentry, 1986) and Antarctic fur seals ( $6.7 \times BMR$  from Costa et al., 1989), and for underwater swimming metabolic rate in southern sea lions and California sea lions ( $6.8 \times BMR$  from Boyd et al., 1995b; Dassis et al., 2012) –

although greater than metabolic measurements for Steller sea lions over one dive cycle in semi-captive conditions when animals stay stationary at a target at depth (4 × BMR, Gerlinsky et al., 2013). In my case, I derived my estimates of diving metabolism from a best fit model (Eq. 2.1), and not from direct measurements (which are virtually impossible to obtain from free-ranging animals). My estimates of metabolism also accounted for the total cost of diving (which included extended recovery that can occur over periods longer than a single dive cycle)—and show that diving remains the most expensive foraging activity despite the physiological mechanisms that pinnipeds have evolved to extend dive durations. Butler et al. (1995) also found that metabolic rate while diving was greater than when transiting in Antarctic fur seals. However, contrary to both my results, Arnould et al. (1996) found that energy expenditure at sea was negatively related to proportion of time spent diving in Antarctic fur seals, and thus concluded that the most costly activity was transiting rather than diving. They estimated time spent diving using depth recorded every 10 s. The finer scale and more detailed time-activity budget in my study might explain the difference in the results.

Transiting at the surface (surface movement  $\geq 1$  m/s) was 0.96  $\pm$  0.10  $\times$  MR<sub>sea</sub> which was less expensive than diving and roughly equivalent to the average daily metabolic rate at sea  $(0.9 \pm 0.1 \times MR_{sea}$  for NFS, and  $1.1 \pm 0.1 \times MR_{sea}$  for AFS). Transiting fur seals usually porpoised at the surface or swam at depths equivalent to 3 times their body diameters, which would reduce drag forces and cost of transportation (Boyd, 2002; Hindle et al., 2010). Surface activities were less expensive than diving and transiting, and overall less expensive than average daily metabolic rate at sea (*i.e.*,  $0.77 \pm 0.10 \times MR_{sea}$ ). My calculated metabolic rate on land (as well as the surface activity/grooming metabolic rate) was ~  $3.3 \pm 0.1 \times BMR$ which was very close to the  $3.4 \times BMR$  metabolic rate of female Antarctic fur seal females nursing on land (Costa and Trillmich (1988). It is also noteworthy that my calculated metabolic rates during surface activity/grooming and during time on land (Table 2.3) were roughly similar (13-14 MJ/d). This confirms that thermoregulation at sea is likely not an issue in fur seals—at least for most of their time at sea (Costa and Williams, 1999; Dalton et al., 2014a). Resting and sleeping at sea also had negligible metabolic costs for my fur seals (Table 2.3) due most likely to the low cost of doing so and the small proportion of overall atsea time they spent resting (<10%).

The similarities between my field and modeled estimates of metabolism give confidence that my models provide accurate activity-specific metabolic rates. Similarly, the high correlations between the activity-specific energy expenditures and the activity-specific DBAs (all  $R^2 \sim 0.94$ ) gives confidence that mean DBA is a much better proxy for energy expenditure when broken down by activity type. However, the relationships (*i.e.*,the mechanical to energy efficiency) varied by activity type. Changes in DBA affect mechanical power and thus energy expenditure more drastically while diving than while transiting or during surface activities (Figure 2.3). This means that small changes in measures of DBA can lead to larger changes in estimates of diving energy expenditure than of energy spent in surface behaviours.

The mechanical energy efficiency of fur seals differed between underwater activity (diving) and switching from water to air (transiting and surface activity). Similarly, the mechanical-to-energy efficiencies (*i.e.*,slopes of the DBA and energy efficiency relationships) of cormorants have been shown to differ between flying and all other activities (*i.e.*,resting at sea surface, diving and walking on land, Gomez-Laich et al., 2011; Elliott et al., 2013). Collectively, these findings indicate that the mechanical efficiency of different activities are constrained by the medium (air or water), but maybe more so by the mechanics and the types of muscles involved (*i.e.*,the force production to movement relationship of muscles and their contractile properties).

The difference I observed in activity-specific relationships between DBA and energy expenditure was likely due to the media through which they travelled (air/water intersection for transiting and surface activity versus exclusively water for diving) rather than changes in muscle groups involved. Densities of air and water differ by a factor of ~ 800, which undoubtedly affects DBA differently than it affects energy expenditure given that fur seals propel themselves using their fore-flippers. It seems however, that the effect of medium on DBA is inconsequential when DBA is partitioned by types of activity to predict the energy fur seals spent at sea.

The improved accuracy by incorporating activity levels (*i.e.*, swim speeds, frequency of dives, and time spent transiting and diving) to predicting energy expenditure compared to analyses over the full foraging trip may also reflect the fact that my calculations buffered

some of the errors associated with individual DLW measurements. I calculated energy expenditure from an average activity-specific metabolic rate from the energy that each animal spent per activity (in MJ) given their time-activity budget (in d) from the parameter estimates of Eq. 2.1 (in MJ/d, from best fit model between measured energy spent at sea and time-activity budgets of all individuals, results in Table 2.3), and not from an individual DLW measurements. As a consequence, the inherent individual variation and measurement error in DLW measurements was likely reduced in the average. This means that I could attribute  $\sim 20 - 35\%$  of the uncertainty in the global model to errors in the DLW measurements and individual variability in time-activity budgets and foraging strategies (even if it was difficult to tease apart the respective effect of these two parameters).

The impact of factors other than activity levels might have on the total energy expenditure of fur seals is likely minimal given that otariids have high locomotion costs and operate close to their metabolic ceilings (Costa, 2007). For example, the uncoupling that likely exists between DBA and energy expenditures associated with such physiological functions such as digestion, thermoregulation, and growth might be minimal in the case of fur seals foraging at sea with much higher energetic costs due to locomotion. This shows that the time a seal decides to allocate to activities that have different metabolic rates is important for obtaining accurate estimates of energetic costs of foraging in fur seals.

#### 2.5.3 Conclusions

Energy expenditure by fur seals over full foraging trips can be accurately determined from body acceleration, but only if it is done using activity-specific time budgets. The predicted energy expenditure derived from activity-specific measures of body movement (*i.e.*,DBA × activity budget) corresponded well with the DLW measured energy expenditures (Figure 2.4,  $R^2 = 0.70$ ). Including other variables into the model that captured the intensity of activities such as number of dives per trip or total minutes spent diving significantly improved the ability to predict energy expended by the seals during foraging trips ( $R^2 = 0.84$ - 0.89, Table 2.4 and Figure 2.5). However, I also incorporated elements of the time-activity budget in these multivariate models using proxies of diving and transiting effort (intensities) that represent the most expensive parts of the foraging trip. I also included proxies to represent resting and slow surface movement, or total time at sea.

My predictive models reinforce the importance of separating animal behaviours into categories to obtain reliable estimates of energy expenditure. Interestingly, Skinner et al. (2014) found that DBA multiplied by distance traveled and by mass of the animal, and by vertical distance swam were together the best metrics to assess energy expenditure of northern fur seals at sea. This means that they also took into account metrics of effort (DBA, and of diving time). Acceleration is a useful tool for estimating the field metabolic rate of free-ranging fur seals, but it is essential to know how much time the animals spend diving versus transiting and being surface active. It is also important to know the intensity with which the seals perform these activities. Calculating total energy expended by fur seals should thus be done using either the activity-specific DBA paired with time-activity budget (*i.e.*, Eq. 2.5) or the multi-factorial models (Eqs. 2.6 and 2.7).

Being able to accurately calculate foraging costs helps to better understand the energetic requirements of free-ranging seals and other marine mammals, and whether they can be met in the wild. Knowing foraging costs also contributes to assessing the ecological impacts that marine mammals have on trophic webs, and how changes in time-activity budgets due to environmental changes will impact their fitness. Such knowledge is particularly important for the conservation and management of species that are easily impacted by ecosystem shifts and environmental changes, especially for fur seals that are already performing close to their metabolic ceilings, and may have limited scope to adapt to coming climate changes.

# Chapter 3: Do flipper stroke rates and amplitudes accurately predict energy expenditure in foraging northern fur seals and Antarctic fur seals?

# 3.1 Summary

Flipper stroke rate and amplitude have been proposed as proxies to estimate the energy expended by seals and sea lions while foraging at sea, but have not been validated. I investigated how well metrics of flipper strokes correlate with energy expenditure in foraging northern fur seals (Callorhinus ursinus) and Antarctic fur seals (Arctocephalus gazella). Flipper strokes rates and amplitude were measured for 33 lactating females using dynamic bi-axial acceleration obtained from accelerometers deployed on the backs of the fur seals during diving or transiting because residual water movement at the surface clouded the detection of strokes during times of surface activity. I derived activity-specific energy expenditure by calculating time-activity budgets for single foraging trips (time spent diving, transiting at speeds >1m/s, surface activities and sleeping) and by measuring metabolic rate using the doubly-labeled water method. I found that flipper stroke count was a good predictor of energy spent while diving ( $R^2 = 0.76$ ) and to a lesser extent while transiting ( $R^2$ = 0.63), and that it performed better than flipper stroke amplitude. Fur seals spent  $3.79 \pm 0.39$ J/kg per stroke while diving and a cost of transportation ~ 1.6 - 1.9 J/kg/m (and there was no difference between species), but the energetic cost per stroke and cost of transport differed when the seals were diving compared to when they were transiting and routinely porpoising at the surface. In terms of predicting energy spent during a complete foraging trip, I found that flipper stroke amplitude was a better proxy ( $R^2 = 0.63$ ) than flipper stroke count. Flipper stroke amplitude (*i.e.*, an index of power delivered depending on the means of locomotion) encapsulated the type of locomotion or the gait used by the seal better than stroke count. In conclusion, the cost per stroke differed depending on the type of activity the seals were engaged in such that flipper stroke metrics were accurate at the activity level, but were not as accurate an index of energy expenditure over a full foraging trip compared to other acceleration-based proxies such as activity-specific VeDBA.

#### **3.2 Introduction**

Determining energetic costs of foraging is essential for understanding optimal foraging and the fitness of animals yet is difficult to attain. Over the years, different techniques have been developed to determine energy expenditure, such as direct or indirect calorimetry techniques, breathing rates (Le Boeuf et al., 2000), heart rates (Boyd et al., 1995b), and locomotion speed (Hind and Gurney, 1997). While they may be accurate in terrestrial or flying animals, they are usually either impossible to perform in the field or are overall poor predictors in most breath-hold divers species such as marine mammals because of confounding factors inherent to diving physiology and biomechanics (McPhee et al., 2003), changes in buoyancy and gliding (Costa and Gales, 2000), etc.

Cost of transport is defined as the energy needed to move a unit of mass over a distance, usually expressed in J/m or in J/kg/m. As most of the metabolic cost of foraging is thought to come from the cost of transport (Gleiss et al., 2011), proxies of energy spent have been developed to count and characterize stride', 'wingbeats', or—in the case of marine animals—'stroke' rates that are directly linked to the cost of underwater locomotion (Dassis et al., 2012; Maresh et al., 2014). Several methods, such as video images (Hays et al., 2007; Dassis et al., 2012; Dudley et al., 2014) or acoustic recordings (Insley, 2008) can be used to detect wing or flipper stroke rates in top marine predators. However, it is very difficult to assess the intensity of swimming effort (such as amplitude of the strokes) from the count of strides using these techniques. More recently, tri-axial acceleration has been used to investigate the cost of transport and has proven to be a cost-effective means to record data over long periods that can characterize both frequency and relative amplitude of strides as animals propel themselves through air and water (Gleiss et al., 2009; Adachi et al., 2014; Bishop et al., 2015).

In marine predators, the flipper or tail stroke rate can potentially be used to estimate foraging effort. Based on metabolic rates determine in Weddell seals (*Leptonychotes weddelli*) breathing through a hole in the ice at the end of dives, flipper strokes have been found to be a better predictor of energy expenditure during dives than the dive duration itself (Williams et al., 2004). Tail beats correlated with acceleration, which in turn correlated with metabolic rates in salmon (Wilson et al., 2013) and sharks (Gleiss et al., 2009; Gleiss et al.,

2010). Stroke rate has also been shown to vary with activity and dive phase in northern fur seals (Insley et al. 2007). It was also hypothesized for these animals that the cost per stroke multiplied by the average stroke rate of 0.5 Hz for 70% of the diving and traveling time would produce a reasonable correlation with the cost of locomotion (Insley, 2008). However, this needs to be verified under free-ranging conditions, and consideration must also be given to how the amplitude of the strokes affects swimming energetics given that power delivered with each stroke shapes energetics of locomotion (Booth, 2009).

Most studies on marine mammals have been performed either in captive controlled settings (Dassis et al., 2012) or on single types of activities such as continuously diving in species that spend little time at the surface (Williams et al., 2004; Maresh et al., 2014). Otariids such as northern fur seals and Antarctic fur seals are pelagic species that spend a significant portion of their time at sea at the surface where they transit, rest, and groom. The proportion of time spent performing surface activities also greatly affects total foraging effort of free-ranging animals (Chapter 2 and Elliott et al., 2013). Different modes for transport (swimming or porpoising) also affect locomotion costs, as does the media in which the animal evolved (air, water, land) (Gomez-Laich et al., 2013). There is consequently a need to assess whether flipper stroke metrics can accurately reflect energy expenditure of otariids under free-ranging conditions and how it might change with different types of behaviour or media (*i.e.*,air, water and the interface between the two).

My goals were to determine whether flipper strokes of fur seals could be accurately detected at any time and for different types of activities/gaits during foraging trips using accelerometry. I also sought to assess whether stroke rates or amplitude could accurately predict at-sea energy expenditure. Finally, I compared the accuracy of using flipper stroke metrics for quantifying the energetic costs of different types of activities that seals display at sea or over a full foraging trip, and relate it to cost of transport.

# 3.3 Material and methods

#### 3.3.1 Data collection

All data were collected from 20 lactating northern fur seals at the Reef rookery on St. Paul Island (Bering Sea, 57°6'N - 170°17'W) during the breeding season from Aug-Sep 2011, and from 20 lactating Antarctic fur seal females at Pointe Suzanne, Kerguelen Island (Southern Ocean,  $49^{\circ}26$ 'S -  $70^{\circ}26$ 'E) during the breeding season from Jan-Feb 2012. Data were collected under the US NMFS permit # 14329-01, the UBC animal care permit # A10-0364 and the ethical regulations approval from the French Polar Institute (IPEV). All data collection methods are similar as the ones detailed in section 2.3.1 of Chapter 2. Three daily diary tags failed to record any data and 4 stopped recording before the end of the foraging trip. Consequently, sample size for acceleration data was n = 16 for northern fur seal and n = 17 for Antarctic fur seals. I also obtained inaccurate metabolic rate measurements for 7 females during their foraging trip, so they were discarded from further analyses. Consequently, sample sizes for energy expenditure were n = 16 for northern and n = 17 for Antarctic fur seals. Females missing acceleration data were usually not the ones also missing metabolic rate measurements, so sample size for analyses combining both parameters were reduced accordingly (n = 12 for NFS and n = 13 for AFS). All values provided in the result section are mean  $\pm$  SE

# 3.3.2 Energy expenditure, diving and foraging parameters, and time-activity budgets

Methods for measuring energy expenditure during foraging trips at sea from the DLW method, for calculating diving and foraging parameters, and for deriving time-activity budget of individual animals from GPS, depth and acceleration data are similar to the ones detailed in Chapter 2. Please refer to sections 2.3.2 2.3.3 and 2.3.4 for further information.

# 3.3.3 Flipper strokes

Flipper strokes were detected and counted using only the X (surge) and Z (heave) accelerometer axes from back acceleration. Fur seals swim by flapping their fore-flippers which results in an up and forward movement clearly and synchronously visible the 2 signals. The dynamic accelerations  $X_{dyn}$  and  $Z_{dyn}$  were obtained using the same filtering method as mentioned in Chapter 2 and then added to increase the signal clarity compared to residual background noise. Each flipper stroke was detected by counting the spikes resulting from a deviation from the gravitational field above a specific threshold (Figure 3.1).

Thresholds were determined for each animal individually as the break point of a curve showing the number of flipper strokes detected for a range of thresholds going from 0.1 to  $0.4 \text{ m/s}^2$ . Thresholds ranged from 0.19 to  $0.24 \text{ m/s}^2$  depending on individuals.

Maximum flipper stroke frequency is  $\sim 1 - 2$  Hz in dolphins, Weddell seals, and elephant seals (Williams et al., 2000; Sato et al., 2003). However, it is known to vary with the size of the animal (Sato et al., 2007). Baikal seals which are closer in mass to fur seals than other investigated marine mammals ( $\sim 45$  kg) have a maximum stroke frequency of  $\sim 2.5$  Hz during ascent and descent phase of dives (average  $\sim 0.25 - 1.75$  Hz, Watanabe et al., 2006). Northern fur seals were found to have an average and maximum stroke rates of  $\sim 0.5$  Hz and 0.89 Hz, respectively, using acoustic detection in diving females (Insley, 2008). However, these values were averaged and not recorded during chase time, so might be lower than the maximum flipper stroke frequency of 2.5 Hz, same as the maximum stroke rate found in a species of similar mass, and any peak within 0.4 s was not considered a flipper stroke. As the dynamic acceleration is centered on 0, I also used the value at the highest point of the flipper stroke as a relative index of flipper stroke amplitude (*i.e.*, of swimming effort).

Flipper strokes where easily detected when animals were diving (Figure 3.1). However, oceanic/wave movements often clouded the acceleration due to the movement of the animal when the animals were at the surface. I was only able to detect flipper strokes when animals were at the surface travelling at a minimum of 1 m/s. Below this travel speed, the animal seemed to be either not stroking, or stroking with such a low amplitude that it was impossible to separate dynamic acceleration signal coming from oceanic movements or animal movements (Figure 3.1).

Consequently, I only measured flipper strokes at depth or at the surface when the animal was moving at or faster than 1 m/s between 2 GPS points. Accuracy of peak threshold and flipper stroke detection were verified visually on ten 5-min periods randomly chosen over the entire foraging trip for each seal. Relationship between flipper strokes and distance traveled was also verified for each animal at depth and while transiting at a speed  $\geq 1$  m/s at the surface (example for 2 NFS and 2 AFS on Figure 3.2) to validate the detection method.


Figure 3.1: Example of detection of flipper strokes from the dynamic acceleration on the X and Y accelerometer axes while diving, while transiting at the surface at speeds > 1m/s, or while performing slower movements at the surface at speeds < 1m/s.



Figure 3.2: Number of flipper strokes detected per distance traveled either vertical distance while diving (red triangles) or horizontal distance while transiting (blue circles) for 2 northern fur seals (NFS 2 and NFS10) and 2 Antarctic fur seals (AFS 4 and AFS 9) between 2 GPS locations during a foraging trip. Transiting at the surface is defined as periods of time when animals travel at speeds > 1m/s between 2 GPS locations.

#### 3.3.4 Statistical analyses

The number of flipper strokes or cumulated stroke amplitude (indicative of swimming effort) swam during dives or between 2 GPS locations were calculated using linear mixed effects models with animal ID nested in species. Fixed effects were horizontal distances for number of flipper strokes while transiting at the surface and vertical distance for flipper strokes while diving. Autocorrelation in the data was corrected using an autoregressive moving average structure (lag 1) and a power structure for any deviance of heterogeneity of variances.

Relationships between energy spent during total trip (or while diving and transiting) and number of flipper strokes or cumulated stroke amplitude were estimated using general linear models (lm, 'stats' package, R 3.0.3) or general linear model using generalized least squares that allows for unequal variances (gls, 'nlme' package, R 3.0.3) upon verification of model assumptions. These methods minimize the squared residuals in the dependent variable.

Cost of transportation (COT) was estimated using 2 separate methods. First COT was estimated from the slope of the relationship between diving energy expenditure and vertical distance traveled using a using general linear model (lm, 'stats' package, R 3.0.3) as mentioned above (in J/kg/m). I also calculated cost of transportation by multiplying the energetic cost per stroke (*i.e.*, 3.79 J/kg/stroke) by the number of strokes necessary to travel one meter (stroke/m) for each individual.

Finally, I tested which associations of parameters (cumulative or average stroke amplitude, number of flipper strokes while diving, while transiting or total) would best predict energy expenditure at sea over a full foraging trip time-scale. I tested all variable combinations and selected best models based on the AICc (second-order information criterion or AIC adjusted for small sample size compared to the number of estimated parameters). I also computed AICc weights to provide information on the model probability given the data and a set of models where one of the models is considered the best model for the situation at hand (Burnham and Anderson, 2002).

# 3.4 Results

Mass of the animals, flipper length and width, foraging parameters, energy expenditure during total foraging trip or during diving or transiting are detailed in Table 3.1. Both species had similar trip duration, distance traveled and time spent diving versus transiting but AFS performed on average more dives of shorter duration. They also had similar at-sea metabolic rates and spent on average similar amount of energy for foraging and for transiting. However, individuals spent more energy while diving than while transiting (Table 3.1). During these times, fur seals beat their flippers  $21.5 \pm 2.3$  times per dive ( $22.6 \pm 3.6$  for AFS and  $20.3 \pm 2.7$  for NFS, p = 0.759), which corresponds to  $0.41 \pm 0.03$  Hz while diving ( $0.44 \pm 0.06$  for AFS and  $0.37 \pm 0.03$  for NFS, p = 0.004). Similarly, rate of flipper stroke at the surface when animals were transiting (surface speed > 1 m/s) was  $0.44 \pm 0.02$  Hz ( $0.53 \pm 0.03$  for AFS and  $0.35 \pm 0.02$  for NFS,  $p < 3.10^{-4}$ ). Rates of flipper strokes were similar between surface transiting and diving within species (all p > 0.2). Rates of flipper stroke while diving and transiting were negatively related to body mass (smaller animals have a greater stroke frequencies, p < 0.01) but differences remained significant between species even when rates were mass corrected (all  $p < 1.10^{-5}$ ).

Antarctic fur seals beat their flippers on average  $270 \pm 14$  times and northern fur seal only 190 ±19 times to travel a distance of 1 km at the surface of the water ( $p < 10^{-4}$ ). Both species used the same number of flipper strokes to swim 1 km under water, *i.e.*,496 ± 30 (478 ± 35 for AFS and 514 ± 48 for NFS, p = 0.418). Traveling at the surface of the water used fewer flipper strokes than traveling the same distance underwater while diving. There was also no significant inter-species difference in mass corrected number of flipper strokes

The average flipper stroke amplitude (indicative of the average swimming effort) while diving was  $0.61 \pm 0.03 \text{ m/s}^2 (0.75 \pm 0.02 \text{ m/s}^2 \text{ for AFS greater than } 0.46 \pm 0.01 \text{ m/s}^2$  for NFS,  $p < 7.10^{-12}$ ). This was overall greater (all p > 0.05) than the average swimming effort while transiting at the surface  $0.45 \pm 0.01 \text{ m/s}^2 (0.48 \pm 0.01 \text{ m/s}^2 \text{ for AFS greater than } 0.43 \pm 0.01 \text{ m/s}^2$  for NFS, p < 0.007). This translated into a cumulative swimming effort per min of diving (sum of flipper stroke intensities,  $16.4 \pm 1.6 \text{ m/s}^2$  per min for all seals, or 22.6  $\pm 1.5 \text{ m/s}^2/\text{min}$  for AFS and  $10.26 \pm 1.0 \text{ m/s}^2/\text{min}$  for NFS,  $p < 10^{-4}$ ) greater than per min of

transiting at the surface  $(11.9 \pm 1.0 \text{ m/s}^2 \text{ per min for all seals, or } 15.0 \pm 1.1 \text{ m/s}^2/\text{min for AFS}$ greater than  $8.7 \pm 0.5 \text{ m/s}^2/\text{min for NFS}$ ,  $p < 10^{-3}$ ).

The number of flipper strokes during diving was a good indicator of energy expenditure while diving ( $R^2 = 0.76$ , AIC = 46.4,  $p < 2 \ 10^{-10}$ , Figure 3.3) according to:

3.1 
$$\text{EE}_{\text{Dive}} \sim (0.16 \pm 0.22) + (3.79 \ 10^{-5} \pm 3.96 \ 10^{-6}) \times \text{Stroke.Count}_{\text{Dive}} + (0.30 \pm 0.17 \text{ for NFS only})$$

•

where EE is the energy spent while diving (in MJ/kg) and  $Stroke.Count_{Dive}$  is the total number of flipper strokes detected during diving.

Table 3.1: Morphometric, metabolic and foraging parameters for northern and Antarctic fur seals during a single foraging trip at sea. Values are mean  $\pm$  SE and \* show between species significant differences (p < 0.05).

Parameters	Northern fur se	als	Antarctic fur seals		
Mass (kg)	$37.9 \pm 1.3$		$31.0 \pm 0.8$		
Fore flipper length (cm)	$39.4 \pm 0.9*$		$34.3 \pm 0.3*$		
Fore flipper width (cm)	$12.5 \pm 0.2*$		$11.1 \pm 0.1^*$		
Trip duration (d)	$7.96 \pm 2.17$		$7.65 \pm 3.88$		
Distance traveled (km)	$750 \pm 50$		635 ± 77		
Number of dives	2551 ± 323*		3949 ± 597*		
Dive depth (m)	19.4 ± 2.4		19.9 ± 2.7		
Dive duration (s)	62.8 ± 7.3*		42.6 ± 4.5*		
Time spend diving (%)	$28.6 \pm 2.0$		$29.0 \pm 0.7$		
Time spent transiting (%)	$30.5 \pm 1.8$		$26.4 \pm 1.6$		
Time spend diving (h)	$53.9 \pm 4.7$		$47.7 \pm 6.0$		
Time spent transiting (h)	$59.9 \pm 5.9$		$45.9 \pm 7.7$		
At-sea metabolic rate (MJ/d)	20.9 ± 1.5		17.7 ± 1.1		
At-sea metabolic rate (MJ/d/kg)	$0.56\pm0.04$		$0.59 \pm 0.04$		
Total energy expenditure (MJ)	155.1 ± 13.0		121.4 ± 17.1		
Total energy expenditure (MJ/kg)	$4.2 \pm 0.4$		$4.1 \pm 0.6$		
	Diving	Transit	Diving	Transit	
Energy expenditure (MJ)	69.28 ± 24.22*	46.20 ± 18.37	61.34 ± 30.78*	35.39 ± 23.74	
Energy expenditure (MJ/kg)	$1.89 \pm 0.81$ $1.24 \pm 0.51$		2.00 ± 1.03	1.15 ± 0.75	
Flipper stroke Frequency (Hz)	0.37 ± 0.03 0.35 ± 0.02		0.44 ± 0.06	0.53 ± 0.03*	
Average swimming effort per stroke (m/s <sup>2</sup> )	0.46 ± 0.01	$0.43 \pm 0.01$	0.75 ± 0.02*	$0.49 \pm 0.01$	
Cumulative swimming effort (m/s <sup>2</sup> /min)	10.26 ± 1.0	8.70 ± 0.52	22.60± 1.51*	15.01 ± 1.13	



Figure 3.3: Relationships between number of flipper strokes during diving or while transiting at the surface (speed  $\geq 1$ m/s) and activity-specific energy expenditure (in MJ) for northern fur seals (orange triangles, n = 16) and Antarctic fur seals (green squares n = 16).Plain lines show the results of linear models that treated species as an independent variable, and dotted lines are the 95% CI on the predicted values. Results from linear models while diving are:  $EE_{Dive} = (0.15 \pm 0.22) + (3.8.10^{-5} \pm 4.10^{-6}) \times \text{Stroke.Count}_{Dive} + (0.29 \pm 0.17 \text{ for NFS only}) (R^2 = 0.76, slope p < 2.10^{-10}) and while transiting at the surface: <math>EE_{Transit} = 0.08 \pm 0.18) + (4.10^{-6} \pm 6.10^{-7}) \times \text{Stroke.Count}_{Transit} + (0.28 \pm 0.14 \text{ for NFS only}) (R^2 = 0.63, slope p < 1.10^{-7}). Activity-specific cumulative swimming effort was also related to activity-specific energy expenditure whether while diving: <math>EE_{Dive} = (0.34 \pm 0.26) + (4.10^{-5} \pm 6.10^{-6}) \times \text{Swim.Effort}_{Dive} + [(0.17 \pm 0.45) + (3.10^{-5} \pm 2.10^{-5}) \times \text{Swim.Effort}_{Dive}$  for NFS only] (R<sup>2</sup> = 0.72, AIC = 53, slope p < 5.10^{-8}) or while transiting:  $EE_{Transit} = (0.03 \pm 0.17) + (9.10^{-6} \pm 1.10^{-6}) \times \text{Swim.Effort}_{Transit} + (0.42 \pm 0.14 \text{ for NFS} only)$  (R<sup>2</sup> = 0.67, AIC = 33, slope p < 3.10^{-8}).

Based on this this, the energetic cost per flipper stroke at depth was  $3.79 \pm 0.39$  J/kg/stroke for all seals ( $3.77 \pm 0.47$  J/kg/stroke for AFS and  $3.84 \pm 0.91$  J/kg/stroke for NFS similar between the 2 species, p = 0.74).

The relationship was also significant during transiting time at the surface, but slightly less accurate ( $R^2 = 0.63$ , AIC = 36.4, slope  $p < 10^{-7}$ , Figure 3.3):

3.2 
$$\text{EE}_{\text{Transit}} \sim (0.08 \pm 0.18) + (4.28 \ 10^{-6} \pm 6.07 \ 10^{-7}) \times \text{Stroke.Count}_{\text{Transit}} + (0.28 \pm 0.14 \text{ for NFS only}).$$

Finally, activity-specific cumulative swimming effort (from stroke amplitude) correlated to activity-specific energy expenditure while diving and transiting ( $R^2 = 0.66 - 0.72$ ). Interestingly, the change in swimming effort while diving impacted energy expenditure of northern fur seals more rapidly than Antarctic fur seals. However, this was not the case while transiting at the surface (Figure 3.3).

3.3 
$$EE_{Dive} \sim (0.34 \pm 0.26) + (4.27 \ 10^{-5} \pm 5.78 \ 10^{-6}) \times Swim.Effort_{Dive}$$
  
+ [(0.17 ± 0.45) + (3.64 \ 10^{-5} ± 2.03 \ 10^{-5}) × Swim.Effort\_{Dive} for NFS  
only)]  $R^2 = 0.72$ , AIC = 53.6, slope  $p < 5.10^{-8}$ 

3.4 
$$\text{EE}_{\text{Transit}} \sim (0.03 \pm 0.17) + (9.12 \ 10^{-6} \pm 1.20 \ 10^{-6}) \times \text{Swim.Effort}_{\text{Transit}} + (0.42 \pm 0.14 \text{ for NFS only})$$
  
R<sup>2</sup> = 0.67, AIC = 33.3, slope *p* < 3.10<sup>-8</sup>)

where  $EE_{Dive}$  and  $EE_{Transit}$  are energy spent in MJ/kg, and Swim.Effort<sub>Dive</sub> and Swim.Effort<sub>Transit</sub> are the cumulative swimming effort in m/s<sup>2</sup> during dive or transiting.

Over the entire foraging trip, the sum of all flipper strokes showed limited relation to total energy expenditure at sea ( $R^2 = 0.5$ , Figure 3.4). The best models predicting total energy expenditures are summarized in Table 3.2 and Figure 3.4, and involve primarily swimming effort while diving and transiting. The most parsimonious model ( $R^2 = 0.63$ , AIC = 88.2) was:

3.5. EE ~ 
$$(0.16 \pm 0.69) + (2.52 \ 10^{-5} \pm 4.13 \ 10^{-6}) \times \text{Swim.Effort}_{\text{Total}} + (1.33 \pm 0.54 \text{ for NFS})$$

where EE is the total energy spent during foraging trip (in MJ/kg), Swim.EffortTotal is the cumulative swimming effort while diving and while transiting at the surface (sum of Swim.Effort<sub>Dive</sub> and Swim.Effort<sub>Transit</sub>).

Table 3.2: Best models explaining at-sea energy expenditure in MJ/kg of northern (n=12) and/or Antarctic fur seals (n=13) as a function of flipper stroke count or of swimming effort during diving + transiting (Stroke.Count<sub>Total</sub> and Swim.Effort<sub>Total</sub>), or while diving or transiting at the surface (Total.Effort<sub>Dive</sub>, Total.Effort<sub>Transit</sub>, Swim.Effort<sub>Dive</sub> and Swim.Effort<sub>Transit</sub> respectively). Only models with  $\Delta$ AICc < 2 are presented and the model with only stroke count during diving + transiting.

Model	Parameters	Estimates	SE	р	$\mathbf{R}^2$	ΔAICc	AICc weight
3.5	Intercept	0.16	0.69	0.820	0.63	0.00	0.185
	$Swim.Effort_{Total}$	2.5 10 <sup>-5</sup>	4.1 10-6	<10 <sup>-5</sup>			
	Species (NFS)	1.33	0.54	0.022			
3.6	Intercept	0.15	0.67	0.824	0.67	0.48	0.146
	$Swim.Effort_{Total}$	5.0 10 <sup>-5</sup>	1.7 10 <sup>-5</sup>	0.007			
	Stroke.Count <sub>Total</sub>	-1.4 10 <sup>-5</sup>	8.9 10 <sup>-6</sup>	0.138			
	Species (NFS)	1.97	0.67	0.008			
3.7	Intercept	-0.20	0.71	0.775	0.66	0.82	0.123
	Stroke.Count <sub>Dive</sub>	4.3 10 <sup>-5</sup>	1.5 10 <sup>-5</sup>	0.008			
	Swim.Effort <sub>Transit</sub>	1.9 10 <sup>-5</sup>	5.7 10 <sup>-6</sup>	0.003			
	Species (NFS)	0.96	0.51	0.075			
3.8	Intercept	0.46	0.65	0.487	0.61	1.52	0.087
	Stroke.Count <sub>Dive</sub>	4.2 10 <sup>-5</sup>	1.5 10 <sup>-5</sup>	0.011			
	$Swim.Effort_{Transit}$	1.7 10 <sup>-5</sup>	5.9 10 <sup>-6</sup>	0.008			
3.9	Intercept	-0.02	0.71	0.920	0.65	1.78	0.076
	Stroke.Count <sub>Transit</sub>	7.7 10-6	2.7 10-6	0.010			
	Swim.Effort <sub>Dive</sub>	6.0 10 <sup>-5</sup>	1.7 10 <sup>-5</sup>	0.002			
	Species (NFS)	1.53	0.58	0.016			
3.10	Intercept	0.95	0.67		0.50	4.63	0.045
	Stroke.Count <sub>Total</sub>	1.2 10 <sup>-5</sup>	$2.510^{-6}$				

Parameter estimates of other models are detailed in Table 3.2. It is interesting to note that only models involving swimming effort (*i.e.*,cumulative stroke amplitudes) showed a significant difference in the relationships between fur seal species, the ones involving flipper stroke count did not. Finally, rate of energy expenditure (in MJ/d) was not accurately predicted by flipper stroke counts or amplitudes, whether as rates per day or as cumulated over time (all  $R^2 < 0.05$ ).



Figure 3.4: Relationship between energy expenditure (mass-corrected MJ/kg) and cumulative swimming effort (A: model 3.5 from Table 3.2,  $R^2 = 0.63$ ) or total number of flipper stroke during time at sea (B: model 3.10 from Table 3.2,  $R^2 = 0.50$ ) for female northern fur seals (n = 12, orange triangles) and Antarctic fur seals (n = 13, green squares). Plain lines are results of the general linear model and dotted lines are the 95% CI on the predicted values.

Cost of transport (COT) calculated using the slope of the relationship between diving energy expenditure and vertical distance traveled was  $1.63 \pm 0.18$  J/kg/m ( $p < 3.10^{-10}$ , R<sup>2</sup> = 0.71) in northern and Antarctic fur seals (not significantly different between species, p > 0.10). I also calculated a cost of transportation of  $1.88 \pm 0.15$  J/kg/m from the cost per stroke (*i.e.*, 3.79 J/kg/stroke) and the number of strokes necessary to travel one meter for each seal individually (not different between species, p = 0.55).

## 3.5 Discussion

Flipper strokes or tail beats are generally believed to be good proxies for the cost of transportation and metabolic rates in marine mammals and fish (Williams et al., 2004; Gleiss et al., 2009; Wilson et al., 2013). In the case of fur seals, I found that flipper stroke count was, indeed, a good predictor of energy expenditure during diving, and to a lesser extent during transiting. However, flipper stroke count was a poor predictor overall of total energy

spent during a full foraging trip at sea for fur seals ( $R^2 = 0.50$ , Figure 3.4). In this case, cumulative amplitude of flipper strokes represented total energy expenditure better than flipper stroke counts.

### 3.5.1 Underwater locomotion

Detecting flipper stroke count and relative intensity was easy to resolve when animals were diving (Figure 3.1). Numbers of flipper strokes accurately predicted energy spent while diving in both species of fur seals ( $R^2 = 0.76$ , Figure 3.3), although energy spent while diving were not measured directly but predicted from total energy expenditure and time activity budget (Chapter 2). This is nevertheless consistent with diving Weddell seals which show a good correlation between flipper stroke count and energy expenditure measured from respirometry ( $R^2 = 0.87$ ), post-dive at sea ice breathing holes (Williams et al., 2004). In addition, each Weddell seal hind flipper stroke had a predictable impact on oxygen consumption and cost 2.39 J/kg/stroke. However, their stroke patterns only took into account lateral movements from one side or from the other side. This means that total side-to-side stroking cost would be 4.78 J/kg/stroke. This falls within the range of values for other phocid seals, from 2.88 J/kg/stroke in harp seals to 5.74 J/kg/stroke for harbour seals (Davis et al., 1985; Fish et al., 1988). For fur seals, I found the cost per stroke was also within this rage at  $3.79 \pm 0.39$  J/kg/stroke (with no significant difference between NFS and AFS) and similar to the 3-4 J/kg/stroke values found for northern elephant seals (Maresh et al., 2014). However, it was lower than the 7.33 J/kg/stroke found for another otariid, the California sea lion (range 1.93 - 23.16, from captive experiments, Dassis et al., 2012). These values show that energy expenditure per stroke differs by a factor of 2.5 among species, and by ~ 1.8 between otariids (*i.e.*, between the California sea lions and the study fur seals).

Buoyancy, dive angle, dive depth and mode of locomotion all affect the cost of transport in marine predators (Williams et al., 2000; Sato et al., 2003; Aoki et al., 2011; Richard et al., 2014; Jouma'a et al., 2015). Phocid seals mostly swim using their hind flippers in a pseudo-axial locomotion, while otariids use their fore flippers in an 'underwater flight' locomotion (Kuhn and Frey, 2012). The wide difference in these two modes likely results in different stroke power and efficiencies. In my case, northern fur seals and Antarctic fur seals

both showed similar flipper stroke rates and numbers of strokes to cover a given distance while diving, or while transiting at the surface, which supports the idea that similar stroking patterns have similar efficiencies.

Buoyancy in diving elephant seals has been shown to change stroking patterns and the numbers of strokes required to cover a given distance (Adachi et al., 2014). This in turn affects locomotor costs in different ways depending on seal density and whether the elephant seal descends or ascends with the cost of transport increasing when departing from neutral buoyancy (Miller et al., 2012). The northern and Antarctic fur seals in my study used their flippers at faster rates during descent than ascent (visually assessed but not reported here). This is consistent with another study of northern fur seals (Insley, 2008). As depth increased, less stroking or less energy was necessary to maintain a constant swimming speed. Locomotion costs are usually reduced with neutral buoyancy, which can be attained at different pressures or depths depending on body fat, body size, quantity of internal gas, or to a lesser extent seawater temperature or salinity. Otariids have a lower ratio of fat to lean mass than do phocid seals. Otariids also dive with fully inflated lungs and with air trapped in their fur, while phocid seals exhale before diving and are devoid of thick fur. In addition, otariids are shallower divers on average compared to phocid seals, and might not dive deep enough to attain neutral buoyancy as do elephant seals. Thus, differences in buoyancy could explain, at least partially, the difference in cost of stroking between taxa.

I estimated the mechanical power that fur seals generated while swimming by measuring the relative magnitude of dynamic acceleration output during a flipper stroke in  $m/s^2$ , which can vary depending on speed, drag, chasing behaviour, and other factors. It showed that the intensity of the average flipper stroke acceleration (flipper stroke amplitude or swimming effort) was greater for Antarctic fur seals while diving (0.75 ± 0.02 m/s<sup>2</sup>) compared to northern fur seals (0.46 ± 0.01 m/s<sup>2</sup>, Table 3.1). Since both species of fur seals had similar energy costs per stroke, it indicates that measured acceleration was intrinsically greater per joule spent for Antarctic fur seals to swim at depth than for northern fur seals. Another possible explanation for the difference is that underwater behaviours of Antarctic fur seals were such that they might have required a different filtering of the dynamic acceleration compared to northern fur seals.

## 3.5.2 Full foraging trip locomotion

Most studies trying to link energy expenditure to cost of locomotion through flipper stroke rates in the wild have done so for species that spend most of their time underwater (Williams et al., 2004; Maresh et al., 2014). However, fur seals spend ~70% of their at-sea time transiting at the surface and performing surface activities (grooming, slow movements), or resting (Chapter 2). Quantifying flipper stroke rates for fur seals at the surface was not as straightforward as counting them at depth due to the environmental acceleration 'noise' from wave movements at the surface that confounded interpretation of the dynamic acceleration signal. Another confounding factor is that fur seals tend to porpoise while transiting at the surface and water, which have different densities that likely affect acceleration signals compared to underwater flipper stroke. It is, therefore, not surprising that flipper stroke count did not correlate as well with energy expenditure while transiting (estimated indirectly from a best fit model using Eq. 2.1 in Chapter 2) as it did while the seals were diving ( $R^2 = 0.63$ , Figure 3.3).

Fur seals transiting at the surface usually porpoise or travel at depths where wave drag is the lowest to reduce energy waste (3 body diameters below the surface, Hindle et al., 2010). These possibilities for reducing the cost of transportation are not available at depth, which means that energy spent per stroke likely differs between dive time and surface transiting. Unfortunately, the change of mode of locomotion and the residual parasitic noise of the surface interface prevented us from estimating a biologically accurate cost per stroke while transiting. However, it is clear from the slopes in **Figure 3.3** that it costs fur seals less to transit than it does to dive, which is opposite to what was found in another Antarctic fur seal study based on diving behaviour only (Arnould et al., 1996).

Diving and transiting only accounted for ~ 60% of total time at sea in fur seals. This means animals spent ~ 40% of their time performing other activities (surface activities, slow movements, grooming, sleeping). Whether animals do not stroke, or whether I could not detect it amongst environmental movements is unclear. In any case, most cost of transportation was spent during diving and transiting at sea. Resting and surface activities do not involve as much locomotion and are by definition less energetically expensive (Chapter 2). In the end, is it still possible to use flipper stroke metrics to estimate total energy

expenditure at sea despite the significant proportion of time not accounted for in my measures of flipper strokes and the uncertainties inherent to flipper stroke detection at the surface?

The best models to predict the total energy expended by the fur seals during their foraging trips all included swimming effort during diving and transiting (*i.e.*, relative intensity of acceleration during stoking), rather than flipper stroke count. Flipper stroke rate better reflected energy expenditure at the activity level while flipper stroke intensities performed better at the foraging trip level. Since the relationships between energy spent and flipper stroke differ between dive and surface transportation, the proportion of time that individuals allocate to these activities will result in different amounts of energy being spent over the full foraging trip. It, therefore, is logical that the acceleration intensities of flipper strokes (which provide an index of swimming effort during each type of activity) rather than stroke counts correlate better with energy expended over the entire foraging trip. Acceleration intensities of flipper strokes better accounts for surface activity, i.e. ~40% of total time not accounted for by flipper strokes. However, the overall accuracy of flipper stroke intensity as a predictor for total energy expenditure at sea remains limited ( $\mathbb{R}^2 = 0.63$ ).

Insley (2008) suggested that fur seals spend 70% of their time swimming (whether at depth or at the surface), and that multiplying the cost per stroke at depth by an average stroking frequency was a good approximation for energy spent at sea. However, this assumes that costs of transport at depth and at the surface are similar, which was not the case in the present study. For example, two fur seals could spend 70 % of their time swimming at sea, but allocate their swimming times differently (25 % dive time and 45 % transit time versus 45 % dive time and 25 % transit time). In my case, the northern and Antarctic fur seals spent 20 – 46 % of their time diving and 15 - 47 % of their time transiting. Time-activity budgets are, thus, important considerations when trying to estimate the cost of foraging given how different activities have different associated costs (Chapter 2).

## 3.5.3 Flipper stroke cost versus cost of transportation

The cost of transport (COT) in J/kg/m has been hypothesized to be consistent between marine mammals and to scale with body mass according to the equation COT =  $7.79 \times \text{mass}^{-0.29}$  (Williams, 1999). I calculated the underwater COT to be ~ 1.6 - 1.9 J/kg/m in northern and Antarctic fur seals based on the slope of the relationship between diving energy expenditure and vertical distance traveled or the price of stroke and the number of strokes necessary to travel one meter. These values are somewhat lower than the theoretical value from Williams' equation (1999) of 2.7 J/kg/m for an average female fur seal mass. There is a ~30% discrepancy between the measured (1.9 J/kg/m) and calculated (2.7 J/kg/m) COT for my fur seals. Based on a cost per stroke ~ 3 - 4 J/kg/stroke (Maresh et al., 2014) and the 0.4 strokes on average required to cover 1 m (0.35-0.45 strokes over short foraging, Adachi et al., 2014), elephant seals also have a COT (~1.2 - 1.4 J/kg/m) 15 - 22% lower than the 1.55 J/kg/m theoretical value for a 250 kg female (1.55 J/kg/m from Williams (1999) equation).

It is interesting to note that the only otariid (California sea lion) included in the dataset used by Williams (1999) to derive the theoretical COT equation for all marine mammals also had a lower measured COT values compared to the predicted value (open circles in Figure 1 in Williams (1999)). The 2 extreme ranges of body mass estimates between pinnipeds (15 – 100 kg) and cetaceans (2000 – 10000 kg) might have driven the regression equation between the 2 clusters on their figure. Pinniped values on their own do not seem to follow a clear trend (Figure 1 in Williams (1999)). It could therefore be the case that the theoretical COT equation overestimates the COT of fur seals. There are also inherent errors in the detection of flipper strokes at sea from acceleration (although they are minimized during diving), and in calculated flipper stroke rate per distance that might have accounted for the difference between measured and calculated COT in fur seals.

# 3.5.4 Comparison of flipper stroke vs VeDBA as an index of total energy spent at sea

Swimming effort derived from the intensities of flipper strokes is one way to measure dynamic body acceleration associated with locomotion and estimate energy expenditure. ODBA or VeDBA are other acceleration-based metrics that have been used to estimate energy expended by different species while foraging (Fahlman et al., 2008; Halsey et al., 2008; Gleiss et al., 2009; Halsey and White, 2010; Gomez-Laich et al., 2011; Halsey et al., 2011a). Overall, ODBA or VeDBA were found to be better proxies for energy expenditure than flipper stroke metrics during specific types of activities (analyses on same individuals,  $R^2 > 0.90$  and Figure 2.3 of Chapter 2). Flipper strokes only reflect one specific type of movement, while active swimming and foraging usually involves many other body motions (undulations, etc.). In addition, flipper strokes are only detected over 2 body axes (up/down and forward/backward) and are not reflected on the third axis, while VeDBA is a measure of acceleration over the 3 body axes. Finally, using tri-axial acceleration takes into account the types of surface movements that flipper strokes could not (~ 40% of the activity budget). It is, thus, not surprising that VeDBA would correlate better with energy expenditure either at the activity specific level or at the full foraging trip timescale, as it considers more body movements than just flipper strokes. Consequently, VeDBA are better predictors of total energy spent at sea than are metrics of flipper strokes, especially at the activity-specific level.

#### 3.5.5 Conclusion

Flipper stroke rate is an appropriate means to determine the energy spent by fur seals while swimming underwater (*i.e.*, while diving) as has been shown for other seals (Williams et al., 2004; Dassis et al., 2012; Maresh et al., 2014). However, in the case of otariids that spend a greater proportion of their time at the surface compared with phocids, flipper stroke count or flipper stroke intensities are not the best predictors of energy expenditure because of the time fur seals spend on the surface. Activity-specific ODBA/VeDBA is a more accurate means for predicting energy expenditure than are flipper stroke count or swimming effort (Chapter 2). While morphology and gaits have undoubtedly evolved to maximise locomotion efficiencies in the ocean, there are sufficient behavioral differences between species to confound application of a simple metric of movement that can be applied universally. Rather, foraging ecology studies need to consider the activity-specific metabolic rates and time-activity behavioural strategies that ultimately determine the foraging costs of individuals.

Chapter 4: Combined hard-parts and DNA analyses of scats, and stable isotope analysis of blood from lactating females reveal two distinct diets and foraging strategies for northern fur seals breeding in the Bering Sea

## 4.1 Summary

The key to understanding why the Pribilof population of northern fur seals (Callorhinus ursinus) has been declining may lie in what, where and how lactating females are eating in the Bering Sea. Unfortunately, it is unclear whether fur seal diet determined from hard-part remains in feces (e.g. fish bones and otoliths, and cephalopod beaks) collected on breeding beaches are representative of what the Pribilof population consumes each year. I collected 98 scats from a breeding beach (Reef rookery on St. Paul Island), and captured and tracked 20 lactating female fur seals. Foraging habitats and trophic levels of the tracked females were assessed using the stable isotope method on their blood. Prey were identified from scats using hard-part remains and DNA extracted from their soft matrix for comparison. DNA yielded a higher resolution of prey species and revealed a greater diversity of prey than previously recognized. Collectively, cluster analyses of prey present in individual scats indicated two basic prey associations indicative of two different foraging habitats corresponding to neritic and oceanic waters, *i.e.*, pollock, squid and mesopelagic fish (mostly deep water basin associated prey), and a second diet of pollock, salmon, hexagrammid, flatfish, and forage fish (continental shelf types of prey). These results are consistent with stable isotopes analyses in plasma that show two clusters of  $\delta^{13}C$  and  $\delta^{15}N$  values corresponding with whether the tracked fur seals fed over the shelf or off the shelf. In all likelihood, the scats collected on the Pribilof rookeries are representative of diets consumed in the preceding 12-24 hours and are biased toward shelf prey species given the location of the breeding islands on the shelf. These results show that combining clustering analysis of co-occurring prey species from results of hard-part- and DNA-based methods, C and N stable isotopes, and at-sea tracking of individuals can be used to adjust diet summaries and properly reflect the average diet of fur seals.

## 4.2 Introduction

Most of the world population of northern fur seals once bred on the Pribilof Islands in the Bering Sea, but this population has declined by ~6% per year for the last 30 years with no sign of slowing down (Towell et al., 2006). The decline coincided with a major environmental change that altered the distribution and relative abundances of different species of fish in the Bering Sea (Anderson et al., 1997; Anderson and Piatt, 1999). The decline also coincided with an increase in the dominance of walleye pollock and a decrease in squid in the fur seal diet (Sinclair et al., 2008). Such dietary changes can affect the foraging efficiency of lactating females and their capacity to successfully reproduce. Assessing diets is thus important for understanding the balance between energy input from food, and the energy spent foraging.

Diets of northern fur seals have been assessed in recent years by identifying the undigested hard-part remains of prey found in feces (scats) or to a lesser extent in spews (Perez and Bigg, 1981; Perez and Bigg, 1986; Antonelis et al., 1990). Taxon, size and relative number of prey ingested can be determined from hard-parts (Tollit et al., 2003), but the method is biased by differential digestion or retention rates of prey parts depending on species and sizes of prey consumed, and on species, sex, activity level, GI tract morphology and meal size of predator that lead to under- or over-estimation of prey occurrence (or size of prey consumed) in the diets (Bigg and Fawcett, 1985; Jobling and Breiby, 1986; Harvey and Antonelis, 1994; Tollit et al., 1997; Bowen, 2000; Arim and Naya, 2003; Gudmundson et al., 2006). In addition, assumptions underlying how the presence-absence based methods determines relative proportion of each species consumed are often not met (Olesiuk et al., 1990).

More recently, methodological advancements have increased interest in molecularbased tools that overcome some of the limitations of hard-part analyses (Deagle et al., 2005; Casper et al., 2007a; Deagle and Tollit, 2007; Deagle et al., 2009; Deagle et al., 2010; Cristescu, 2014; Thomas et al., 2014). Among them, the DNA meta-barcoding method allows species or groups of organisms to be identified within a sample. This molecular method relies on high-throughput DNA sequencing to identify prey species by detecting DNA remains present in the soft organic matter of the feces. It is non-invasive, has the advantages of being quick, relatively inexpensive, and is more sensitive than the hard-part remains methods (Deagle et al., 2009). This new DNA approach is still being evaluated, however, and has its own inherent biases at the technical and interpretation levels (Casper et al., 2007a; Deagle et al., 2010; Thomas et al., 2014).

Scat samples have proven to be a useful means to estimate diet composition of pinnipeds. However, scats only represent what animals recently consumed (~ 48h, Orr and Harvey, 2001; Staniland, 2002; Deagle et al., 2005; Casper et al., 2007a). Stable isotope concentrations in blood, hair and bone synthesize dietary information over longer timescales;  $\delta^{15}$ N provide insights into the trophic level at which the consumer has fed (Wada et al., 1991; Hobson and Welch, 1992) while  $\delta^{13}$ C indicates the foraging habitat based on the isotopic composition of primary producers at the base of the food chain (France, 1995; Burton and Koch, 1999). In the marine environment,  $\delta^{13}$ C values allow characterizing inshore versus offshore, or pelagic versus benthic consumers (Hobson et al., 1994; Cherel et al., 2007). Pairing high-resolution telemetric data that provide movement and diving information with their respective stable isotope analyses can provide a better understanding of foraging strategies and diets, and disentangle spatial, trophic and physiological differences between individuals (Newsome et al., 2010).

The Bering Sea where northern fur seals forage during the breeding season is composed of a shallow continental shelf separated from a deep oceanic basin by a shelf break. These differences in bathymetry and hydrographic structures define the prey associations within the ecosystems. Northern fur seals feed widely in the Bering Sea, but appear to have foraging areas that depend upon the location of the breeding beaches (Goebel, 2002; Robson et al., 2004; Kuhn et al., 2014). These foraging areas determine the northern fur seal's foraging behaviours, dive patterns and ultimately their diets (Sinclair et al., 1994; Antonelis et al., 1997; Gudmundson et al., 2003) given yearly environmental conditions (Gentry and Kooyman, 1986). It is thus important to assess diet composition and foraging behaviours of individual female fur seals for a given year to better understand foraging efficiency and reproduction success within specific environmental conditions. My primary goal was to evaluate the limitations of previous diet determinations for northern fur seals on the Pribilof Islands, by combing DNA and hard-part analyses of scat samples, thereby creating enhanced fur seal diet data. I also sought to provide context for dietary analysis of the scats using the isotopic signatures of blood samples and the foraging habitats for 20 lactating females that tended to pups on the same rookery where the scats were collected. I thus sought to assess the reliability of current descriptions of fur seal diets and provide insights into the implications of having different diets and ways in which the diet estimates might be corrected to better reflect the average diet of northern fur seals breeding on the Pribilof Islands.

#### 4.3 Material and methods

#### 4.3.1 Data collection

All scats and blood samples were obtained from Reef Rookery on St. Paul Island (Bering Sea, 57° 6 'N - 170°17' W, Aug - Sep 2011) under the US NMFS permit #14329-01 and the UBC animal care permit #A10-0364. Isotopic data were collected on 20, mature, full-sized lactating northern fur seals with a confirmed suckling pup. Standard morphometric measurements were made to the nearest 0.5cm, and mass was recorded using scale at  $\pm 0.2$ kg. A blood sample was collected while the animals were anesthetized from the back flipper in Monovette syringes (Sarstedt) coated with Li-Heparin, and plasma was isolated from red blood cells by centrifugation at 1000 g for 10 min. Each female was equipped with a fastloc GPS datalogger (MK10, Wildlife Computers Inc.) glued to their fur roughly 8cm below their shoulder blades. The data loggers also recorded depth at 1 Hz, and were used to identify foraging locations and provide diving data to determine whether dives were benthic or pelagic (Tremblay and Cherel, 2000; Schreer et al., 2001). Upon recovery from anaesthesia, animals were allowed to return to the colony and to go on a single foraging trip before recapture upon their return on land. A second blood sample was taken following the same procedures and the GPS logger was retrieved by cutting the hair. Blood samples were frozen at -20 °C until analyses in the lab.

## 4.3.2 Morphological identification from hard-parts in scats and prey size

I collected 98 scats on Reef rookery in September 2011, above the ~ 60 scats required to reach the maximum prey diversity (Zeppelin and Orr, 2010). Each scat was placed in a zip-lock plastic bag and was frozen at -20 °C until processing at the UBC Marine Mammal Research Unit lab in Vancouver, Canada. Scats were then thawed and transferred into a nylon mesh paint strainer lining a 500 ml disposable container. 95 % Ethanol was added to the samples, and scat matrix was separated from the hard prey remains by manual homogenization. The hard-parts contained in the mesh bag were removed for analysis and cleaning procedures following a standard protocol (Trites and Calkins, 2008). Cleaned and dried hard-parts (i.e., otoliths, bones, and lenses from teleost fishes; cartilage from cartilaginous fishes; and beaks, pens, and statoliths from cephalopods) were sorted out and identified to lowest possible taxon by Pacific IDentifications Inc. (Victoria, BC) using a reference collection of prey species skeletons. Types of hard-part present, the species from which they came, the estimated size of the prey by species, and the minimum number of individuals present (all species combined based on maximum paired numbers of eye lenses or other unique identifiable structures) were recorded. Unknown prey items were categorized as "unidentified" and were excluded from analyses. The relative diet composition was determined by first calculating the frequency of occurrence (FO) of a specific prey item as the percentage of scats that contained it among all the scats collected according to the following equation:

$$FO(\%) = \frac{\sum_{i}^{n} O_{pi}}{n} \times 100$$

where  $O_{pi}$  is the absence (0) or presence (1) of prey *p* in sample *i* from total number of scats *n*. Second, the proportion of a specific prey to the overall diet (adding up to 100%) was estimated using the split-sample frequency of occurrence SSFO (Olesiuk et al., 1990; Olesiuk, 1993).

## 4.3.3 Prey species identification from DNA sequences analyses

Ethanol preserved scat organic matter (matrix) was subsampled from thawed scats and used for genetic analyses. DNA extraction was performed following the protocol from Thomas et al. (2014) on approximately 20 mg of scat sediment material from the same scats analysed for hard-part remains. Extraction was done using QIAamp DNA Stool Mini kit (Qiagen) and eluted in 100  $\mu$ L elution buffer (10 mM Tris-Cl, 0.5 mM EDTA; pH 9.0).

The scat DNA metabarcoding protocol used for this study followed the procedures outlined in Thomas et al. (in review). In brief, a ~260bp fragment of the 16S mitochondrial ribosomal gene was amplified by PCR from each scat sample using the Qiagen Multiplex PCR kit. This reaction contained two different primer sets, which were developed to amplify a species-diagnostic region of 16s for both Chordates and Cephalopods (Deagle at al. 2009). Sample identification was done using a two-tiered barcoding approach, employing 96 uniquely labeled forward primer sets and the ligation of barcoded sequencing adapters to sample amplicon pools. Sequencing was done on an Illumina MiSeq using the MiSeq Reagent Kit v3 (600 cycle) for PE 300 bp reads, and the samples were run alongside 288 harbour seal scat DNA samples from an unrelated study.

Demultiplexing of DNA sequences was done with the MiSeq platform itself (separating pools of 96 samples by adapter sequence) and in the software package QIIME (further separating samples by forward primer tags) (Caporaso et al., 2010). Taxonomic identification of prey species DNA was done with on a nucleotide BLAST search for each sequence against a reference database of potential prey, using a minimum similarity threshold (Thomas et al. *in review*). To ensure that no major prey groups were missing from the existing database, I also applied a clustering approach with scat sequences, picking a representative of each cluster and applying a GenBank query to add any new prey species to my database that were not there initially (Altschul et al., 1990; Edgar, 2010). For details see Thomas et al. *in review*. DNA detections of prey species in samples were treated as occurrences (presence/absence), similar to the treatment of prey hard-parts information.

### 4.3.4 Combined results and prey associations

I combined the results from the hard-part and the DNA analyses by merging occurrence of unique prey species detected by either technique per individual scat. When a species of salmon was detected with DNA analysis and presence of unidentified salmon was detected from hard-parts within the same scat, I considered hard-part remains were coming from the specific species identified by DNA sequencing and did not duplicate occurrence of salmon. I did the same for cephalopod species, and obtained a combined presence-absence matrix of individual prey species per scat. Frequency of occurrence and split sampling frequency of occurrence in the diet from DNA analyses of DNA and hard-parts combined were calculated as above.

Association between prey groups within individual scats were identified and illustrated using a cluster analyses (*hclust* in *stats* package, in R 3.1.3). I first calculated the gamma coefficient (Goodman & Kruskal 1954) for each pair of prey groups from a presence-absence matrix of prey within the different scats. The distances between 2 prey groups were estimated from "1 – gamma" and were used as the dissimilarity matrix for the cluster analysis (with the "complete" method).

# 4.3.5 Stable isotopes

Stable isotope analyses were performed at the LIENSs laboratory (La Rochelle University, France). Red blood cells (collected before the seal's foraging trip) and plasma samples (collected before and after the seal's foraging trip) were freeze-dried and powdered. Blood plasma can contain high and/or variable lipid content in animals (Cherel et al., 2005), so I performed a lipid extraction using cyclohexane following the method used for seabirds (Kojadinovic et al., 2008). Tissue subsamples were weighed (~0.4 mg) with a microbalance, packed in tin containers, and nitrogen and carbon isotope ratios were determined by a continuous flow mass spectrometer (Micromass Isoprime) coupled with an elemental analyser (Euro Vector EA 3024). Stable isotope concentrations were expressed in conventional notation ( $\delta X = [Rsample/Rstandard) - 1] \times 1000$ ), where X is <sup>13</sup>C or <sup>15</sup>N and R is the corresponding ratio <sup>13</sup>C:<sup>12</sup>C or <sup>15</sup>N:<sup>14</sup>N. R standard is Vienna PeeDee Belemnite and atmospheric N2 for  $\delta^{13}$ C and  $\delta^{15}$ N respectively. Replicate measurements of internal laboratory standards (acetanilide) indicate measurement errors < 0.15 % for  $\delta^{13}$ C and < 0.20 % for  $\delta^{15}$ N. The units are expressed in parts per mil (%c).

## 4.4 Results

# 4.4.1 Foraging location and behaviour

The female northern fur seals weighed on average  $37.9 \pm 1.34$  kg prior to departure (range 30.8 - 55.6 kg, n = 20), and gained  $1.13 \pm 2.98$  kg after their single foraging trips (a  $3.5 \pm 1.8$  % gain in body mass). The average foraging trip was ~  $750 \pm 50$  km (range 391 - 1200 km) and lasted  $7.96 \pm 2.17$  d (range 4.26 - 12.03 d). The females travelled widely, but 12 of the 20 stayed on the Bering Sea shelf (hereafter called on-shelf females), while the remaining 8 went off the shelf into more pelagic waters (hereafter called off-shelf females, Figure 4.1). Females foraging off the shelf spent on average 2.5 more days at sea and traveled 200 km more than on-shelf females ( $9.7 \pm 1.8$  d and  $892 \pm 144$  km for off-shelf females versus  $7.1 \pm 2.4$  d and  $655 \pm 218$  km for on-shelf females).



Figure 4.1: Foraging tracks of 20 lactating northern fur seal females breeding on Reef rookery on St. Paul Island, and foraging in the Bering Sea, AK. Tracks are derived from fastloc GPS locations. Orange and yellow tracks show the females that stayed foraging on the shelf (n = 12) and the females that went foraging off the shelf (n = 8) respectively.



Figure 4.2: Number of prey species detected in scats depending on the analytical method used, DNA analyses in scat organic matter (in black, n = 98) or hard-part remains (in grey, n = 60). The difference in sample size is due to no identifiable hard-parts being recovered in 38 of the scats (collected on Reef rookery St. Paul Island, Bering Sea, Alaska in Aug-Sept 2011).

Females foraging on the shelf made a significant amount of benthic dives  $(15.2 \pm 2.6 \%)$  of all dives) and occasionally foraged during the day, while females foraging off the shelf rarely performed benthic dives  $(2.2 \pm 0.4)$  of all dives and 0% when off the shelf) and were specific nocturnal divers. Off-shelf females also performed significantly more foraging dives  $(2526 \pm 155)$  dives) than the on-shelf females  $(1648 \pm 2526)$  dives, p < 0.001, but the dives were on average shallower and lasted half the time  $(16 \pm 2)$  m deep and  $42 \pm 6$  s duration for off-shelf females and  $28 \pm 4$  m deep and  $84 \pm 10$  s duration for on-shelf females, p < 0.05.

## 4.4.2 Scats hard-part remains

At least 14 prey species from 11 taxonomic families were found in the scats collected on Reef rookery (including fish, cephalopods and worms). There were no identifiable hardparts (or unidentifiable hard-parts) in 38 of the 98 scats collected, so the hard-part analyses were done only with 60 scats. On average  $1.6 \pm 0.2$  species were detected per scat with the hard-part technique, and only one species was detected per scat in more than 70% of them (maximum 6 species detected in one scat).

Frequency of occurrence of different prey species (Figure 4.3) showed the most ubiquitous prey present in scats was walleye pollock (*Theragra chalcogramma*, in ~ 75% of scats), followed by salmon (in ~ 30% of scats), cephalopods (squids and octopus, in ~ 18% of scats) and Atka mackerel (in ~ 10% of the scats, *Pleurogrammus monopterygius*). The prey species were categorised in different taxonomic or ecological groups, according to their importance in the diet (Figure 4.3, Table 4.1). Frequency of occurrence of each of these groups were then converted into relative proportion of prey in the diet using the Split-Sample Frequency of Occurrence methods (Olesiuk, 1993). The main prey category in the diet of reef rookery fur seals was gadids (~69%), mostly composed of walleye pollock, followed by cephalopods (squids and octopus of unknown species, 12.6%), salmonids (6.4%) and hexagrammids (3.6%, mostly Atka mackerel). Flatfish, forage fish, mesopelagic fish, rockfish and others make the remaining 8.5% in the diet.

#### 4.4.3 Scat DNA analyses

Twenty-six species of prey were detected in all the scats collected (n = 98) from DNA analyses in the scat organic matter. The DNA method detected more prey per scat and more prey species overall, and was able to tease apart species of the same family better than the hard-parts method (such as the 5 species of salmon; Figure 4.3). On average,  $3.0 \pm 0.2$  species were detected per scat (max 9 species in a single scat), which is significantly greater than the 1.6 ± 0.2 species determined from the hard-parts technique (p > 0.05, Figure 4.2).



Figure 4.3: Proportion of prey species in the diet (split sample frequency of occurrence) of northern fur seals in scats collected on Reef Rookery, St. Paul Island, Bering Sea during the breeding season 2011 from hard-part remains analyses in yellow (n = 60), from DNA analysis in red (n = 98) and presence-absence results from both techniques combined in blue (n = 98, FO and SSFO values available in Table A.3 in the Appendices section).

Gadids (particularly walleye pollock) were the most ubiquitous prey found with this technique (~63% of the scats, Figure 4.3, Table 4.1). Overall, DNA analyses detected fewer cephalopods (0.5%) and more forage fish (10.6%), flatfish (5.2%) and salmon (~ 13%) than the hard-part analyses. Hexagrammids were detected within the same proportion between

methods. In the 38 scats in which no hard-parts were recovered, DNA analyses detected walleye pollock (n = 37), chum salmon (*Oncorhynchus keta*, n = 17), Pacific herring (*Clupea pallasii*, n = 16), Pacific hake (*Merluccius productus*, n = 13), Pacific cod (*Gadus microcephalus*, n = 10), Pacific halibut and kelp greenling (*Hippoglossus stenolepis* and *Hexagrammos decagrammus*, n = 6 each), among other less frequent prey species (n < 4).

## 4.4.4 Combined methods

There were 321 occurrences of prey in 98 scats with combined results from both methods, 225 of which were only detected with DNA analyses, 29 only with the hard-parts method, and 67 were simultaneously detected with both techniques. Given the overwhelming dominance of DNA identified species, the frequency of occurrence (FO) of prey were quite similar between the combined and the DNA only methods (Figure 4.3 and Table 4.1), with the exception of cephalopods. The same pattern was seen for the relative importance (SSFO) for all prey groups, except for gadids that were of slightly less importance in the population diet (~57 % combined compared to ~ 63 - 69% separately), and cephalopods that increased in importance to ~ 3.5% compared to DNA method only (which was still lower than estimated by the hard-part method).

	Hard-part remains		DNA analyses			Combined			
Prey group	FO (%)	SSFO (%)	Rank	FO (%)	SSFO (%)	Rank	FO (%)	SSFO (%)	Rank
Gadid	83.3	68.9	1	96.9	62.6	1	96.9	57.1	1
Cephalopod	20.0	12.6	2	2.0	0.5	8	14.3	3.5	7
Salmonid	15.0	6.4	3	39.8	12.8	2	42.9	13.2	2
Hexagrammid	13.3	3.6	5	19.4	4.2	5	23.5	4.9	5
Mesopelagic fish	5.0	2.6	6	4.1	1.1	7	5.1	1.4	8
Forage fish	5.0	1.5	7	34.7	10.6	3	35.7	10.4	3
Flatfish	1.7	0.3	8	20.4	5.2	4	19.4	5.2	4
Others	7.7	4.2	4	13.3	3.0	6	18.3	4.4	6

Table 4.1: Frequency of occurrence (FO) and split sampling frequency of occurrence (SSFO) of prey in scats collected on Reef rookery, St. Paul Island during the breeding season 2011 measured from hard-part remains and from DNA analyses in scats.

### **Combined methods**



Figure 4.4: Hierarchical cluster dendrogram of prey groups found in 98 northern fur seals scats from Reef rookery on St. Paul Island, Bering Sea, AK, in Aug-Sept 2011 identified from both DNA analyses of organic remains and of morphological identification of hard-parts methods combined.

## 4.4.5 Prey associations

Prey in fur seal diet were clustered in 3 distinct groups (*i.e.*, consistently found together and not randomly distributed amongst the foraging range of the fur seals). Based on cluster analyses from the combined DNA and hard-parts results (Figure 4.4), gadids made up one group on their own. A second group was made of mesopelagic fish and cephalopods. Finally, a third group included a mixture of forage fish, salmonids, flatfish, hexagrammids and others, even though within-group associations varied with the method used. The same clusters were found using results from the DNA method only. However, numbers of prey found in each scat were not high enough to yield accurate cluster analyses from the hard-parts method only.

## 4.4.6 Stable isotopes

The stable isotope signature of lactating females breeding on St. Paul Island (Reef rookery) was  $-19.13 \pm 0.48$  for  $\delta 13C$  and  $16.24 \pm 1.37$  for  $\delta 15N$  in plasma collected after their foraging trips. There were two distinct signatures, one centered around  $-18.77 \pm 0.30$   $\delta 13C$  and  $17.34 \pm 0.64 \delta 15N$ , and the other centered around  $-19.56 \pm 0.15 \delta 13C$  and  $14.86 \pm 0.50 \delta 15N$  (significantly different, both p < 5.10-5, Figure 4.5). These groups corresponded to females foraging on the shelf versus females foraging off the shelf (Table A.4 in Appendices section). The latter group is not significantly different from the isotopic signature found on fur seals breeding on and foraging off Bogoslof Island (*i.e.*, off the Bering Sea shelf ; ~  $-19.5 \pm 0.2$  for  $\delta 13C$  and ~  $14,23 \pm 0.3$  for  $\delta 15N$  from Fig. 3 in Zeppelin and Orr 2010), despite the  $\delta 15N$  being slightly higher in my data (see red squares on Figure 4.5).



Figure 4.5: Stable carbon and nitrogen isotope values in the plasma of 20 adult northern fur seal females breeding on St. Paul Island, Bering Sea (open symbols) from blood samples taken after their foraging trip at sea. Open circles are values for females foraging on the shelf (n = 12) and the open squares for females foraging off the shelf (n = 8). Closed symbols are averages  $\pm$  SD for on- and off-shelf groups. The red crossed circle represents stable isotopes values ( $\pm$  SD) for adult fur seals breeding on St. Paul Island and the crossed square isotopes values for adult fur seals breeding on Bogoslof Island from Fig. 3 in Zeppelin and Orr (2010).



Figure 4.6: Stable carbon isotope values in the blood of 20 adult northern fur seal females breeding on St. Paul Island from samples collected before versus after their foraging trip. Graph A and C shows the relationship between carbon and nitrogen isotope levels in plasma after and before their foraging trip. Graph B and D shows the relationship between carbon and nitrogen isotope levels in red blood cells collected before the foraging trip and carbon isotope levels in plasma collected after the foraging trip. Circles are values for females foraging on the shelf (n = 12) and the squares for females foraging off the shelf (n = 8). Dotted lines show 1:1 lines.

Isotopic values from blood collected before the seals' foraging trip had average  $\delta 13C$ in plasma of -19.05 ± 0.49 (-18.75 ± 0.36 for on-shelf group and -19.51 ± 0.21 for the offshelf group) and average  $\delta^{15}N$  of 16.03 ± 1.31 (17.00 ± 0.84 for on-shelf group and 14.92 ± 0.56 for the off-shelf group). These values were both significantly greater than isotopic values measured on RBC collected before the foraging trip, *i.e.*,-19.44 ± 0.36 for  $\delta^{13}C$  (-19.23 ± 0.29 for on-shelf females and -19.67 ± 0.25 for off-shelf females) and 15.02 ± 1.58 for  $\delta^{15}N$  (15.24 ± 0.73 for on-shelf females and 14.03 ± 1.45 for off-shelf females). However, none of the isotopic values in plasma collected before and after the seals' foraging trip differed significantly from each other (p > 0.34). Plasma  $\delta^{13}$ C and  $\delta^{15}$ N before and after the foraging trip showed a clear separation between on- and off-shelf groups (Figure 4.6, graphs A & C), with the exception of a couple of females (# 12, # 14 and # 9) whose signatures shifted between the two groups.

### 4.5 Discussion

### 4.5.1 DNA versus hard-parts techniques

DNA analyses added considerable value to determining the diet of northern fur seals by providing a more refined global picture of what they eat. In particular, my results show that northern fur seals consume a much more diverse diet than estimated from the traditional hard-part remains approach. My results also show that DNA meta-barcoding analysis was better than hard-part analysis at detecting the presence of prey in individual scats—detecting almost twice the number of taxonomic groups than did hard-parts (26 species detected in 100% of the scats for DNA analyses vs 14 species in 60% of scats for the hard-part method). As a result, the minimum sample size needed to detect of the maximum prey diversity is lower using the DNA method than the hard-part method.

DNA analyses were able to detect prey to a finer taxonomic level than hard-parts analysis, and performed particularly well for prey species that were morphologically identified with structures other than otoliths. For example, molecular analyses assigned occurrences of salmonids and cephalopods to the species level, while hard-parts could only identify them at the family level. Similar results have been shown for Steller sea lions (Tollit et al., 2009). Interestingly however, hard-part analysis did detect two species (rockfish and capelin) that the DNA method failed to pick up (although there was only 1 occurrence of each). These rare species were either not present in the soft matrix or may have been filtered out in the analytical process that removed DNA sequences that were proportionally less than 1%.

Despite biases in both techniques, results on the global diet composition of harbour seals and Steller sea lions have shown that they provide similar results at the population level (only species of minor importance differed between methods, (Austen Thomas thesis (being submitted), Tollit et al., 2009). In the case of northern fur seals, walleye pollock and gadids were the most ubiquitous prey group in their diet (Figure 4.3 and Table A.3) irrespective of the method used. This is consistent with previous findings from hard-parts in scats and spews (Sinclair et al., 1994; Gudmundson et al., 2003; Waite et al., 2012). However, there were some differences in their ability to detect some taxonomic groups. Most notably, hard-parts overestimated cephalopods and underestimated mesopelagic, flatfish, and salmons compared to DNA analyses.

Hard-parts detected the cephalopod group ~25 times more frequently than did the DNA sequences (SSFO ~12.6% for hard-parts and ~ 0.5% for DNA, Table 4.1). Squid beaks are hard chitinous-fibrous structures resistant to digestion and are known to accumulate by hooking themselves onto the lining of the digestive system before being excreted (Bigg and Fawcett, 1985; Harvey and Antonelis, 1994). This 'pulse-like' and unpredictable excretions of beaks might create an artificially inflated detection of cephalopods (Tollit et al., 1997; Staniland, 2002; Casper et al., 2007a). In contrast, the DNA meta-barcoding method may underestimate the occurrence of squids because the soft parts of squids and octopuses have a high water content and consequently a low total proportion of DNA in the body to be detected in the organic matter of scats compared to other prey species (A. Thomas PhD thesis, being submitted). The overestimation from hard-part and the underestimation from DNA might thus explain the 25-fold difference in detection between the two techniques.

DNA analysis found more salmon, forage fish and flatfish in the scats than did hardparts analysis. The relative absence of hard-part structures of salmons might mean that seals consumed young salmon whose hard-parts were highly digested, as was found in a similar study performed on harbour seals (A. Thomas PhD thesis, being submitted), or that seals might have not ingested the heads of larger salmon (*i.e.*, a 'belly-biting' strategy, Hauser et al., 2008). This is consistent with the fact that the only bones I recovered were vertebrae and gill rakers whose sizes suggest the ingested salmon were 20 to 45 cm long (although this is based on only 10 occurrences). Thus the discrepancy in FO between hard-parts and molecular methods either reflects differential digestibility of salmon hard-parts or different feeding strategies or both.

Flatfish and forage fish also seemed to be detected much more frequently by the DNA analyses compared to the hard-part analysis. Pacific halibut were detected 15 times out of 25 occurrences of flatfish, but hard-part remains were not present. The same pattern was seen for the 25 occurrences of Pacific herring out of 33 for forage fish. The low recovery rate of herring otolith (Tollit et al., 1997) could partly explain why hard-part analysis failed to detect herring, but this would not be the case for halibut. DNA sequences of flatfish and herring generally occurred in low quantities in the fur seal scats, suggesting they were relatively insignificant prey species. Species frequently consumed in low quantities will tend to be given greater importance when their occurrence is calculated using SSFO. This bias can be corrected using the proportion of sequence occurrences within individual scats averaged over all scats. However, interpreting the percent occurrence of sequences is not straightforward because it is species-specific and proportion-dependant (Deagle et al., 2013; Thomas et al., 2014). In addition, DNA meta-barcoding is known to have a proportiondependent bias in the read of sequence abundance whereby species present in low proportion tend to be overestimated while species in high proportion are comparatively underestimated (although this proportion dependence decreases the more complex the diet) (Kembel et al., 2012; Thomas et al., In prep.). As a consequence, I relied on the presence and absence of prey in individual scats to calculate SSFO, but provide the calculated percent of sequence occurrence in Appendices section for comparison (Table A.3).

Combining results from the DNA and hard-parts techniques can increase the accuracy of diet interpretation than either method alone, and has the potential to increase diet richness by suppressing specific biases (Casper et al., 2007b; Tollit et al., 2009). Only 21% of the prey I found were simultaneously detected with both techniques in the same scats (67 occurrences out of the 321 in total). Out of the 254 occurrences detected with a single method, ~ 88% of the prey were identified by the DNA-based method, and only 12% by the hard-part method. The numbers of scats containing multiple prey items also increased from ~ 28% using hard-parts alone—to ~71% with combined methods (~ 64% with DNA method, Figure 4.3). In my case, the DNA results drove the combined results, which was not necessarily the case in other studies (Casper et al., 2007b; Tollit et al., 2009). This means that biases of the DNA methods will likely have a greater effect on the interpretation of final results than will the biases inherent to the hard-parts method.

The digestion process likely explains why the DNA and hard-parts methods did not detect similar prey within single scat samples. Hard-parts pass though the GI tract slower than soft parts, and are more likely to be excreted in pulses compared to the continuous passage of soft tissues (Gudmundson et al., 2003; Casper et al., 2007a; Tollit et al., 2009). Morphological identification of prey taxa requires that hard-parts retain their morphological integrity, which was not the case as ~ 40% of my samples had no identifiable hard-parts. Similarly, the quality and integrity of DNA impacts detection rates, as would the numbers of sequences in the prey tissue.

Combining the outputs of the DNA and hard-part analyses may yield better estimates for some species such as cephalopods, reduce the overall importance of other dominant species in the diet such as gadids, and probably over-estimate the proportion of forage and flatfish in the diet as discussed above. In my case, the proportion of gadids in the diet decreased from ~69% using the hard-part method to ~ 57% using the combined methods. This decrease occurred because the DNA method detected increased numbers of prey species while the cumulated occurrence of pollock remained stable (high detection in both method), such that the overall relative proportion of pollock in the diet decreased. Squid, on the other hand, became 7 times more important compared to using just DNA alone—or conversely, was reduced in importance by a factor of 3.6 when the DNA offset the presence of squid beaks. Collectively, combining the DNA and hard-part diet descriptions shows northern fur seals are generalist predators that consume a diet more diverse than detected from hard-parts remains alone. It is still dominated by pollock and salmonids, but squids appear to be less important than the presence of beaks alone might suggest, while forage fish and flatfish may be overestimated in the combined results.

Despite the clear benefits of DNA-based methods in scats to determine diet composition of predators, they cannot provide the size or age class of prey ingested. This can only be obtained from the lengths of sagittal otoliths, squid beaks or other hard-parts remains in scats. Reconstructing the size of ingested prey based on relationships with otolith and beak lengths (Harvey et al., 2000; Tollit et al., 2004) is important given that the energy content of fish and squids differ depending on age and size (Paul and Paul, 1999; Logerwell and Christiansen, 2000; Iverson et al., 2002). Following the flow of energy from prey to

predators require this information from the hard-part method. Integrating morphological and molecular-based data thus improves the quantitative and qualitative information about diets, and contributes to a better understanding of the trophic relations and foraging efficiencies of northern fur seals.

## 4.5.2 Stable isotopes and prey associations reveal two foraging strategies

The DNA meta-barcoding and morphological identification are scats-based methods that provide useful information about diet composition. However, they are only representative of what animals consumed over a short 10 - 48 h period before they returned to land (Orr and Harvey, 2001; Staniland, 2002; Deagle et al., 2005; Casper et al., 2007a). Longer timescales are sometimes equally or more relevant in ecological studies for understanding foraging behaviours and efficiencies, as well as the relationships between prey fields and predators, and the link to fitness.

Plasma isotopic turnover is considered to be ~ 2 weeks (Kurle, 2002; Orr et al., 2008), so I assumed it reflects the foraging strategy during the foraging trip I monitored. The stable isotope results showed that female fur seals foraged in two different habitats ( $\delta^{13}$ C) and on prey at different trophic levels ( $\delta^{15}$ N, Figure 4.5) When combined with foraging locations (Figure 4.1), these groups corresponded to 1) females foraging off the shelf on pelagic prey at lower trophic levels, and 2) females foraging in a more nearshore/benthic system on higher trophic level prey. The females foraging on the shelf (~100 m depth) had access to the benthos and its associated prey, which was not possible for fur seals feeding in the basin (>1000 m). Females foraging over the shelf dove to deeper depths (sometimes to the bottom) for longer times overall compared to the strictly pelagic, shallow and short dives of the off-shelf females. Such differences in on-shelf and off-shelf foraging behaviours have been previously noted for females breeding on Reef rookery (Gentry, 1998; Nordstrom et al., 2013).

The 'on-shelf' isotopic signatures did not differ significantly from previous measurements (~ -18.9  $\pm$  0.5 for  $\delta^{13}$ C and ~ 16.4  $\pm$  1.4 for  $\delta^{15}$ N from Zeppelin and Orr (2010) added in red on Figure 4.5 and the 17.3  $\pm$  0.1  $\delta^{15}$ N measured for females from St.

Paul and St. George in July August (Kurle and Worthy, 2001). Unfortunately, the foraging locations associated with these additional measures of isotopic signatures were not recorded.

It is noteworthy that female fur seals foraging off the shelf had  $\delta^{13}$ C and  $\delta^{15}$ N isotopic signatures that were similar to females foraging off of Bogoslof Island (53.9N, 168.0W) north of the Aleutian Islands within the Bering Sea deep Basin (~ -19.6 ± 0.2 for  $\delta^{13}$ C and ~ 14.2 ± 0.3 for  $\delta^{15}$ N from Zeppelin and Orr (2010) shown with a red square Figure 4.5). This indicates that the off-shelf Pribilof fur seals have similar diet and foraging habitats than fur seals breeding on Bogoslof Island. The exception was female #6 which stayed foraging on the shelf but showed an isotopic signature corresponding to off-shelf diet (see Table A.4 in Appendices section). Her foraging trip lasted only 5.5 days however. This means that my isotopic values likely take into account diet from previous foraging trips. In addition, she was the only 'on-shelf' female that went foraging SW towards the shelf. Even though she did not seem to go foraging in the basin, she might have stayed around the shelf break and consumed basin prey advected on the shelf break.

Diet composition estimated from scats collected on Bogoslof island indicate that fur seals breeding there consume mid-water prey in oceanic domains consisting mostly of mesopelagic fish and gonatid squid (both 73.2% FO in Zeppelin & Orr 2010, >75 % FO Trites A.W., Pers. Comm.) and, to lesser extent, myctophids (17.1% FO, Zeppelin and Orr, 2010) and salmon (~ 15 % FO, Trites A.W., Pers. Comm.). Fewer than 10 % of the scats in both studies contained walleye pollock. In all likelihood, females from the Pribilofs that travelled off shelf likely consumed a similar diet to those breeding on Bogoslof Island. Such a possibility is consistent with the cluster analyses of prey associations (Figure 4.4) showing one cluster of mesopelagic and cephalopods species associated with pelagic mid-water oceanic ecosystems, and a second cluster of flatfish, hexagrammids, forage fish and salmon that are associated with shelf systems.

It is noteworthy that 100% of the scats that contained mesopelagic species also contained cephalopods (although cephalopods were not exclusively associated with mesopelagic), but the scats with the cluster mesopelagic/cephalopod represented only ~ 5% of the total number of scats. If we consider this cluster representative of the diet of the off-shelf females, then 40% of my tagged animals foraged off-shelf while only 5% of the scats

collected represented their diet. This means that this foraging strategy is under-represented in the scats collected on St. Paul Island, such that the dietary results would have to be multiplied by a factor of 8 to reach the same ratio and represent its overall importance. This is not surprising knowing that scats are indicative of food ingested within the last 10 - 48 h in fur seals (Bigg, 1981; Casper et al., 2007a), and that females foraging off the shelf travel for ~150 km (~ 24 h and up) to return to land. The probability they defecate their meals eaten off-shelf before reaching the shore is consequently greater than for females foraging in waters surrounding the rookery.

The slightly higher  $\delta 15N$  for off-shelf females from St. Paul compared to the Bogoslof females (Figure 4.5) likely reflects greater amounts of walleye pollock in the St. Paul diets. My cluster analysis shows an association of gadids with the other 2 clusters (Figure 4.4). This indicates that even if the pollock is less strongly associated with the mesopelagic + cephalopod cluster (longer distance), it is still ubiquitously present in the scats containing these prey. Juvenile pollock are widely distributed in basin and shelf regions of the Bering Sea (Benoit-Bird et al. 2013), and are likely consumed off the shelf in the deep oceanic domain to a certain extent. Dive profiles of the females I tracked indicate that they also foraged in these relatively shallow waters while transiting, and thus likely consumed gadids and other shelf-associated prey during their transit over the Bering Sea shelf to and from their rookery. This could explain the slight but not significant difference in  $\delta 15N$  signature between off shelf and Bogoslof females.

Nitrogen isotopic signature  $\delta^{15}$ N of females foraging off the shelf would increase compared to the Bogoslof females only if they consumed prey with isotopic signatures above the isotopic signatures of squid and mesopelagic species (~10 - 13 %<sub>0</sub>, Kurle and Worthy, 2001; Kline, 2010; Barger and Kitaysky, 2012). Pollock isotopic signatures increase with age and size as they consume higher trophic level species (Gorbatenko et al., 2008). Age > 2 pollocks have a  $\delta^{15}$ N isotopic signature of 15 - 16 %<sub>0</sub> (Gorbatenko et al., 2008; Barger and Kitaysky, 2012), while younger pollock have isotopic signatures ~11 - 12 %<sub>0</sub>. Consequently, given the ~ 4.5 %<sub>0</sub> enrichment between prey and predators for fur seals (Kurle, 2002), consuming juvenile pollock would have not increased overall  $\delta^{15}$ N of the off-shelf females from the Pribilof Islands to the point that it is statistically different from the exclusive basin-
foragers that feed around Bogoslof Island, but marginal consumption of older pollock (or other shelf associated prey with greater  $\delta^{15}N$ ) might have.

Carbon and nitrogen isotopic levels in plasma before and after the foraging trip (Figure 4.6 graphs A & C) show that overall female fur seals maintain their foraging strategy, *i.e.*, habitat and trophic levels over successive foraging trips. This is consistent with females tending to maintain their foraging habitat over successive trips (Call et al., 2008). However, isotopic levels in red blood cells collected before the foraging trip at sea (*i.e.*, between 20th of August and the 10th of September) were lower overall than plasma levels, consistent with the results of controlled studies comparing plasma and serum levels (Kurle, 2002; Orr et al., 2008). Given a 2-month or greater turnover rate of a red blood cell (Kurle, 2002; Zhao et al., 2006), the change in isotopic levels might also partially reflect the diet and feeding habitats during the spring migration through the Pacific Ocean to the Bering Sea (lower trophic level and/or in more offshore/pelagic/low latitude locations than during the foraging trip I studied.).

#### 4.5.3 Conclusions

DNA analyses of scats indicate a more diverse diet of fur seals than hard-part methods alone do, and thus allow more accurate analyses of prey association. In addition, C and N stable isotopes measurements complement the dietary information gleaned from scats. They show distinct differences in the trophic levels of prey consumed that are related to offshelf and on-shelf habitats. Combining this information with the clustering analysis of prey species co-occurring in scats suggest that females foraging off-shelf likely consumed mostly cephalopods and mesopelagic fish (with some pollock), while females foraging on the shelf likely consumed mostly walleye pollock and salmon. Teasing apart individual diets based on multiple techniques has important consequences for studies seeking to describe diets, energetic costs of foraging and ultimately foraging efficiency. It also provides a means to adjust summaries of diets derived from scats collected on rookeries to properly reflect the average diet of fur seals breeding on the Pribilof Islands.

# Chapter 5: Foraging efficiency is related to the reproductive fitness of lactating Antarctic fur seal through increased prey catch success and reduced time at sea

## 5.1 Summary

The efficiency with which individuals extract energy from their environment defines their survival and reproductive success, and thus their selective contribution to the population. Individuals that forage more efficiently (i.e., energy gained exceeds energy expended) are likely to be more successful at raising viable offspring than less efficient individuals. I tested this by equipping 20 lactating Antarctic fur seals (Arctocephalus gazella) breeding on the Kerguelen Island in the Southern Ocean with tags that recorded GPS locations, depth and tri-axial acceleration to determine at-sea behaviours and detailed timeactivity budgets during their foraging trips. I also simultaneously measured energy spent at sea using the doubly labeled water method, and estimated the energy acquired while foraging from 1) type and energy content of prey species present in scat remains, and 2) numbers of prey capture attempts determined from head acceleration. Finally, I followed the growth of 36 pups from birth until weaning (of which 20 were the offspring of my 20 tracked mothers), and used the relative differences in body mass of pups at weaning as an index of first year survival and thus the reproductive success of their mothers. My results show that females with greater foraging efficiencies produced relatively bigger pups at weaning. These mothers achieved greater foraging efficiency by extracting more energy per minute of diving rather than by reducing energy expenditure. This strategy also resulted in the females spending less time diving and less time overall at sea, which allowed them to deliver higher quality milk to their pups, or allowed their pups to suckle more frequently, or both. The linkage I demonstrate between reproductive success and the quality of individuals as foragers supports the optimal foraging theory, and provides a quantitative framework to investigate how changes in the availability and accessibility of prey affect the time-activity budgets and foraging efficiencies of mothers—and ultimately the growth and survival of their pups.

#### 5.2 Introduction

Optimal foraging theory postulates that natural selection favours animals that forage more efficiently, where foraging efficiency is defined as the ratio of energy gained to energy expended to acquire food (MacArthur and Pianka, 1966; Pyke, 1984; Stephens and Krebs, 1986). The energy gained in excess of energy expended can be allocated to reproduction, survival and growth (Boggs, 1992). Individuals that maximise their energy return per unit of energy (and time) spent have more energy (and time) to allocate to reproduction over their lifetime and thus a greater fitness than less efficient conspecifics. Foraging efficiency thus ultimately shapes the dynamics of populations.

Empirically testing optimal foraging theory requires knowing the amount of energy spent foraging, the nutritional quality and quantity of resource ingested, and a concomitant measure of reproductive success of individuals. Studies that have controlled energy gain and energy expended, as well as controlled studies of species with 'rapid' reproductive rates have yielded findings consistent with the optimal foraging theory (Ritchie, 1990; Lemon, 1991). However, validating the theory in the wild is more complicated because of the difficulty of simultaneously measuring individual energy intake and output, as well as an individual's own reproductive success. This is particularly true for marine mammals that are long-lived and elusive, and live in an environment where direct observation of foraging is impossible.

Some studies have investigated life history traits including reproductive rates in marine predators (Chastel et al., 1993; Boyd et al., 1995a), but have generally not linked them to foraging efficiency. Others have looked at foraging efficiency indices, but often assume that these indices are linked to fitness without explicitly linking the two parameters (Luque et al., 2007). There is therefore a need to link reproductive success with measures of foraging efficiency, which would allow predictions to be made about how the individual fitness and population trends of top predators are affected by changes in prey availability and foraging behaviours.

The energetic cost of foraging in free-ranging pinnipeds can be assessed using indirect calorimetry techniques such heart rates or doubly labeled water (Lifson and McClintock, 1966; Butler et al., 1992; Butler, 1993; Speakman, 1997; Froget et al., 2004). In

contrast, the energy gained while foraging has been traditionally measured by identifying prey species in regurgitates, scats, or stomach contents (Arim and Naya, 2003; Gudmundson et al., 2003; Karnovsky et al., 2012) and estimating numbers consumed from changes in body water pool (Costa, 1993), or with stomach temperature sensors that measure the changes in temperature between the predator's body and the cold prey ingested (Grémillet and Plös, 1994; Kuhn and Costa, 2006), both of which present challenges in wild otariids. Consequently, studies have tended to either report foraging effort but not gain (Costa et al., 1989; Shaffer et al., 2003), foraging gain but not effort (Lea et al., 2006; Staniland et al., 2007), or have used behavioural indices rather than quantitative measurements of foraging efficiency (Bailleul et al., 2005; Weimerskirch et al., 2005; Lea et al., 2006). More recently, however, tri-axial accelerometers have been validated and used to measure numbers of prey capture attempts in free-ranging marine predators (Skinner et al., 2009; Suzuki et al., 2009; Viviant et al., 2010; Watanabe and Takahashi, 2013; Ydesen et al., 2014). This technological innovation makes it possible to quantitatively estimate foraging efficiency of individual marine predators by combining cost of foraging through doubly-labeled water, with gain of foraging from diet composition analyses and measure of prey capture attempts using accelerometers.

Antarctic fur seals give birth to a single pup once a year. Mothers then nurse their pups for 4 months during which time they alternate periods of foraging at sea to replenish reserves and fasting periods on land while nursing their pups. Allocation of energy to their pup will dictate their growth rate and mass at weaning. This is directly linked to survival of the pup during their first year at sea, the critical period in the life cycle of fur seals (Doidge and Croxall, 1989; Baker and Fowler, 1992; Boltnev et al., 1998; McMahon et al., 2000; Hall et al., 2001; Beauplet et al., 2005). Consequently, growth rates and mass at weaning of pups can be used as indices of annual reproductive success of female fur seals. As central place foragers, mothers are also time-limited during their foraging trip by the fasting capacity of their pups and must trade-off the time they take to replenish their reserves with the nutritional needs of their pups. Thus, given the time constraints mothers face while feeding, the allocation of their time to different activities at sea will affect both energy expenditure and return. It is thus important to study foraging efficiency linked to reproduction success within the context of individual time-activity budgets.

Antarctic fur seals employ a range of foraging strategies depending on environmental conditions and the distribution of prey patches (Lea et al., 2002b). The disproportional effects of climate change have made environmental conditions and the availability of prey less predictable in high-latitude marine ecosystems (Serreze et al., 2000; Walther et al., 2002). Top predators such as marine mammals living in high-latitude areas are expected to be affected more by these changes, but it is difficult to predict their response with any confidence. One means of predicting how marine mammals will respond to climate change is to quantify foraging efficiency within the context of optimal foraging theory and investigate how foraging efficiency varies depending on behavioural choices and strategies of individuals at sea, and how it impacts fitness via reproductive success. I sought to test optimal foraging theory on a wild population of pinnipeds (Antarctic fur seals) by linking foraging strategies to efficiencies and proxies of fitness. I determined 1) whether foraging efficiency affected reproductive success as indicated by the body size of pups at weaning.

#### 5.3 Material and methods

#### 5.3.1 Data collection

Data were collected on 20 lactating Antarctic fur seal females at Pointe Suzanne, Kerguelen Island (Southern Ocean,  $49^{\circ}26$ 'S -  $70^{\circ}26$ 'E) during the breeding season (Jan-Feb 2012) under the ethical regulations approval of the French Polar Institute (IPEV) and the UBC Animal Care Committee. Doubly-labeled water data were collected on healthy-looking females with a confirmed suckling pup. They were captured using a hoop net, carried over a short distance to a restraint board where they were anesthetized with isoflurane gas. Standard morphometric measurements were made to the nearest 0.5 cm, and mass was recorded using a scale at  $\pm 0.2$  kg. Measurements of daily energy expenditure (kJ/day) were performed using the Doubly-Labeled Water (DLW) method (Lifson and McClintock, 1966; Butler et al., 2004). This method has been previously validated by comparison to indirect calorimetry in a range of small mammals (Speakman and Krol, 2005), as well as on seals (Sparling et al., 2008). A blood sample was collected from the back flipper in Monovette syringes (Sarstedt) coated with Li-Heparin and plasma was isolated from red blood cells by centrifugation at 1000g for 10 min before injection of the doubly-labeled water intravenously. More details on the doubly-labeled water material and method used in this study can be found in Chapter 2.

Females were also equipped with data loggers glued to the dorsal mid-line fur using a 2-part Devcon 5min epoxy glue. Daily Diary tags (DD, Wildlife Computers) recording triaxial acceleration and tri-axial magnetic field at 16 Hz, and depth, light level, and water temperature at 1 Hz were glued as close as possible to the projected center of mass on the back of the animal (roughly between the scapula). Fastloc GPS MK10 loggers (Wildlife Computers) were glued lower down the back from the DD tags and recorded GPS coordinates along the track of the animal at sea, as well as depth and water temperature at 1 Hz. Finally, GCDC X6 or X8 accelerometers were glued on the head of the animals that recorded tri-axial acceleration at 16 or 20 Hz. Once the devices were securely attached and a second blood sample was taken at the end of the equilibration period (~ 2 h), the females were released on full recovery from the anaesthesia and allowed to rejoin the colony. Individuals were recaptured after a single foraging trip and anaesthetized as previously described. A final blood sample was taken to determine isotope levels of <sup>3</sup>H and <sup>18</sup>O at the end of the foraging trip, and all the data loggers were removed by cutting the fur beneath the devices. Blood samples were frozen at -20 °C until analyses in the lab. A second set of morphometric measurements were also taken at this time.

# 5.3.2 Energy expenditure

All methods for analyses and calculations of the energy expenditure of female fur seals using the doubly-labeled water are detailed in Chapter 2. In short, the isotope ratios <sup>18</sup>O: <sup>16</sup>O and <sup>2</sup>H: <sup>1</sup>H were analysed using gas source isotope ratio mass spectrometry (Optima, Micromass IRMS and Isochrom  $\mu$ G, Manchester, UK). Isotope enrichments were converted to CO<sub>2</sub> production for each individual using a two-pool model and initial isotope dilution spaces were calculated using the plateau method (Halliday and Miller, 1977). I used the equation from Speakman et al. (1993) to calculate metabolic rates from DLW concentrations and a respiratory quotient of 0.80. Three of the 20 females had isotopic levels too close to background levels for accurate measurements and were discarded.

Individuals spent time on land after the post-equilibration sample and upon return to the colony before recapture and the final blood sample was collected. Energy spent during this 'non-foraging' time was part of the DLW measurement, and given I were interested in estimating only energy expenditure at sea, energy expenditure at sea was calculated by subtracting on-land expenditure from the total estimate using previously determined values for lactating Antarctic fur seals (4.56 W/kg in Costa and Trillmich, 1988) while on land. Energy spent while diving or transiting were calculated using activity-specific metabolic rates determined in Chapter 2.

#### 5.3.3 Foraging behaviours

I used depth data recorded by the DD tags to determine diving behaviours, or depth data recorded by the fastloc MK10 during the rare times that the DD tags malfunctioned. Any drift in the pressure sensors or error spikes were corrected prior to analyses. Diving behaviours were reconstructed using a custom-made R program previously developed for Antarctic fur seals. Dives were defined as periods of time that animals spent under water below a minimum depth of 3 m and for a minimum of 4 s until they went back to the surface and I derived dive duration and maximum dive depth for each of them.

Distances traveled at the surface of the ocean (or horizontal distances) were calculated by measuring the linear distance between 2 successive GPS locations taking into account the curvature of the Earth using the Haversine formula (Sinnott, 1984). GPS locations have a high spatial and temporal resolution (they were set to record a location every 5 min), so GPS tracks did not required interpolation (Tremblay et al., 2006). Part of the distance traveled under water while diving was inherently taken into account in the measured horizontal distance traveled. Consequently, I only estimated vertical distance traveled while diving by doubling the maximum dive depth of each dive, which should provide an acceptable estimate because fur seals are known to dive at relatively shallow depths (~35m average).

Time spent foraging or diving was calculated from the time when the animal was below the water surface and performing confirmed dives plus the post-dive interval as calculated from the Bout-Ending Criterion using the R package DiveMove (Luque and Guinet, 2007). The animals were considered to be transiting (*i.e.*, traveling fast between 2 locations) whenever they were neither diving, nor sleeping and when the calculated speed at the surface (*i.e.*, time needed to travel a distance between 2 GPS points) was >1 ms<sup>-1</sup>. All methods to determine time-activity budgets in fur seals (and time-activity budget results) are detailed in Chapter 2.

Average dive parameters, such as for dive depths and dive durations, were nested within animals and were calculated using linear mixed effect models with no fixed effects (only the intercept was calculated) and with individual as a random effect to take into account that each animal performed a different number of dives.

#### 5.3.4 Prey capture attempts

Prey capture attempts (PrCA) were measured using acceleration data recorded on the head of the animals at 16 Hz (Skinner et al., 2009; Suzuki et al., 2009; Viviant et al., 2010; Watanabe and Takahashi, 2013; Ydesen et al., 2014). Only acceleration while the animal was diving below 3 m was kept for analyses. The head acceleration was recorded from a GCDC accelerometer, and dive depth was recorded by the DD tags attached to the back of the animals. Upon careful investigation of the signals, I noticed clock drifts between the 2 loggers for most of the animals. The first step in my analyses was thus to synchronise the 2 loggers using a custom-made code in Matlab so that the head acceleration matched the dive depth and back acceleration. The first and last dives of each foraging trip were singled out for both the head and the back loggers. Whenever a clock drift was noticed between the 2 acceleration channels, I estimated the time difference, and added data points at random throughout the data on the channel that was shorter following a uniform distribution to compensate for this time difference. Extra data points were added as the average acceleration from 2 successive data points to not change the shape of the signal. The resulting synchronized channels were thus within 2 s of each other at the end of a foraging trip.

Once synchronization was satisfied, I only used the heave and surge channels to detect PrCAs (movement up and forward) (Viviant et al., 2010). I filtered the raw acceleration for these 2 channels using a  $3^{rd}$  order high-pass filter at 3 Hz (Viviant et al., 2010; Iwata et al., 2012) to filter out the static acceleration due to the position of the animals.

This only left out the signal from rapid head movements. The surge (x) and heave (z) dynamic accelerations were then summed and a running variance was applied over a 2-sec window. A cluster analyses on the resulting variance of dynamic acceleration was then performed using the k-mean function in R, which provided each animal with an individual threshold above which the signal was considered to correspond to a PrCA. Events detected within less than 1 sec of each other were considered coming from the same PrCA event (as assessed from feeding trials with live fish on harbour seals—Austen Thomas, pers. comm.) and from video recordings of Steller sea lions (Viviant et al., 2010). PrCAs represented only an index of attempts to capture prey, since it was impossible to determine whether animals were successful at capturing the targeted prey.

The PrCA detection method with head accelerometers has been validated on otariids in captive settings (Viviant et al., 2010) and has been used on phocid seals in the wild (Liebsch et al., 2007; Gallon et al., 2013; Naito et al., 2013; Guinet et al., 2014; Volpov et al., 2015). It has also been used on diving birds (Kokubun et al., 2011; Watanabe and Takahashi, 2013). However, I tested the accuracy of using the acceleration signal from the back of the animals by performing the same analyses on the data collected from the DD tag glued between the shoulder blades and by comparing it to the results obtained from the head signals. I found that back acceleration could estimate PrCA as well as acceleration on the head of the seals with this method (Figure 5.1, linear regression:  $PrCA_{Back} = 34.25 + 1.00 \times$  $PrCA_{Head}$ ;  $p < 10^{-15}$ ,  $R^2 = 0.90$ ). Similar results have been found for head and back accelerometers deployed on southern elephant seals (C. Guinet, pers. comm.). Consequently, I calculated PrCA using back acceleration corrected for the ~ 34 PrCA overestimation per night whenever the head accelerometer failed to record data over the full foraging trip.

Differences between prey capture attempts per days or per dives were estimated using linear mixed effect models with no fixed effect and with individual as a random effect to account for each animal performing a different number of dives or days at sea.



Figure 5.1. Prey capture attempts (PrCA) detected from back acceleration compared to PrCAs detected from head acceleration signals. Each dot represents one animal. The dotted line shows the results of regression model  $PrCA_{Back} = 34.25 + 1.00 \times PrCA_{Head}$  ( $R^2 = 0.90$ , *p* slope < 10<sup>-15</sup>).

## 5.3.5 Diet estimates

I determined diet composition of Antarctic fur seals from 20 scats collected at Pointe Suzanne rookery on Kerguelen Island in March 2012. Samples were kept frozen at -20°C until ready to be processed in the lab as detailed in Chapter 4. Hard part remains were identified to the smallest taxon possible at the Centre d'Etudes Biologiques de Chize (France) following recommendations from Lea et al. (2002c). Frequency of occurrence (FO) and relative proportion of each prey item in the diet (SSFO) were calculated the same way as for NFS (Chapter 4) and were compared to previously determined diet composition on Antarctic fur seals (Lea et al. 2002b). Upon identifying the main prey items, the size and energy content of the fish and squids found in the scats were taken from published sources (Cherel et al., 2002; Lea et al., 2002a; Lea et al., 2002c; Tierney et al., 2002; Fernández et al., 2009). The size and energy content of squids were averaged per year and then over the 3 years from Lea et al. (2002b) to obtain a squid estimate for my study as most of the squid beaks I found were unidentifiable.

I obtained energy density of the diet  $(ED_{Diet})$  by averaging the energy density of different prey  $(ED_i)$  weighted by their proportion within the diet  $(P_i)$  over the number of prey in the diet *N* using:

Eq. 5.1 
$$ED_{Diet} = \frac{\sum (ED_i \times P_i)}{N}$$

Whenever information was missing for prey of low frequency of occurrence in the diet, I replaced it with the energy density of the closest related prey item or by the average of the energy content for the specific prey group. Once the mass (BM in g) and the energy density (ED in kJ/g) of each prey item (i) were estimated, I calculated the average energy content of a specific fish (EC in kJ) using:

Eq. 5.2 
$$EC_i(kJ) = BM_i(g) \times ED_i(kJ/g)$$

The average energy content (in kJ) of a random non-specific prey (p) consumed by fur seals given a specific diet was calculated by weighting the energy content of a specific prey item by its relative proportion in the diet (P):

Eq. 5.3 
$$EC_p(kJ) = \sum_i (EC_i(kJ) \times P_i)$$

Means  $\pm$  SD of energy content of each prey (*EC<sub>i</sub>* in kJ) were calculated by generating 1000 values of mass and 1000 values of energy density (*ED<sub>i</sub>*) using normal distributions of their respective means  $\pm$  SD from Table A.5. I calculated the error around *P<sub>i</sub>* by bootstrapping scats (*i.e.*,random sampling with replacement of individuals scats), and recalculating FO and SSFO for each new generated dataset (n = 1000). I then obtained the 95% CI and the SD from these values. Means  $\pm$  SD of energy densities (*ED<sub>Diet</sub>*), and energy content of an average prey (*EC<sub>p</sub>*) in the 2 diets were calculated by generating values of *EC<sub>i</sub>* and *ED<sub>i</sub>* for each prey type (*i*) in proportion to their respective mean  $\pm$  SD. For prey species with no *ED<sub>i</sub>* or *EC<sub>i</sub>* values, I used the average *ED* or *EC* from the prey group as their values weighted by their own proportion within the diet. As the prey group 'Other' did not have values for mass or energy density, I considered it as an average of the rest of the diet weighted by its relative importance in the diet.

## 5.3.6 Foraging efficiency

The foraging efficiency (*FE*) of each seal (*i*) was calculated as the ratio between the energy expenditure at sea obtained from the DLW measures (*EE<sub>i</sub>*) per animal *i* and energy gained while foraging at sea. Energy gained was estimated as the specific energy content of a non-specific prey (*EC<sub>p</sub>*) in their diet multiplied by the number of time seals *i* attempted to capture prey (*PrCA<sub>i</sub>*).

Eq. 5.4 
$$FE_i = \frac{EC_p \times PCA_i}{EE_i}$$

Seals with DLW results that were too close to background and seals that did not have acceleration data for the complete foraging trip were omitted from calculations. I am aware that PrCAs represent attempts and not confirmed prey captures, but I assumed that unsuccessful PrCA were minor compared to successful ones (93% of attempts were successful in Australian fur seals, Volpov et al., 2015), and that proportion of unsuccessful attempts were consistent between seals.

As the 3 parameters used to calculate the foraging efficiency of each individual animal ( $FE_i$ ) contain inherent errors, I calculated the resulting uncertainty around  $FE_i$  using the following 3 steps: **1**) I calculated error in  $EE_i$  by generating 1000 values following a normal distribution of 1.8 ±7.2% of the measured values of DLW. This error was estimated by Dalton et al. (2014b) when DLW was compared to respirometry on northern fur seals (using same equation and in the fall months). **2**) I calculated error in  $EC_p$  by generating 1000 values using a normal distribution following the means ± SD for mass and energy density for each prey in the diet mentioned above (see Diet estimates section). **3**) I calculated error in  $PrCA_i$  by adding a detection error and subtracting a false positive error generated using uniform distribution between the ranges mentioned above to the measured  $PrCA_i$  value (1000 values generated). Detection rate of PrCA (true positive rate) is known to range from 68 to 97% (underestimation of true PrCA) and the false positive rate from 6 to 48 % (overestimation of true PrCA) in Steller sea lions and Australian fur seals (Viviant et al., 2010; Volpov et al., 2015).

Mean  $\pm$  SD of  $FE_i$  was calculated over the 1000 generated  $PrCA_i$ ,  $EE_i$  and  $EC_p$  using Eq. 2.1. I calculated uncertainty over the average FE per group of foraging strategy (on-shelf

versus off shelf) using the bootstrap method over 1000 simulations, where the random sampling with replacement were taken within the 1000 values of  $FE_i$  generated per animal within the group.

A sensitivity analysis was performed to estimate the contribution of each input variables uncertainty to the overall variance in the resulting foraging efficiency. This was done by computing the standardized regression coefficients (SRC), its bias and its 95% confidence intervals for each of the input variable using the *src* function in R (*'sensitivity'* library, R3.0.3) over 1000 simulated values.

#### 5.3.7 Pup growth

Thirty-six pups were randomly chosen in the Pointe Suzanne colony and were followed from birth until they could no longer be found on the colony. Mothers of 20 of the 36 pups initially followed were selected to be tracked. Standard morphometric measurements were recorded at birth and every 7-10 days or longer as the pups started to wander further from the colony. Length and girth were measured to the nearest 0.5 cm, and mass was recorded using scale at  $\pm 0.1$  kg.

Growth from birth to weaning of each individual pup was modeled with the von Bertalanffy equation (von Bertalanffy, 1938) using the nls function (nlme package in R):

Eq. 5 
$$BM = A \times (1 - e^{(-K \times age - t_0)})$$

where BM is body mass in kg, A is the asymptotic mass of pup at weaning, K is the curvature parameter (day<sup>-1</sup>), age is the pup age (day) and  $T_0$  is the age (day) at which the pups have a mass equal 0 kg. I also modeled the average male growth and the average female growth separately because male pups have a higher growth rate than female pups in fur seals (Kerley, 1985; Trites, 1993; Davis et al., 1996; Chambellant et al., 2003; Osman et al., 2010).

# 5.3.8 Linking foraging efficiency of mothers and growth of pups

I calculated the difference between individual foraging efficiencies of each female and the average foraging efficiency of all the females as a metrics of relative quality of the mothers as foragers. I simultaneously calculated the individual estimated mass at weaning (127d) from individual pup growth curves and calculated the difference with the average mass at weaning calculated from the average sex-specific growth curve as described above as a metrics of relative size at weaning of pups. I tested the relationship between these two metrics using a type II linear regression that took into account the fact that there were errors associated with both the response and the explanatory variables (*lmodel2* package in R) using the ranged major axis (RMA) method. I also tested the relative size of pups at weaning against other foraging metrics of mothers, such as time spent at sea or diving or rate of energy gain while diving using the same methods.

## 5.4 Results

# 5.4.1 Foraging behaviours

Antarctic fur seal females all foraged east to south-east of Kerguelen Island on the Kerguelen plateau (plateau with depths > 500 m). They weighed  $31.1 \pm 0.9$  kg prior to departure (range: 24.0 - 34.0) and gained  $0.6 \pm 0.6$  kg during their trip ( $2.2 \pm 1.8$  % of their body mass). They traveled  $635 \pm 77$  km (271 - 1295) with foraging trips lasting for  $7.6 \pm 3.8$  d (2.5 - 15.5 d). During these trips, they performed an average of  $3949 \pm 597$  dives to a mean depth of  $19.9 \pm 2.7$  m (75.5 % of which were less than 15 m deep, and exclusively nocturnal) that lasted  $42.6 \pm 4.5$  s on average. They spent  $29.0 \pm 0.7$  % ( $51.3 \pm 5.9$  h) of their time diving,  $26.4 \pm 1.6$ % ( $49.8 \pm 7.9$  h) transiting,  $36.3 \pm 2.0$ % ( $60.9 \pm 7.6$  h) performing surface activities, and  $8.2 \pm 1.7$ % ( $12.9 \pm 3.0$  h) resting.

## 5.4.2 Energetic cost of foraging

Rates of energy expenditure per day at sea averaged  $17.7 \pm 1.1 \text{ MJ/d} (0.59 \pm 0.04 \text{ MJ/d/kg} \text{ for all females, n = 17})$ . This translated into animals spending an average of  $66.0 \pm 7.5 \text{ MJ} (2.2 \pm 0.3 \text{ MJ/kg})$  while diving and  $38.4 \pm 6.4 \text{ MJ} (1.3 \pm 0.2 \text{ MJ/kg})$  while transiting from activity-specific metabolic rate equations in Chapter 2.



Figure 5.2: Example of a dive profile of a female Antarctic fur seal during a foraging trip (left panel) and over 5 min (right panel). Orange dots show where prey capture attempts occurred during the dives.



Figure 5.3: Foraging locations of the 20 Antarctic fur seal females tracked on the Kerguelen plateau. Red dots show where the animals attempted to capture prey within the range of their foraging location (with yellow showing no prey capture attempts). Trips of females with no acceleration are shown in yellow only.

## 5.4.3 Prey capture attempts

Females attempted to capture prey  $0.87 \pm 0.11$  times per dive while foraging (Figure 5.3). When only selecting dives in which at least one PrCA occurred, their capture rate increased to  $2.04 \pm 0.11$  prey per capturing dive. This translated in females capturing on average  $336 \pm 38$  prey per night of foraging and  $2328 \pm 387$  prey over their entire foraging trip (range 704 - 6613 prey). When corrected for detection rate of PrCA (true positive rate) ranging from 68 to 97 % and a false positive rate of PrCA from 6 to 48 % (Viviant et al., 2010; Volpov et al., 2015), the total PrCA over the foraging trip was slightly lower at  $2139 \pm 424$  (704 – 6612).

Table 5.1: Relative proportion (%), average prey mass (in g), prey energy density (ED in kJ/g), energy content (in kJ) of prey groups in diets of female Antarctic fur seals breeding to Pointe Suzanne on Kerguelen Island. Groups 'Other' were assigned average diet values weighted by its percentage in the total diet for a calculation of  $ED_{Dieb}$  and  $EC_p$ . Bold values are for the total Prey group.

	Prey group	Perc. in diet (%)	Mass (g)	ED (kJ/g)	EC (kJ)
Cephalopod	Cephalopod	$12.11 \pm 3.38$	$82.67 \pm 32.05$	$4.05\pm0.10$	347.11 ± 4.14
Myctophid		$75.50 \pm 7.01$	$12.19 \pm 0.11$	$8.56 \pm 0.25$	$112.9 \pm 0.94$
	E. antarctica	$3.50 \pm 1.50$	$3.20 \pm 1.80$	$13.30 \pm 2.60$	$40.97\pm0.82$
	E. subaspera	$10.78 \pm 1.66$	$11.80 \pm 4.30$	$7.40 \pm 1.00$	$88.45 \pm 1.09$
	G. fraseri	$2.08 \pm 1.11$	$5.17 \pm 0.22$	$10.20 \pm 3.50$	$52.63 \pm 0.57$
	G. nicholsi	$9.11 \pm 1.77$	$17.33 \pm 1.95$	$9.80 \pm 1.00$	$168.92 \pm 0.84$
	G. piabilis	$14.11 \pm 2.06$	$24.93 \pm 0.87$	$9.50 \pm 1.70$	$235.25 \pm 1.36$
	<i>G. sp.</i>	$9.53 \pm 1.88$	$15.81 \pm 0.62$	$9.83 \pm 0.90$	$155.48\pm0.50$
	K. anderssoni	$0.83 \pm 0.55$	$0.47 \pm 0.12$	$8.10\pm0.30$	$3.82 \pm 0.03$
	P. bolini	$3.75 \pm 0.85$	$0.87 \pm 0.03$	$5.93 \pm 0.38$	$5.15\pm0.01$
	P. choriodon	$1.83 \pm 2.27$	$0.87 \pm 0.03$	$6.08 \pm 0.55$	$5.28 \pm 0.01$
	P. tenisoni	$12.11 \pm 1.85$	$0.77 \pm 0.20$	$6.23 \pm 0.12$	$4.75 \pm 0.04$
	Myctophidae sp.	$7.86 \pm 1.52$	NA	NA	NA
Nototheniid	Nototheniid	$4.44 \pm 2.59$	$58.40 \pm 0.00$	$5.03 \pm 0.17$	$293.99 \pm 0.31$
Other		$7.94 \pm 2.54$	NA	NA	NA
	S. hamiltoni	$1.25 \pm 0.95$	NA	NA	NA
	crustacean	$3.33 \pm 2.5$	NA	NA	NA
	mollusc	$2.36 \pm 1.16$	NA	NA	NA
	penguin	1.0 0.69	NA	NA	NA

### 5.4.4 Diet and prey energy contents

Diets of female Antarctic fur seals (Table 5.1) contained mostly myctophids (~75 %) and cephalopods (~12 %). Myctophids are small mesopelagic fish of mass 1 - 25g with a high energy density (range ~ 6 - 13 kJ/g). The largest prey item of the seals were cephalopods (~83 g) which are also the least energetically dense of their prey (~ 4 kJ/g). Overall, the energy content of their prey ranged from 5 to 350 kJ (Table 5.1). Given the contribution of each prey item to the total diet composition, the female fur seals ingested an average of 7.75 ± 2.47 kJ per gram of prey ( $ED_{Diet}$ ) with an energy content of 152.46 ± 1.08 kJ per average prey ( $EC_P$ ).

# 5.4.5 Foraging efficiency and pup growth

Only 14 females out of the 20 tracked had simultaneous data for energy expenditure and for prey capture attempts available to calculate foraging efficiency (Table 5.2). The calculated rates of energy gains for these 14 animals were  $177.7 \pm 21.4$  kJ/dive, which was  $130.6 \pm 16.3$  kJ/min spent diving and  $37.6 \pm 4.6$  kJ per min spent at sea. The average foraging efficiency for Antarctic fur seals was  $3.44 \pm 0.45$  (range 1.24 - 6.86, 95% CI: 2.54 - 4.38).

Sensitivity analyses showed that the largest contributor of the uncertainty around FE is related to diet estimates, and within the diet component, the estimation of mass of the fish ingested. Uncertainty around PrCA estimates comes second, and estimates of energy expenditure contributes the least to overall FE uncertainty (Table 5.3).

Deviation from average foraging efficiency of individual females ranged from -1.85 to 3.56. Neither the mass, nor the change in body mass of the females before and after foraging trips or the body condition (estimated as mass/length) were linked to the foraging efficiencies of the females (all p > 0.62). Foraging trip duration or time spent performing different types of activities at sea were also not related to foraging efficiencies (all p > 0.21). Foraging efficiency was positively related to rate of energy gained per min spent diving (p = 0.034) but not to rate of energy expenditure (p = 0.231).

Out of the 36 pups monitored, the 13 female pups weighed 4.6  $\pm$  0.5 kg and the 23 male pups were 4.9  $\pm$  0.5 kg at birth (Table 5.2, p = 0.02). Three pups disappeared rapidly

and I could not fit any growth curve on their data points, and 3 of them had no data close to weaning which means the model could not reach an asymptote. Growth model for male and female pups (Figure 5.4) were:

$$BM_{Male} = 12.04 \times (1 - e^{(-0.038 \times age + 14.07)})$$

$$BM_{Female} = 10.26 \times (1 - e^{(-0.020 \times age + 27.65)})$$

Calculated average masses at weaning (127-day old) from these equations were 11.73 kg for male pups and 9.74 kg for females. Deviation from average sex-specific mass at weaning of individual pup mass ranged from +3.46 kg to -4.56 kg (see Table 5.2).

Table 5.2: Measured and corrected energy expenditure (EE in MJ), measured and corrected number or prey capture attempts (PrCA) during a foraging trip at sea for female Antarctic fur seals (n=14). The corrected values of EE and PrCA were used with the estimated energy content per average prey (EC prey in kJ) to calculate the energy animals gained while foraging at sea (in MJ) and their foraging efficiency (*i.e.*, the ratio of energy gain/EE). The calculated mass at weaning (from individual Von Bertalanffy growth models) of pups from the tracked mothers are indicated with their sex in brackets.

Mother ID	Meas.EE (MJ)	Corr.EE (MJ)	Meas. PrCA	Corr. PrCA	Energy gain (MJ)	Foraging efficiency	Pup mass at 127d (sex)
21	229.31	225.20	7229	6613	1008.23 ± 168.09	4.50 ± 0.83	12.53 (F)
22	120.91	118.72	1976	1819	277.36 ± 45.11	2.35 ± 0.42	7.28 (F)
23	46.50	45.70	1656	1519	231.63 ± 39.70	5.10 ± 0.96	14.27 (M)
26	246.67	242.49	4417	4082	622.28 ± 104.31	2.58 ± 0.47	10.69 (M)
27	60.64	59.59	2093	1921	292.86 ± 49.79	4.94 ± 0.92	11.29 (M)
28	50.06	49.17	768	704	107.32 ± 17.87	2.19 ± 0.40	13.66 (M)
29	35.88	35.25	1719	1576	240.33 ± 40.19	6.86 ± 1.27	12.29 (F)
31	46.09	45.26	1066	981	149.53 ± 24.92	3.32 ± 0.61	8.34 (F)
32	59.00	57.90	1498	1364	207.97 ± 35.31	3.61 ± 0.67	11.08 (F)
33	112.36	110.24	1251	1148	174.95 ± 28.94	1.60 ± 0.29	8.76 (F)
34	89.51	87.86	1410	1295	197.46 ± 33.31	2.26 ± 0.42	11.76 (M)
36	130.43	128.11	3844	3557	542.17 ± 92.39	4.26 ± 0.80	7.17 (F)
37	193.83	190.14	2072	1909	291.04 ± 48.96	1.54 ± 0.29	7.18 (M)
40	185.52	182.25	1600	1469	224.00 ± 38.24	$1.24 \pm 0.23$	8.02 (M)

The deviation of individual mothers foraging efficiency from average foraging efficiency was positively correlated to the deviation of individual pups mass to average mass at weaning (p = 0.0078,  $R^2 = 0.41$ , Figure 5.5). The relationship in which the foraging efficiency of mothers was corrected for their size (mass-specific foraging efficiency) was also positive and significant (p = 0.02), but did not explain as much of the variation in the data ( $R^2 = 0.29$ ). Relative body condition of the pups (as expressed by the deviations compared to the sex specific average mass at weaning), was however negatively correlated to the number of hours females spent diving at sea (p = 0.0067,  $R^2 = 0.36$ ) and positively related to rate of energy gain per min of diving in kJ/min (p = 0.0050,  $R^2 = 0.49$ , Figure 5.6). The relationship was also significant with foraging trip duration, but was not as tight (p = 0.0166,  $R^2 = 0.28$ ).



Figure 5.4: Growth of 36 individual Antarctic fur seal pups (n = 23 for males and n = 13 for females) from birth to weaning on Pointe Suzanne colony, Kerguelen Island in the breeding season 2012. Twenty of these pups belonged to mothers I tracked at sea. Dashed lines represent Von Bertalanffy growth curves fitted over individual pups during the nursing season and the solid lines are the average growth curve for all the pup of each sex. Plotting symbols are unique to individual pups).

Table 5.3: Standardized regression coefficients (SRC), minimum and maximum 95% confidence interval, bias and standard error (SE) of the sensitivity analysis on the calculated foraging efficiency from energy expenditure at sea (EE in MJ), prey capture attempts (PrCA), mass (g), energy density (ED in kJ/g) and relative proportion in the diet (Prop.) of myctophids (Myct.) and cephalopods (Ceph.). I omitted prey groups with SRC below 0.1 in the table.

Parameters	SRC	Min 95%Cl	Max 95%Cl	Bias	SE
EE	-0.576	-0.586	-0.564	0.000	0.006
PrCA	0.687	0.675	0.697	0.001	0.006
Myct. Mass	0.332	0.320	0.341	0.001	0.005
Myct. ED	0.103	0.094	0.111	0.001	0.004
Myct. Prop.	0.050	0.041	0.058	0.000	0.004
Ceph. Mass	0.221	0.211	0.229	0.000	0.005
Ceph. ED	0.145	0.135	0.154	-0.001	0.005
Ceph. Prop.	0.166	0.157	0.174	0.000	0.004



Figure 5.5: Deviation of individual Antarctic fur seal pup mass at weaning from average pup mass at weaning as a function of the deviation of foraging efficiency of individual pup's mothers from the average mothers foraging efficiency over one foraging trip during the 2012 breeding season on Kerguelen Island. Orange dots are male pups and green dots are female pups. Solid lines show the type II linear regression output (Y = 1.38 X - 0.51, p = 0.0078, R<sup>2</sup> = 0.41, n = 15. Spearman rank correlation rho = 0.62, p = 0.032.)



Figure 5.6: Relationship between relative mass of pups at weaning (calculated as deviation from mass-specific average pup mass at weaning) and time mothers spent diving during their foraging trip at sea in h (graph A, p = 0.0067,  $R^2 = 0.36$ ), and energy gain per min spent diving (Graph B, p = 0.0050,  $R^2 = 0.49$ ). Time at sea was also linked to pup mass at weaning (p = 0.0166,  $R^2 = 0.28$ ).

## 5.5 Discussion

I calculated foraging efficiencies of individual females from quantitative measures of at-sea energy expenditure and energy gained while foraging, and linked this to time-activity budgets and pup growth as an index of reproductive success. Foraging efficiencies differed between individuals irrespective of their time-activity budgets. Most notably, individuals generally attained high foraging efficiencies by increasing their rate of prey capture attempts rather than by decreasing the energy they expended at sea. Pups of females that were relatively more efficient at foraging were bigger than average at weaning, which implies that pup growth rates reflect the quality of their mothers as foragers. This direct link between individual efficiencies at foraging and reproduction success via maternal investment provides empirical support for the Optimal Foraging Theory.

# 5.5.1 Estimation of foraging efficiency (FE)

Antarctic fur seal females gained more energy on average than they spent during their foraging trips at sea (all ratios > 1) with an average FE of 3.4. This value was much lower than the gain/cost ratio of 23 calculated for northern fur seals (Costa, 1993) or 20 for semi-captive Steller sea lions foraging on simulated high-density patches (Goundie et al., 2015). The differences reflect a much greater rate of energy gain of 406 kJ/min of diving for northern fur seals or ~ 438 kJ/min for Steller sea lions compared to my estimate ~131 ± 16 kJ/min for Antarctic fur seals. My results are more consistent with foraging efficiencies calculated for northern elephant seals (FE of 4), California sea lions (FE of 4 and energy gain of 112 kJ/min) and the same semi-captive Steller sea lions feeding on a simulated low preydensity patch ~ 206 kJ/min (Costa, 1993; Goundie et al., 2015).

This 6-fold difference between different calculated foraging efficiencies shows that either changes or errors in estimating costs or gains of foraging can greatly affect the final results. Costa (1993) estimated foraging benefit from water influx using H and O stable isotopes, which can overestimate prey ingestion (Nagy and Costa, 1980), and did not provide information on what type of prey was ingested (they estimated a diet of squids) or their size. Steller sea lions in the Goundie et al. (2015) study were also trained to dive at specific depths and wait to obtain prey via a feeding tube. Energetic costs and gains were controlled and were thus accurately measured, but energy spent by these animals was most probably significantly less compared to wild counterparts that must search for their food. It is thus important to identify and estimate sources of uncertainties around the results of foraging efficiency values.

Foraging efficiency was estimated from foraging costs (using the DLW method) and from foraging success (energy gain) on free-ranging animals by combining the number of prey capture attempts at sea with an estimation of diet composition, mass, and energy density of prey from scats and from previously published data. Each of these parameters have inherent errors. For example, the DLW method is known to be accurate at the population level, but to either over- or under-estimate metabolic rate at the individual level (Sparling et al., 2008; Dalton et al., 2014b). As I calculated FE individually for each female, the low precision could have affected my final results.

Estimation of diet composition from scat hard-parts is also biased by differential digestion and retention rates depending on prey consumed and on meal sizes, and by the fact that it represents only the last 48 h before the seals come back to land (where the scat is collected). It is also limited by the assumptions underlying the presence-absence methods of calculating relative proportion of prey in the diet (SSFO) (Bigg and Fawcett, 1985; Jobling and Breiby, 1986; Olesiuk et al., 1990; Harvey and Antonelis, 1994; Tollit et al., 1997; Bowen, 2000; Arim and Naya, 2003; Gudmundson et al., 2006). The degree of digestion of different hard parts also incorporate errors in the estimation of prey mass from size of fish otoliths, squid beaks or other hard parts (Tollit et al., 1997; Bowen, 2000). It is known that fish energy density varies by age, season and year (Anthony et al., 2000; Vollenweider et al., 2011) but I used a fixed value in my estimates.

Finally, detection of prey capture attempts from acceleration has been used for several species of top marine predators (Skinner et al., 2009; Suzuki et al., 2009; Viviant et al., 2010; Watanabe and Takahashi, 2013; Ydesen et al., 2014). However, validation studies on captive or free-ranging otariids using cameras simultaneously to accelerometers showed that the accuracy of head acceleration to detect PrCA depends on animal behaviours, type of foraging (benthic versus pelagic) and on type/size of prey ingested (Viviant et al., 2010; Volpov et al., 2015).

I accounted for these aforementioned errors as much as possible by using correction factors when available or by including uncertainties in the final calculations (see Material and Method section). The sensitivity analyses performed over the final estimates of foraging efficiency revealed that errors associated with prey capture attempts affected the final result the most. This is not surprising given that different validation studies did not consistently agree on their detection and false positive rates (Viviant et al., 2010; Volpov et al., 2015), and provided overall wide ranges that were incorporated into my calculations. Errors associated with mass of the prey affects FE more than energy density or relative proportion of prey in the diet, at least given the errors I estimated from the bootstrap method. Ultimately, all parameters related to foraging success tended to overestimate foraging efficiency, but this was partially compensated by the error associated with the doubly-labeled water method, which affects the final estimate in the opposite direction. Thus, it is not surprising that errors in foraging success affected the final foraging efficiency more than errors in cost given the larger number of parameters needed to calculate it. Consequently, I advise giving greater care to estimating parameters related to energy gain given the overall higher risk of error around foraging success compared to foraging expenditure.

#### 5.5.2 Foraging efficiency and pup growth

Despite all the sources of uncertainties listed above, foraging efficiencies of Antarctic fur seal females were positively related to the relative body size of their pups at weaning (Figure 5.5). Thus, foraging efficiency reflected maternal investment in the pup, or the extra energy available to allocate to reproduction (even though all energy available might not all be allocated to reproduction, Fedak et al., 1996). It is also important to remember that I measured foraging efficiency over a single foraging trip and that I compared it to overall growth of pups over the entire nursing season until weaning. The fact that the relationship between quality of mothers as foragers and the pup size at weaning is still significant therefore suggests 1) that it is extremely robust, and 2) that individual females were consistently good or poor foragers throughout the breeding season.

The mass of pups at weaning has been shown for a number of species to correlate to its chances of surviving the first year at sea (Doidge and Croxall, 1989; Baker and Fowler, 1992; Boltnev et al., 1998; McMahon et al., 2000; Hall et al., 2001; Beauplet et al., 2005), and thus by extension to fitness of mothers. Consequently, the robustness of the relationship between mother's foraging efficiency and its pup growth provides grounds for the optimal foraging theory. Fitness of an animal in its evolutionary sense should be assessed over its lifetime, but I only looked at the link between foraging efficiency and an index of reproductive success over one single reproduction cycle. I am aware that one reproduction and survival over their lifetime (Boyd et al., 1995a), but parents are usually consistent in their quality as foragers over years with few individuals producing a large portion of the next generations in top marine predators (Pomeroy et al., 1999; Beauplet et al., 2006; Lescroël et al., 2010). Consequently, if I apply these findings to this study, fur seal females that were

better foragers during the breeding season would consistently produce bigger pups more capable of surviving their first year at sea.

# 5.5.3 Effect of time-activity budget

It is difficult to assess which behaviours are 'optimal' because the mechanisms by which an animal can 'optimize' them are unclear. It is also unlikely that animals consciously behave to optimise energy gain per unit of energy expended at all time. In any case, flexibility in strategies reflects individual variability and the wide ranges of their adaptive behaviours to environmental conditions, food availability and accessibility. Fur seal females typically display this wide variation in foraging behaviours and time-activity budget at sea (Guinet et al., 2001; Lea et al., 2002b). In my study, foraging trip duration ranged from 2.5 to 15.5 days, distance traveled 225 to 1295 km, and time spent diving or transiting from 22 to 34 % or 15 to 43 % of their foraging trip. These are within the ranges of previously reported values (Gentry and Kooyman, 1986; Lea et al., 2002b; Bailleul et al., 2005), but translate into a 7 - 8-fold difference between the minima of 15-16 h to the maxima of 105 - 125 h allocated to different activities between individuals.

Given the difference in metabolic rates associated with different types of activities (Chapter 2), time activity budgets would be expected to affect the foraging efficiency. However, I found no statistical relationships between foraging efficiency and time at sea or with any metrics of time-activity budgets, which indicates that quality of females depended more on individual capacities rather than on time spent performing different activities at sea. The more efficient females in my study attained greater foraging efficiencies by having a greater rate of energy gain per min of dive time rather than by reducing their energy expenditure (which translated into rate of energy gained per min also being positively related to rate of pup growth). Females that were better at extracting food from their environment during dive time irrespective of energy costs produced bigger pups at weaning. This provides a direct quantitative linkage between the quality of females as foragers (*i.e.*, at catching prey per unit of time spent diving), and their quality as breeders.

Although time activity budgets did not directly affect foraging efficiency, it did impact pup growth rates. Most notably, females that spent less time diving during their foraging trips and less time foraging produced relatively bigger pups at weaning. Foraging trip duration is a common measure of foraging effort in fur seals (Costa et al. 1991, Francis et al. 1998, McCafferty et al. 1998, Georges & Guinet 2000). In my study as in others (Lunn et al., 1993), foraging trip duration was negatively related to relative pup mass at weaning, although the relationship I found was not as strong as with foraging efficiency, time spent diving, and rate of energy gain while diving (which were overall all better indexes of pup growth rates). It is interesting to note that trip duration (or time spent diving at sea) and foraging efficiency (or rate of energy gained while diving) were both linked to pup growth— but that trip duration or time diving were not related to foraging efficiency. This suggests that the two currencies that shape maternal investment in offspring and thus pup growth (*i.e.*,time and energy are important currencies for maternal investment in terms of diving (time spent diving and rate of energy gained while diving) and overall trip duration (foraging efficiency and time at sea).

It is difficult to tease apart whether females spending less time at sea produce bigger pups because they feed their pups more frequently, or because they are more efficient foragers and return to land with greater energy overhead to allocate to feeding their pups, or both. In my case, foraging efficiency and feeding frequency (determined from trip duration) were both related to the size of pups at weaning, but foraging efficiency was a more accurate predictor. Marine mammals with a high maternal investment such as otariids are thought to optimize the frequency of feeding their offspring rather than their foraging efficiency to increase their success as reproducers (Costa, 1993; Boyd, 1999), but both could be confounding factors that relate to the quality of individuals in terms of efficiency at acquiring prey.

The time that mothers allocate to different activities while foraging varies with resource availability (Gentry and Kooyman, 1986; Trillmich and Ono, 1991). Antarctic fur seals, like all otariids, have an expensive reproductive system (Costa, 1991). Northern fur seal mothers consume 80% more food than non-lactating females (Perez and Mooney, 1986), and lactating Steller sea lions consume 70% more (Winship et al., 2002). The high energy requirements of lactation can likely only be sustained in highly productive areas with

concentrated and predictable high-energy content prey (Costa, 1993; Boyd et al., 1995a). This also means that lactating fur seals operate closer to their metabolic ceiling and might be physiologically and behaviourally limited in their capacities to adapt to drastic changes in environmental conditions (Costa et al., 1989; Costa, 2007). This is consistent with the females in my study attaining higher foraging efficiencies by increasing their rate of energy gain rather than by decreasing their energy expenditure.

## 5.5.4 Effect of phenotypic traits

Phenotypic traits that facilitate foraging efficiency should increase fitness of the animals if the efficiency with which mothers capture prey is the ultimate determinant of weaning mass and pup survival. Bigger females or females with better body conditions are thought to be better foragers, as they can dive aerobically for longer (have a higher ADL, Kooyman, 1989) and might be better able to buffer the cost of lactation through higher energy stores (Iverson et al., 1993). Foraging efficiency has been routinely estimated by measuring foraging success through changes in body mass during foraging trip (Gentry and Kooyman, 1986; Lea et al., 2006; Luque et al., 2007). However, the foraging efficiency values I found were not related to body mass, changes in body mass, or changes in body condition indices during their foraging trips. It is well known that female mass fluctuates during nursing bouts (Gentry and Kooyman, 1986; Arnould, 1997), but I could neither control for, nor estimate, how long the females had been on land with their pups prior to capture. Consequently, my measures of female body masses were not standardized, which probably skewed my results and explains why foraging efficiency was not related to mass metrics. Mass of female fur seals is also related to age (which was unknown in my study) and thus experience. Foraging efficiency is thought by some to increase with time as young and small females gain body mass and experience (Lunn et al., 1994).

The fact that mass-related metrics did not relate to foraging efficiency could also indicate different strategies in energy allocation between different essential physiological functions for females. Fur seals are income breeders, which means that they do not accumulate and store all the energy they need to provide their pups prior to the breeding season, but rather rely on energy obtained during frequent foraging trips within the nursing season (Boyd, 2000). In this case, animals have to determine energetic priorities between conflicting functions such as growth, maintenance, and reproduction during the breeding season itself. The uncoupling I observed between changes in body mass and foraging efficiency might indicate that some females compromised the growth of their pups to the benefit of their own physiological functions, or that some females might actually supplement the energy they acquire from foraging trips with limited body reserves, while others do not. This is for example the case of Blue petrels (*Halobaena caerulea*) which alternate long trips to replenish their reserves with shorter trips close to the nest to feed their chick during which they lose mass (Weimerskirch et al. 2003).

The lack of a relationship between the mass of females and their foraging efficiency is consistent with previous studies of Antarctic fur seals performed during years with favorable environmental conditions (Boyd et al., 1991; Lea et al., 2006). It indicates that 2011 was not a particularly challenging year for lactating fur seals. However, while maternal size did not contribute to differences in foraging efficiency between individuals in years of high food availability, it did positively influence pup growth rates during years of bad environmental conditions (McCafferty et al., 1998; Lea et al., 2006). This means that the physical advantages of larger females—that are probably older and more experienced—makes a difference during years when environmental conditions are poor (Lunn et al., 1994; Lea et al., 2006) because accessibility of prey is likely to be more challenging and females are more likely to be foraging closer to their metabolic limits (Boyd, 1999; Costa, 2007). A similar conclusion has been drawn for Adélie penguins (Lescroël et al., 2010) for which better foragers only held a reproductive advantage during challenging years. On the other hand, foraging costs become greater for individuals of larger size during years with normal conditions (Massardier-Galatà et al., In review). Consequently, the evolutionary pressure dictated by optimal foraging theory might select specific heritable phenotypic traits such as size, but the fact that these phenotypic traits might only become an advantage during years of challenging conditions could explain why there is so much variability in the population.

# 5.5.5 Conclusions

My results show that quantitative measures of maternal foraging efficiencies and offspring growth rates in free-ranging marine mammals provided empirical support for the optimal foraging theory. Direct energetic links between maternal investment and maternal foraging behaviours and efficiencies can help infer the fitness of individuals and the dynamics of populations. My findings further provide a quantitative energy-based framework to investigate and model the impacts of hypothetical and forecasted environmental and preyrelated changes on the behaviours, and energetic costs and benefits of foraging by individual animals.

# Chapter 6: Lactating females in a declining population of northern fur seals compromise either on foraging efficiency or pup feeding rates during the breeding season

## 6.1 Summary

Understanding the efficiency (energy gain/cost ratio) of foraging can shed light on how animals cope with environmental changes and how such changes affect population trajectories. I determined the foraging efficiency (energy gained to energy spent ratio) of foraging trips made by lactating females from a declining population of northern fur seals (Callorhinus ursinus) breeding on St. Paul Island in the Bering Sea to gain insights into why they are declining. I captured and equipped 20 females with tags that recorded GPS locations, depth and triaxial acceleration to determine at-sea foraging behaviours and detailed time-activity budgets. Energy expenditure for each foraging trip (the cost) was measured using the doubly-labeled water method—while the energy gained while foraging (the benefit) was measured by 1) determining the types and energy densities of prey consumed (based on scat remains and blood C and N stable isotopes), and by 2) estimating numbers of prey capture attempts from head acceleration data. I also collected milk samples from 10 of the tracked fur seals and randomly selected other pups three times during the breeding season to determine growth trajectories. The 20 tracked fur seals employed two foraging strategies. Eight of the females fed mostly on small high energy-density prev (5.75  $\pm$  1.44 kJ/g or  $194.64 \pm 8.49$  kJ/prey) in deep waters off the Bering Sea shelf—while the 12 others stayed on the relatively shallow Bering Sea shelf feeding mostly on larger fish of lower quality  $(3.89 \pm 0.74 \text{ kJ/g or } 324.47 \pm 10.40 \text{ kJ/prey})$ . Females foraging off the shelf made 3 times more prey capture attempts on average than females foraging on the shelf, which meant that off-shelf females had consistently greater foraging efficiency compared to on-shelf females (*i.e.*, benefit/cost =  $1.44 \pm 0.71$  on shelf vs.  $3.02 \pm 0.60$  off shelf). The females that fed in the Bering Sea basin obtained 1.61 - 5.27 more energy than they expended, but some of the

females that foraged over the shelf failed to meet their costs of obtaining prey. However, the on-shelf females that had the lower efficiencies also made comparatively shorter trips, which likely resulted in their pups being fed  $\sim 20\%$  more frequently. Foraging strategies with lower efficiencies or with lower nursing rates meant that either lactating females reduced the available energy they transferred to their pups or they reduced the time they allocated to feeding them. Both options resulted in less overall energy being transferred to their offspring over a breeding season and would have resulted in pups attaining a lower mass at weaning and incurring higher mortality during their first year at sea. This need for lactating females to have to compromise on one or the other of these parameters provides an important clue into the decline of this population.

#### 6.2 Introduction

A central concept in life-history theory is that organisms must optimally allocate resources between the competing demands of reproduction, body maintenance, and growth in order to maximize survival and reproductive success. Changes in environmental conditions that result in sub-optimal energy intake will likely result in animals modifying how they allocate energy. In long-lived animals, self-maintenance has priority over reproduction because fitness, and lifetime reproductive success depend on longevity Bouten et al. (1994). Hence, reductions in energy intake or foraging efficiency are expected to affect breeding performances before it affects the adult survival. However, the extent to which reproduction is impaired by food availability depends in part on the flexibility and capacity of animals to acquire energy (foraging efficiency) and allocate it (physiological plasticity). Consequently, how well an animal adjusts to localized environmental changes and the availability and distribution of prey will determine its body condition, fitness and capacity to survive and successfully reproduce.

Northern fur seals breeding in the central Bering Sea have been declining by 6% per year for the last 10-15 years (Towell et al., 2006), as have other top predators in this area (Wanless et al., 2005). Female fur seals sustain high energy output while nursing their pups for 4 months by making frequent foraging trips. The breeding season is a crucial period in the life history of otariids when foraging success of mothers determines the quantity of

energy available to transfer to their pups and ensure their survival (Reid and Forcada, 2005). Marine predators such as otariids utilise areas of high productivity that are linked to higher and more predictable prey concentrations (Costa, 1993).

In the Bering Sea, major environmental changes starting in the mid-1970s altered the distribution and relative abundances of different species of fish (Anderson et al., 1997; Anderson and Piatt, 1999). The main prey available to top marine predators in the Bering Sea switched from mainly high energy fish to mainly low energy-density prey (Trites et al., 2007; Sinclair et al., 2008). Free-ranging predators usually buffer changes in the availability of prey by changing their foraging areas. However, northern fur seals are limited in where they can search by the fasting constraints of their young. They are also limited by time to optimise their foraging and provide enough energy to their pups before they abruptly wean. As a consequence, localized oceanographic variations within the foraging range of northern fur seals during the breeding season likely affect foraging success of fur seals and the rearing of their pups.

When faced with environmental changes, animals have to employ strategies that balance the energetic costs and time spent foraging, against the potential energy gained from prey. Foraging efficiency is the difference between the energy gained through prey intake and the energy spent to extract the prey from the environment. Estimating foraging efficiency requires that foraging costs and energy obtained be measured at the same time. Unfortunately, foraging efficiency has not been traditionally measured, but has rather been only inferred by comparing the change in mass before and after a foraging trip (Gentry and Kooyman, 1986; Lea et al., 2006; Luque et al., 2007), which is insufficient (Insley, 2008).

The foraging efficiency of lactating fur seals may have declined and reduced the fitness (i.e. reproductive success) of the breeding females in the central Bering Sea compared to pre-1970s. If so, it may explain the declining population trend of northern fur seals. There is therefore a need to determine, at the finest scale possible, the foraging and feeding behaviours of individual fur seals to accurately link specific environmental parameters, fitness, and demographic parameters. It is rare that both sides of the 'efficiency equation' have been studied on same individuals (Costa, 1993). Estimation of field metabolic rate (FMR) with the doubly-labelled water (DLW) method was validated for pinnipeds

(Speakman, 1997; Butler et al., 2004; Sparling et al., 2008) but only seldom used on northern fur seals since the 1980s on very few animals (Costa and Gentry, 1986; Costa, 1987). Feeding success, or the energy gained while foraging, is more difficult to measure. Proxies for feeding success such as changes in mass and body composition, changes in buoyancy during drift dives, changes in stomach temperatures, transit rate, dive shape and bout structures have all been tried and validated to some extent on marine predators under certain conditions (Hindell et al., 2010), but none of these techniques provide a quantitative measure of energy gain. The development of high sampling rate of multiple channels and the use of tri-axial accelerometers can provide indices of prey capture attempts which when paired with an estimate of diet composition can provide a quantitative estimate of energy gained during a feeding trip. Together with measures of foraging effort, foraging efficiency of individuals can be calculated.

Comparing foraging efficiencies relative to foraging strategies can further shed light on the trade-offs animals make between spending and gaining energy. It is these trade-offs that likely impact the energy reserves that mothers have to feed their pups, which in turn influences their chances of surviving their first year at sea (Doidge and Croxall, 1989; Baker and Fowler, 1992; Boltnev et al., 1998; McMahon et al., 2000; Hall et al., 2001; Beauplet et al., 2005). In this context, the objectives of my study were to 1) determine the foraging strategies of lactating northern fur seals, including locations, diving patterns and targeted prey, 2) measure foraging efficiency (foraging gain/cost ratio) and determine how it relates to foraging strategy, and 3) assess how different foraging strategies and efficiencies might be related to reproductive success and the declining trend of the fur seal population breeding in the central Bering Sea.

#### 6.3 Material and methods

#### 6.3.1 Data collection

Data were collected from 20 healthy lactating northern fur seals at Reef rookery on St. Paul Island (Bering Sea, 57° 6 'N - 170°17' W) during the breeding season (Aug-Sep 2011) under US NMFS permit # 14329-01 and UBC animal care permit # A10-0364. Data collection procedures were similar to those reported for Antarctic fur seals in Chapter 5.

#### 6.3.2 Energy expenditure, foraging behaviours and prey capture attempts

Analyses and calculations of the energy expenditure, foraging behaviours and prey capture attempts were done as per Chapter 5. In brief, measurements of daily energy expenditure (DEE, kJ/day) were performed using the DLW method (Lifson and McClintock, 1966; Butler et al., 2004). Energy spent during the 'non-foraging' time was corrected by subtracting on-land energy expenditure from the total estimate using previously determined values specifically for lactating northern fur seals (4.67 W/kg, Chapter 5 in Gentry and Kooyman, 1986). The females were also equipped with Daily Diary tags (DD, Wildlife Computers) recording tri-axial acceleration and tri-axial magnetic field at 16 Hz, and depth, light level, and water temperature at 1 Hz and Fastloc GPS MK10 loggers (Wildlife Computers) recorded GPS coordinates along the track of the animal at sea, as well as depth and water temperature at 1 Hz. Finally, GCDC X6 or X8 accelerometers glued on the head of the animals recorded tri-axial acceleration at 16 or 20 Hz. Prey capture attempts (PrCA) were measured while the animal was diving below 3 m using acceleration data recorded on the head of the animals (Skinner et al., 2009; Suzuki et al., 2009; Viviant et al., 2010; Watanabe and Takahashi, 2013; Ydesen et al., 2014).

Statistical differences between groups (fur seals that stayed and fed over the continental shelf vs. those that travelled off the shelf) were tested with two-sample *t*-tests ( $\alpha = 0.05$ ) or Mann-Whitney tests depending on normality. Averages for dive parameters, such as for dive depths and dive durations, were nested within animals and were calculated using linear mixed effect models with no fixed effects (only the intercept was calculated) and with individual as a random effect to account for each animal performing a different number of dives. Differences between prey capture attempts per night or per dives between on-shelf and off-shelf groups were estimated using linear mixed effect to account for each animal effect models with group (on-shelf or off-shelf) as the fixed effect, and individual as a random effect to account for their difference in trip duration.

#### 6.3.3 Diet estimates

The female northern fur seals breeding on Reef Rookery had two diets that differed depending on whether they feed on-shelf or off-shelf (see Chapter 4). I applied these two diets to the respective foraging strategies of the individuals tracked, and used hard part remains recovered in fecal samples collected at Reef rookery in 2011 to determine size of prey consumed and estimate energy density of the two diet types. I also relied on hard parts recovered from scats collected at Bogoslof Island in 2009 as a proxy for diets of fur seals that feed off-shelf (see Trites et al. 2015). This method is less accurate than DNA-based methods, but is the only method that provides indication of prey size. As I only have DNA-based results for the scats collected on St. Paul Island, working with hard part remains only allows results to be standardized between the estimated on-shelf and off-shelf diets (consistency of methods and associated biases). Consequently, females foraging on the shelf were considered to feed on a diet with a composition estimated from the scats I collected on Reef rookery in 2011 while the females foraging off the shelf were assumed to feed on a diet similar to the one estimated on Bogoslof island in 2009 (Chapter 4, Trites et al., 2015).

I first estimated the energy densities of the different diets using the relative proportion of prey items in each diet and the energy densities of the different species from the literature (Perez, 1994; Van Pelt et al., 1997; Paul and Paul, 1999; Logerwell and Christiansen; Iverson et al., 2002; Gende et al., 2004; Andrews et al., 2009; Whitman, 2010; Vollenweider et al., 2011). Energy density of walleye pollock differs by age class (Paul and Paul, 1999), so I obtained the overall energy density of pollock (85% of age-0 at 3.36 kJ/g and 15% of age-1+ at 4.73 kJ/g, Whitman, 2010) by accounting for the proportion of each age class in the diet (from their size, see below). Salmonids were not identified at the species level from hard part remains, so I averaged the energy density of several species found in Alaskan water (Hendry and Berg, 1999; Anthony et al., 2000; Rosen and Trites, 2000; Gende et al., 2004) weighted by the proportion of different species identified though the DNA metabarcoding method (Chapter 4). Energy density of squids in the Eastern Bering Sea vary by region (Whitman, 2010). Squids caught in the shelf slope area have a higher average total energy density than both squids in the middle and the outer shelf areas. However, specific values were not available by region, so I considered squids to have the same average energy

density wherever the fur seals fed. I estimated the variation in these parameters by reporting ranges of prey body mass and ranges of published energy density values (min and max values calculated as mean  $\pm$  SD).

Whenever information was missing for prey with low frequencies of occurrence in the diet, I replaced it by energy density of the closest related prey item or by the average of the energy content for the specific prey group. I obtained energy density in diet  $(ED_{Diet})$  by averaging the energy density of different prey  $(ED_i)$  weighted by their proportion within the diet  $(P_i)$  over the number of prey in the diet N:

Eq. 6.1 
$$ED_{Diet} = \frac{\sum (ED_i \times P_i)}{N}$$

The second step in estimating the energy densities of the two diets was to estimate the gross energy intake in kJ for each prey ingested. I did this by estimating size of prey ingested using hard part remains in the collected scats. I relied on all identifiable bones to estimate number of prey items in each scat, but only used unbroken otoliths in good or fair condition to estimate fish length (Tollit et al., 2004). Lengths of otoliths were measured using the method described in Harvey et al. (2000), and correction factors were applied to these measured lengths to account for digestion whenever a species-specific correction factors were available (Tollit et al., 2004) (*i.e.*, 1.1691 for Atka mackerel and 1.1593 for walleye pollock). Measured otoliths of the same size ( $\pm 0.1$  mm) were paired and assumed to come from the same fish (Hunt Jr et al., 1980; Dragoo, 1991).

Squid beaks were measured using guidelines from Wolff (1984). Northern fur seals are known to mostly consume squids from the gonatid family (*Gonatopsis borealis and Berryteuthis magister*) in the Bering Sea, more so than other squids from the *Gonatus* genus (Gudmundson et al., 2006), but I did not identify squid beaks to the species level from hard part remains. I therefore referred to squids as gonatid squids and used the general size equation available for this family (Clarke, 1962; 1986). Squid beaks did not require a correction factor because there are much more resistant to digestion than bones (Tollit et al., 1997).

Fish otolith lengths (OL) or squid lower beak rostrum lengths (LRL) were converted to fish/squid length (FL) then body mass (BM) using the species-specific equations listed in
the Appendices section. For species of fish found in the scats that were identified by other hard parts than otoliths (vertebrae and gill rakers, etc.), the range of sizes was assessed by comparing the size of hard parts with those of reference skeleton from Pacific ID Inc.). These range of sizes are summarized in Table 1 and Figure 6 from Trites and Calkins (2008). I estimated average size  $\pm$  SE of these species fish by simulating random selection of fish sizes from the range of sizes in each category using a uniform distribution (1000 simulations). For example, the hard part remains of salmon showed 55 % of them were between 16 and 24 cm, and 45 % were between 35 and 59 cm. I thus estimated that the average salmon would be  $31.75 \pm 0.14$  cm (and thus determined an average mass of  $797.78 \pm 8.92$  g from the length to mass conversion equation, see Appendices section) from the 1000 simulations. The same analysis was done for arrowtooth flounder, sand lance and rockfish.

Once the mass (BM in g) and the energy density (ED in kJ/g) of each prey item (i) were estimated, I calculated the average energy content of a specific fish (EC in kJ) as follows:

$$EC_i(kJ) = BM_i(g) \times ED_i(kJ/g)$$

The average energy content (in kJ) of a random non-specific prey (p) consumed by fur seals given a specific diet was calculated by weighting the energy content of a specific prey item by its relative proportion in the diet (P).

$$EC_p(kJ) = \sum_i (EC_i(kJ) \times P_i)$$

Means  $\pm$  SD of energy content of each prey ( $EC_i$  in kJ) were calculated by generating 1000 values of mass and 1000 values of energy density ( $ED_i$ ) using normal distributions of their respective means  $\pm$  SD. Means  $\pm$  SD of energy densities ( $ED_{Diet}$ ), and energy content of an average prey ( $EC_p$ ) in the two diets were calculated by generating values of  $EC_i$  and  $ED_i$ for each prey type (*i*) in proportion to their respective importance in the diet ( $P_i$ ) out of 1000 values from normal distributions using their respective mean  $\pm$  SD. For prey species with no  $ED_i$  or  $EC_i$  values, I used the average ED or EC from the prey group (e.g., gadid, cephalopod, etc.) as their values weighted by their own proportion within the diet. As the prey group 'Other' did not have values for mass or energy density, I considered it as an average of the rest of the diet weighted by its relative importance in the diet.

# 6.3.4 Foraging efficiency

The foraging efficiency (*FE*) of each seal (*i*) was calculated as in Chapter 6. In brief, *FE* is the ratio between the energy expenditure at sea obtained from the DLW measures (*EE<sub>i</sub>*) per animal *i* and energy gained while foraging at sea. Energy gained was estimated as the specific energy content of a non-specific prey (*EC<sub>p</sub>*) in their diet multiplied by the number of time seals *i* attempted to capture prey (*PrCA<sub>i</sub>*):

Eq. 6.2 
$$FE_i = \frac{EC_p \times PrCA_i}{EE_i}$$

I calculated uncertainty around  $FE_i$  using Monte Carlo simulations that reflected errors in the 3 equation parameters as described in Chapter 5.

I also used simple liner regression models to test whether mass, body condition (*i.e.*,mass/length ratio) or change in mass during the foraging trips were linked to foraging metrics such as time spent at sea, diving rate or rate of energy gain.

# 6.3.5 Milk energy content

Milk samples were opportunistically collected while females were under anaesthesia. I obtained milk in quantity sufficient for proximate composition analyses (> 3 ml) in only 10 of my 20 females (5 before their foraging trip and 5 after). After collection, milk samples were stored in the freezer at -20°C until analyses. Proximate composition analyses were performed by SGS Canada Inc. Moisture, ash, protein and fat content of milk samples were measured using the AOAC 935.29, 942.05, 990.03, and 989.05 methods respectively. Energy density of milk samples were calculated using the Atwater method. Relationship between time at sea and milk fat content prior or after foraging trip and trip duration was estimated using simple linear regressions.

### 6.3.6 Pup growth

I captured the pups associated with the tracked females before the mothers left the rookery to forage, and took standard morphometric measurements of length and girth to the nearest 0.5 cm and  $\pm$  0.1 kg. Unfortunately, field conditions made it impossible to follow the growth of individual pups associated with each tracked mother. Consequently, I resorted to

capturing random pups on the colony as an index of pup growth over the breeding season. Eighty pups of both sexes were selected at random and morphometric measurements were taken 3 times over the course of the breeding season: between the 18th and the 21<sup>st</sup> of August 2011, between the 7<sup>th</sup> and the 10<sup>th</sup> of September 2011, and between the 20<sup>th</sup> and the 21<sup>st</sup> of September 2011.

Male and female pups were analysed separately as there were significant differences in mass between them (tested with a *t*-test), which is known in otariids (Mattlin, 1981; Kerley, 1985; Trites, 1993; Davis et al., 1996; Chambellant et al., 2003; Osman et al., 2010). A linear model was fit to my 2011 data for each sex individually, as well as to data from the 1980s on Bering Island ((Boltnev et al., 1998)), and from 1996 on St. Paul Island ((Donohue et al., 2000)). Differences in slopes between these models were estimated using a test for equality of regression coefficients (Paternoster et al., 1998). I estimated the significance of relationships between female mass, mass changes, and foraging efficiencies, and the relationships between pup parameters and mother parameters using linear regressions models.

### 6.4 Results

# 6.4.1 Foraging behaviours

Females breeding on Reef rookery, St. Paul Island, are known to disperse in all directions around St. Paul Island to forage (Gentry, 1998; Robson et al., 2004; Nordstrom et al., 2013). Eight females from the 20 tagged went foraging off the shelf in the deep basin of the Bering sea (up to 3000 m deep), hereafter called off-shelf females, and 12 stayed foraging on the shallow Bering Sea shelf (0 – 200 m deep), hereafter called on-shelf females (Figure 6.1). The females weighed on average  $37.9 \pm 1.34$  kg prior to departure (range 30.8 - 55.6 kg, n = 20). Overall, 12 females gained mass (0.7 - 5.2 kg gained), and 8 of them lost mass during their foraging trip (0.2 - 4.2 kg lost). This corresponds to  $3.5 \pm 1.8$  % of initial body mass (range from 9.9% lost to 16.1% gained,  $3.2 \pm 2.5$  % for on-shelf females and  $3.9 \pm 2.7\%$  for off-shelf females). Foraging trips lasted  $7.96 \pm 2.17$  d (range 4.26 - 12.03 d) and average horizontal (surface) distance traveled was ~  $750 \pm 50$  km (range 391 - 1200 km).

Females foraging off the shelf spent on average 2.5 more days at sea during their foraging and traveled 200 km more than on-shelf females (off-shelf females:  $9.7 \pm 1.8$  d and  $892 \pm 144$  km; on-shelf females:  $7.1 \pm 2.4$  d and  $655 \pm 218$  km, p < 0.009, Figure 6.1, Table 6.1).

Females foraging on the shelf regularly performed benthic dives  $(15.2 \pm 2.6 \% \text{ of all})$ dives) and occasionally foraged during the day, while females foraging off the shelf rarely performed benthic dives  $(2.2 \pm 0.4 \% \text{ of all dives and } 0 \% \text{ when off the shelf})$  and were nocturnal divers (Figure 5.3). Off-shelf females also performed significantly more foraging dives  $(3631 \pm 313 \text{ dives})$  than the on-shelf females  $(1831 \pm 374 \text{ dives}, p < 0.001)$ , but were shallower and of shorter duration (~  $17 \pm 2$  m deep and  $37 \pm 6$  s duration for off-shelf females and ~ 27 ± 3 m deep and 80 ± 8 s duration for on-shelf females, p < 0.05). In addition, ~ 73% of off-shelf female dives were shallower than 15 m, while only ~ 55 % were for on-shelf females. This translated into on-shelf seals spending significantly more time diving (~  $33.1 \pm 3.1$  % of their time at depth + recovery period) and less time transiting (28.3)  $\pm 2.3$  % of their time with a moving speed  $\geq 1$  m/sec) over the full foraging trip time scale than off-shelf females (24.7  $\pm$  0.8 % of time diving and 32.0  $\pm$  2.2 % of time transiting, both p < 0.05). In other words, females foraging off shelf travelled ~ 30 h more than on-shelf females, for the same number of hours spent diving (Figure 6.1, Table 6.1). Overall, body condition of females (mass/length) prior to foraging trip duration was negatively related to time at sea (slope p = 0.004,  $R^2 = 0.38$ , Figure 6.2).

# 6.4.2 Energetic cost of foraging

Rate of energy expenditure per day at sea was not significantly greater for females foraging on the shelf than foraging off the shelf whether mass-corrected or not (on-shelf:  $22.1 \pm 1.8$  MJ/d or  $0.56 \pm 0.05$  MJ/d/kg; off-shelf:  $18.9 \pm 2.4$  MJ/d or  $0.54 \pm 0.07$  MJ/d/kg, p > 0.07, Table 6.1, Table 6.1). When measured during a trip at sea in MJ or MJ/kg, females foraging off the shelf tended to spend more energy per trip ( $4.9 \pm 0.8$  MJ/kg) than females foraging on the shelf ( $3.7 \pm 0.4$  MJ/kg, although not statistically significant p > 0.09). However, females foraging off the shelf spent significantly more energy transiting to and from feeding grounds ( $1.6 \pm 0.2$  MJ/kg) than females foraging on the shelf ( $1.0 \pm 0.2$  MJ/kg,)



Figure 6.1: Foraging location, dive profiles over a full foraging trip, and over 5-min of diving of lactating northern fur seals from Reef rookery on St. Paul Island foraging off the shelf (n = 8, A, C, E) or on the shelf (n = 12, B, D, F) in the Bering Sea. Orange and red dots show where and when the animals attempted to capture prey within the range of their foraging location

(with yellow being no prey capture attempt, A and B), or during their dives (C, D, E, F). Females foraging off the shelf attempted to capture prey in limited and localised areas (A) while attempts to capture prey on the shelf occurred over a greater geographic area over longer periods each day (B). Females foraging on the shelf dove on average deeper and attempted to capture prey less frequently over similar periods (E, F, and partly during daytime) compared to females foraging off the shelf (C, E).

Table 6.1: Foraging and metabolic parameters of female northern fur seals (NFS) foraging on (n = 12) or off (n = 8) the Bering Sea shelf for one trip at sea. Values are mean ± SE, and letters in superscript indicate significant differences between the groups (p > 0.05). Values of similar parameters for Antarctic fur seal females (AFS) are added for comparison.

Parameters	<b>On-Shelf NFS</b>	Off-Shelf NFS	AFS
Mass (kg)	$39.4 \pm 2.0^{a}$	$35.7 \pm 1.2^{a}$	$31.1 \pm 0.9^{b}$
Mass change (kg)	09+09	14+0.9	0.6 + 0.6
Mass change (%)	$3.2 \pm 2.5$	$3.9 \pm 2.7$	$2.2 \pm 1.8$
Trip duration (d)	$7.1 \pm 2.4^{a}$	$9.7 \pm 1.8^{b}$	$7.6 \pm 3.8^{a}$
Distance traveled (km)	$655 \pm 63^{a}$	$892 \pm 51^{b}$	$635 \pm 77^{a}$
Dive number	$1831 \pm 374^{b}$	$3631 \pm 313^{a}$	$3949 \pm 597^{a}$
Dive depth (m)	$26.8 \pm 3.3^{a}$	$16.9 \pm 2.0^{b}$	$19.9 \pm 2.7^{a}$
Dive duration (s)	$79.8 \pm 8.5^{b}$	$37.3 \pm 6.4^{a}$	$42.6 \pm 4.5^{a}$
Benthic dives (%)	$15.2 \pm 2.6^{b}$	$2.2 \pm 0.4^{a}$	$0 \pm 0^a$
Time diving (%)	$33.1 \pm 3.1^{a}$	$24.7 \pm 0.8^{b}$	$29.0 \pm 0.7^{a}$
Time transiting (%)	$28.3 \pm 2.3$	$32.0 \pm 2.2$	$26.4 \pm 1.6$
Time diving (h)	$55.8 \pm 7.9$	$55.1 \pm 3.5$	$51.3 \pm 5.9$
Time transiting (h)	$49.0 \pm 7.2^{a}$	$72.2 \pm 7.8^{b}$	$49.8 \pm 7.9^{a}$
At-sea metabolic rate (MJ/d/kg)	$0.56 \pm 0.05$	$0.54 \pm 0.07$	$0.59\pm0.04$
Total EE (MJ/kg)	$3.7 \pm 0.4$	$4.9 \pm 0.8$	$4.1 \pm 0.6$
EE diving (MJ/kg)	$2.0 \pm 0.3$	$2.0 \pm 0.2$	$2.2 \pm 0.3$
EE transiting (MJ/kg)	$1.0 \pm 0.2^{a}$	$1.6 \pm 0.2^{b}$	$1.3 \pm 0.2^{ab}$
PrCA per dive	$0.46 \pm 0.09^{a}$	$0.76 \pm 0.09^{b}$	$0.87 \pm 0.1^{b}$
PrCA per capturing dive	$1.77 \pm 0.23$	$1.75 \pm 0.17$	$2.04 \pm 0.11$
PrCA per day	$100 \pm 26^{a}$	$275 \pm 39^{b}$	$336 \pm 38^b$
PrCA per trip	$789 \pm 163^{a}$	$2404 \pm 338^{b}$	$2328 \pm 387^{b}$
ED <sub>Diet</sub> (kJ/g)	$3.89 \pm 0.74$	$5.75 \pm 1.44$	$7.75 \pm 2.47$
EC <sub>Prev</sub> (kJ)	$324.47 \pm 10.40$	$194.64 \pm 8.49$	$152.46 \pm 1.08$
EGain per dive (kJ)	$176.5 \pm 48.9$	$188.3 \pm 34.4$	$177.7 \pm 21.4$
EGain per time diving (kJ/min)	$82.1 \pm 16.3^{a}$	$145.9 \pm 24.4^{b}$	$130.6 \pm 16.3^{\rm b}$
EGain per time at sea (kJ/min)	$24.9 \pm 6.1^{a}$	$35.21 \pm 5.4^{b}$	$37.6 \pm 4.6^{b}$
Foraging efficiency	$1.44 \pm 0.71^{a}$	$3.02 \pm 0.60^{b}$	$3.44 \pm 0.45^{b}$



Figure 6.2: Time spent at sea by females during a foraging trip as a function of their body condition (mass/length in kg/cm) prior to the trip at sea (in days, n = 19). Red triangles are for females foraging off the shelf, blue circles are for females foraging on the shelf and the line is for the linear regression between the 2 parameters (slope p = 0.004,  $R^2 = 0.38$ ).

#### 6.4.3 Prey capture attempts

Females foraging off the shelf attempted to capture prey  $0.76 \pm 0.09$  times per dive, which is greater than for females foraging on the shelf (0.46 ± 0.09, p = 0.004). However, when looking only at dives in which at least one PrCA occurred, both groups attempted to capture prey ~ 1.76 times per dive (1.77 ± 0.23 for females foraging on the shelf, and 1.75 ± 0.17 for females foraging off the shelf, p = 0.73). Females foraging on the shelf attempted to capture 100 ± 26 individual prey per day of foraging on average, which is equivalent to 789 ± 163 individuals per trip. This is ~3 times less than what females foraging off the shelf attempted to capture per day of foraging (275 ± 39 individual prey, p = 0.24) or per trip (2404 ± 338 individuals, p = 0.003, Figure 6.1graphs C, D, E & F). Most of the attempted prey captures occurred during nighttime (92.9 ± 2.9 % for on-shelf females and 99.0 ± 0.4 % for off-shelf females, p = 0.05). Visually, prey capture attempts for females foraging off the shelf were localised to small and specific areas (orange and red dots, Figure 6.1 graph A), while females that foraged on-shelf had greater geographic and temporal spread in feeding areas (Figure 6.1 graph B).

# 6.4.4 Diet and prey energy contents

Relative proportion of species within the diets of females foraging on or off the shelf differed (Table 6.2). Females foraging on the shelf mostly consumed gadids (~70%), cephalopod (~12%) and salmon (~6%), while diets of females foraging on the shelf (based on scats collected at Bogoslof Island from Chapter 4 and Trites et al., 2015) consisted mostly of mesopelagic fish (~49%), cephalopods (~43%) and salmon (~4%).

Analyses of prey size from otoliths, beaks and other bone structures in the scats showed that adult female fur seals prey mostly on small fish (3 - 20 g). For example, size class of ingested pollock (Figure 6.3) were mostly of age 0 and 1-year-old. The only exception was salmonids which were ~ 31 cm long and weighed ~ 800 g. Energy density of each major fish species as taken from the literature ranged from ~ 3.5 kJ/g for flatfish and gadids to ~ 9 kJ/g for northern lampfish (Table A.5). Given size of ingested fish, this translates into energy content of prey groups ranging from 20 to 75 kJ, except for larger salmon that reached ~ 4500kJ (Table 6.2). Mass, energy density and energy content of individual prey species are detailed in Table A.5 in the Appendices section.

The total energy density of diet  $ED_{Diet}$  for females foraging on the shelf was ~3.89 ± 0.74 kJ/g. Salmonids and gadids contributed the most to the energy content per prey ingested  $EC_P$ , and averaged 324.47 ± 10.40 kJ per prey for females foraging on the shelf. The energy density of the diet  $ED_{Diet}$  for off-shelf females was greater than that of the on-shelf females at 5.75 ± 1.44 kJ/g. However, due to the smaller size of prey ingested, the energy content per prey ingested  $EC_P$  was only 194.64 ± 8.49 kJ (Table 6.2).

Table 6.2: Relative percent (%), average prey mass (g), prey energy density (ED in kJ/g), energy content (EC in kJ) of prey groups in diets of northern fur seal females breeding on St. Paul, and of females breeding on Bogoslof Island rookeries. Groups 'Other' were assigned average diet values. Percent in diet (SSFO%) were obtained from morphological identification of hard part remains in my study and from collection of samples of Bogoslof Island rookeries in 2009, and mass was calculated from the size of hard part remains.

	Prey group	SSFO (%)	Mass (g)	ED (kJ/g)	EC (kJ)
<b>ON-SHELF NFS</b>	Gadid	69.75	$15.49 \pm 0.13$	$3.51 \pm 0.35$	$54.69 \pm 0.14$
	Cephalopod	12.28	$3.74 \pm 1.16$	$4.76 \pm 0.11$	$17.79 \pm 0.04$
	Salmon	6.03	$797.78 \pm 35.65$	$5.53 \pm 0.30$	$4425.00 \pm 9.80$
	Hexagrammid	3.64	$18.67 \pm 2.48$	$3.90 \pm 0.23$	$75.19 \pm 0.31$
	Mesopelagic	2.42	$3.87 \pm 0.25$	$5.67\pm0.25$	$21.91 \pm 0.05$
	Forage	1.53	$10.69 \pm 0.19$	$4.54\pm0.66$	$49.61 \pm 1.34$
	Flatfish	0.28	$3.78 \pm 0.01$	$5.14 \pm 0.65$	$19.45\pm0.08$
	Other	4.08	NA	NA	NA
DFF-SHELF NFS	Gadid	0.86	$16.48 \pm 0.11$	$3.57\pm0.30$	$58.59 \pm 0.20$
	Cephalopod	42.70	$3.74 \pm 1.16$	$4.76 \pm 0.11$	$17.79 \pm 0.04$
	Salmon	3.87	$797.78 \pm 282.00$	$5.53 \pm 0.30$	$4425.00 \pm 9.80$
	Hexagrammid	0.60	$18.69 \pm 2.37$	$4.02\pm0.08$	$75.19\pm0.29$
	Mesopelagic	48.97	$3.77 \pm 0.52$	$5.85\pm0.88$	$21.68\pm0.01$
	Forage	1.55	$16.47 \pm 0.51$	$5.47\pm0.57$	$92.28 \pm 3.14$
0	Other	1.46	NA	NA	NA

# 6.4.5 Foraging efficiency

The calculated rates of energy gains were  $176.5 \pm 48.9$  kJ/dive of females foraging on the shelf and  $188.3 \pm 34.4$  kJ/dive for those foraging off the shelf (p = 0.78). This translates in 82.1 ± 16.3 kJ/min for on-shelf diving and 145.9 ± 24.4 kJ/min off-shelf diving. In terms of the entire foraging trip, females foraging on the shelf gained 24.9 ± 6.1 kJ per min spent at sea, which was lower than what females foraging off the shelf gained ( $35.21 \pm 5.4$  kJ per min spent at sea). Based on this, the average foraging efficiency of all individuals for which I had simultaneous measures of energy expenditure and prey capture attempts (Table 6.3) was 2.13 ± 0.44 and ranged from 0.42 - 5.68. Foraging efficiency was 1.44 ± 0.71 (range 0.83 - 3.05, 95% CI: 0.86 - 2.42) for females foraging on the shelf and 3.02 ± 0.60 (range 1.61 - 5.27, 95% CI: 2.17 - 4.02) for females foraging off the shelf.



Figure 6.3: Frequency of estimated walleye pollock length (in cm) consumed by female fur seals in the Bering Sea estimated from otolith size (n = 173) found in 98 scats collected on Reef Rookery in August 2011.

It was impossible to know how long the females had been on land prior to capture, or how much weight they had lost between the time they came back to land and the time of recapture, as I did not measure the time they spent with their pups. Neither the mass of the females before (p = 0.87) and after (p = 0.19) their foraging trips, nor the change in body mass over the course of the foraging trip (p = 0.26) were related to the foraging efficiencies of the females.

### 6.4.6 Milk energy content

Energy density of milk collected from 10 lactating northern fur seal females averaged 2.06  $\pm$  0.08 kJ/g and consisted of 47.98  $\pm$  1.98 % fat and 11.48  $\pm$  0.49 % protein (Table 6.4). Energy content of the milk was slightly but not significantly greater for females that foraged off-shelf (2.13  $\pm$  0.09 kJ/g energy density, 50.50  $\pm$  2.12 % fat and 12.32  $\pm$  0.66% protein) compared to females foraging on the shelf (2.00  $\pm$  0.11 kJ/g energy density, 46.30  $\pm$ 2.93 % fat and 10.92  $\pm$  0.63 %, protein) (energy density: p = 0.41, fat: p = 0.28 and protein: p = 0.16). On the other hand, the milk of females foraging on the shelf contained greater carbohydrate and water content (4.81 $\pm$  2.77 % carbohydrates and 37.62  $\pm$  3.67 % water) compared to off-shelf females (1.60  $\pm$  0.33% carbohydrates and 35.00  $\pm$  2.65 % water), but differences were not statistically significant (carbohydrates: p = 0.29; water: p = 0.58). Half of the milk samples were taken before the foraging trip and half were taken after the foraging trip, but there was no difference in milk composition between sampling times.

Milk fat (%) in samples collected after females foraging trips (n = 5) were positively related to foraging trip duration (p = 0.007,  $R^2 = 0.93$ ), while milk fat in samples collected prior to foraging trips (n = 5) were negatively linked to foraging trip duration (although not significantly, p = 0.104,  $R^2 = 0.63$ , Figure 6.4 A and B).

Table 6.3: Measured and corrected energy expenditure (EE in MJ), measured and corrected number or prey capture attempts (PrCA) during a foraging trip for female northern fur seal foraging on (n = 9) or off (n = 7) the Bering Sea shelf. The corrected values of EE and PrCA were used with the estimated energy content per average prey (EC prey in kJ) to calculate the energy animals gained while foraging at sea (in MJ) and their foraging efficiency (i.e. the ratio of energy gain to energy expended). See section 'Foraging efficiency' in Material and Methods for how EE and PrCA were corrected.

Group	Seal ID	Meas.EE trip (MJ)	Corr.EE trip (MJ)	Meas. PrCA	Corr. PrCA	EC prey (kJ)	Energy gain (MJ)	Foraging efficiency
F NFS	1	367.43	360.55	1008	921	324.47	$298.78 \pm 52.55$	$0.83 \pm 0.16$
	5	177.37	174.16	346	319	$\pm 10.40$	$103.37 \pm 17.56$	$0.60 \pm 0.11$
	6	115.74	113.58	321	295		$95.56 \pm 16.49$	$0.85 \pm 0.16$
	8	115.07	113.01	152	140		$45.26 \pm 7.81$	$0.40 \pm 0.08$
EL	12	205.27	201.40	718	660		$214.05 \pm 36.42$	$1.07 \pm 0.20$
HS-NO	13	152.06	149.36	1681	1538		$499.01 \pm 86.48$	$3.36 \pm 0.64$
	14	440.71	432.94	635	580		$188.22 \pm 32.68$	$0.44 \pm 0.08$
	17	146.65	143.96	549	503		$163.27 \pm 27.64$	$1.14 \pm 0.21$
	19	122.60	120.56	1693	1567		$508.46 \pm 88.63$	$4.24 \pm 0.81$
	Mean	204.77 ± 39.40	201.06 ± 38.68	789 ± 189	725 ± 174			
OFF-SHELF NFS	2	249.57	244.92	2727	2505	194.64	$487.56 \pm 84.07$	$2.00\pm0.38$
	3	156.54	153.81	1474	1351	$\pm 8.49$	$262.94 \pm 46.22$	$1.72 \pm 0.33$
	4	428.07	420.02	3576	3252		$633.01 \pm 111.82$	$1.52\pm0.29$
	10	73.97	72.67	1996	1839		$357.92 \pm 61.99$	$4.95 \pm 0.93$
	11	120.03	117.89	1518	1397		$271.90 \pm 47.35$	$2.32\pm0.44$
	15	125.03	122.81	3654	3378		$657.34 \pm 112.82$	$5.38 \pm 1.01$
	20	136.22	133.86	2381	2194		$427.34 \pm 73.81$	$3.21 \pm 0.61$
	Mean	184.20 ± 45.40	180.85 ± 44.53	2475 ± 339	2274 ± 310			

Table 6.4: Proximate composition of milk samples collected on 10 lactating female northern fur seals (NFS) during the breeding season in August 2011 on Reef rookery, St. Paul Island, Bering Sea, Alaska. Six samples were collected from females foraging on the Bering Sea shelf, and 4 from females foraging off the shelf within the deep basin. Samples were either collected before (pre) or after (post) their foraging trip at sea.

Group	Seal ID	Time	Moisture (%)	Ash (%)	Carb. (%)	Protein (%)	Fat (%)	Energy (kJ/g)
S	1	pre	39.8	0.5	5.4	10.8	44.5	1.945
ON-SHELF NF	14	pre	51.1	0.7	0.2	10.6	37.4	1.59
	19	pre	30.2	0.5	3.1	11.6	54.6	2.302
	6	post	44.5	0.6	0.5	11.2	43.2	1.823
	7	post	30.6	0.3	18.1	8.3	42.6	2.047
	17	post	29.5	0.4	1.6	13	55.5	2.333
OFF- SHELF NFS	2	pre	37.1	0.6	0.8	12.5	48.9	2.066
	3	pre	39.4	0.5	2.4	11.1	46.6	1.98
	4	post	27.3	0.6	1.5	14.1	56.5	2.39
	15	post	36.2	0.5	1.7	11.6	50	2.103



Figure 6.4: Foraging trip duration of females as a function of the milk lipid content after the trip (A, n = 5) or prior to it (B, n = 5). Regression lines between the 2 parameters are added on both graphs (A: slope p = 0.007,  $R^2 = 0.93$ , B: slope p = 0.104,  $R^2 = 0.64$ ).



Age from median population birth date (d)

Figure 6.5: Mass of male (in orange) and female (in green) northern fur seal pups over the course of the nursing period, from age 0 to weaning (in days). Closed circles show data collected on random pups during the nursing season 2011 on St. Paul Island. Triangles show the mass of pup associated with the females tracked at sea during the same field season. Open circles show the mass-at-age data collected on St. Paul Island in 1996 (Donohue et al., 2000) and open squares show the mean mass  $\pm$  SD of pups weighed on Bering Island, Russia in the 1980s from Boltnev et al. (1998).

### 6.4.7 Pup growth

Female pups weighed 7.93  $\pm$  1.00 kg at age 42 - 45 days, 9.51  $\pm$  1.82 kg at 62 - 65 d, and 10.24  $\pm$  1.6 1kg at ages 75 - 76 d (Figure 6.5). In contrast, male pups were about 15% bigger, weighing 9.45  $\pm$  1.53 kg at age 42 - 45 days, 10.88  $\pm$  1.98 at 62 - 65 d, and 11.95  $\pm$  1.79 kg at 75 - 76 d (Figure 6.5). Ages were approximated based on a median date of pupping of the 6th of July (Trites, 1992). My measures of pup body mass are consistent with weights of pups from Bering Island, Russia in the 1980s (for example, 10.56  $\pm$  1.53 kg for female pups and 12.04  $\pm$  0.97 kg for males at age 70 – 80 d from Boltnev et al. (1998); Figure 6.5).

A single measure (prior to foraging trip) of mass of pups associated with mothers were not linked to foraging efficiency of mothers (p = 0.98). I could not determine if captured pups had been feeding or fasting, which created variation in the data that can heavily skew the pup mass data on single measurements. Linear models fitted to data collected in 2011 were: BM (kg) =  $0.072 \times \text{Age}$  (d) + 4.88 for female pups and BM (kg) =  $0.078 \times \text{Age}(d) + 6.08$  for male pups. Linear models fit to data from the 1980s (Boltnev et al., 1998) were: BM (kg) =  $0.072 \times \text{Age}(d) + 4.88$  for females and BM (kg) =  $0.077 \times \text{Age}(d) + 5.03$  for males. Linear models fit to data from 1996 (Donohue et al., 2000) were: BM (kg) =  $0.082 \times \text{Age}(d) + 4.67$  for females and BM (kg) =  $0.102 \times \text{Age}(d) + 5.62$  for males. Analyses of variances on the slopes of these models showed that pups in 2011 had a slower growth than pups in the 1980s or in 1996 ( $p < 10^{-5}$  in all cases).

# 6.5 Discussion

Food is not evenly distributed in space and time in the environment, so acquiring food in the wild requires animals to make behavioral 'decisions' as to where, when, and for how long to search for food. These strategies will affect the amount of energy predators spend searching for food and obviously the amount of energy they manage to extract from the environment. Energy is the fundamental currency underlying behavioral, physiological, reproductive and developmental processes and is crucial to an animal's growth, survival and reproductive fitness. Research often addresses only one side of the cost/benefit ratio equation—either studying energy input or energy output in isolation. In my study, I looked simultaneously at energy in and energy out during a foraging trip, depending on foraging behaviors or strategies at sea.

### 6.5.1 Foraging efficiency varies with foraging strategy

Out of the 20 females I tracked from Reef Rookery, 40% went to forage in the deep basin and 60% stayed to forage on the Bering Sea shelf. I selected breeding females for my study from Reef rookery because these individuals were known to use a wide range of foraging locations (Robson, 2001). These two strategies have also been observed in female fur seals from Russian and Alaskan breeding islands (Gentry, 1998). However, quantitative estimates of energetic costs associated with the two strategies had not been previously measured.

Females foraging off the shelf traveled greater distances, spent  $\sim 2.5$  more days at sea, and spent  $\sim 10\%$  less time at sea actually diving to capture prey compared to females foraging

on the shelf. However, their dives were overall shallower and of shorter duration. Foraging behaviours of marine mammals indicate prey type, prey distribution and availability in the marine environment (Boyd, 1996; Forcada and Staniland, 2009). The main prey of the off-shelf females were mesopelagic fish such as northern smoothtongue or myctophids, small energy-dense schooling fish that display a diel migration to the surface so they were feeding only at night (Chapter 4). In contrast, females foraging over the shelf dove deeper and for longer times than those that went off-shelf. Female fur seals were capable of diving benthically to the relatively shallow bottom of the Bering Sea shelf (~150 m). On average they spent 15 % of their time diving to the bottom (and up to 30 %) including diving during daytime, and fed mostly on gadids or other shelf-associated prey. Females foraging on the shelf tended to obtain less energy per g of prey ingested as they fed on a diet with an energy density 44% lower than females foraging off the shelf (~3.9 vs ~5.8 kJ/g). However, they fed on prey of bigger size than females foraging off the shelf, which means that the average prey item ingested (~325 kJ) provided them with ~60 % more energy than the average prey consumed by females foraging off the shelf (~195 kJ).

Females foraging in the deep basin attempted to capture 3 times more prey than females on the shelf per day at sea. Both strategies captured ~ 1.76 prey per dive when at least one capture occurred, which is in the range of the ~ 1.88 (Viviant, M. pers. comm.) and ~ 2 prey capture attempts (Chapter 5) measured in Antarctic fur seals. However, the fact that females foraging on the shelf have significantly lower prey capture rates over all the dives (0.46 vs 0.76) indicates that they have a greater rate of unsuccessful dives compared to females foraging off the shelf. Given the greater energy intake per prey ingested for the onshelf group, energy gained per dive (including unsuccessful dives) was similar between strategies (~ 180 kJ/dive). However, as females foraging in the basin performed shallower dives that were shorter than females on the shelf, they could perform more dives within a specific timespan and thus gained almost twice as much energy per minute spent diving (146 vs 82 kJ/min diving). Since females foraging off the shelf spent on average ~ 10 % less time diving that females staying on the shelf, the difference in rate of energy gain was reduced over the entire foraging trip (35 vs 25 kJ/min at sea) but was still greater for females feeding in the deep basin.

Females foraging in the deep basin might have had a greater rate of energy gain, but they also expended more energy during their foraging trip than females staying on the shelf. The increase in energy expenditure while at sea was not due to a greater metabolic rate, but rather to them spending more time and energy transiting to and from their foraging grounds, which were further away from the colony. This compromise on energy expenditure did not completely offset the relative advantage in rate of energy gain that the strategy of feeding in the deep basin offers. Consequently, females foraging off the shelf remained more efficient at foraging (3.02) compared to females on the shelf (1.44).

My estimates of foraging efficiency are much lower than the ratio of 23 reported for northern fur seals in the 1980s (Costa, 1993). However, this high foraging efficiency is based upon food intake that was indirectly derived (using an isotopic method to estimate changes in body water content, Costa 1993) and likely overestimated energy gain by a factor of 3 to 5 (diet of squids). Estimates of prey capture attempts and diet-related parameters carry the most uncertainty to the final calculation of foraging efficiency (Chapter 5). As a consequence, errors in terms of ingestion rates and on type of prey consumed could lead to a grossly overestimated final value such as theirs.

My results show that both on-shelf and off-shelf strategies led to average foraging efficiency above 1, indicating that the lactating females obtained enough energy to support their own physiological needs. However, ~ 30 % of the females foraging on the shelf lost more energy than they gained (ratio significantly lower than 1), and that 45 % seem to break even or were slightly above 1 (Table 6.3). Thus, a majority of females choosing this strategy would have had to rely heavily on internal reserves to support themselves and their pup, which was not the case for females foraging in the deep basin. Lactation and the cost of growing a pup is substantial (Goldsworthy et al., 2004). As foraging efficiency determines the amount of energy available for different physiological functions and as compromises are likely to be made on reproduction before one's survival (Bouten et al., 1994), females gaining 3 times more energy than what they spend are likely to be able to invest more energy in their pup than females that obtain only 1.5 times more.

I am aware that foraging efficiency of a single trip might not be representative of an entire breeding season. However, fur seals are known to remain faithful to their foraging grounds and strategies over successive foraging trips unless a drastic change in environmental assemblage or physical features occurred (Call et al., 2008). This was confirmed by  $\delta^{13}$ C and  $\delta^{15}$ N stable isotopes ratios in plasma and RBC of the same individuals (Chapter 4).

It may seem surprising that females foraging on the shelf should continue foraging here despite the lower energetic efficiency of doing so compared to the females foraging off the shelf. However, presence/absence of prey is a better predictor of foraging habitat than prey density for northern fur seals (Benoit-Bird et al., 2013), and if prey are widely and predictably present on the Bering sea shelf, females might be inclined to keep foraging there. It is highly unlikely that individuals are 'aware' of their rate of energy gain at sea or the energetic values of their prey if they are feeding frequently in predictable patches. Larger size of females is usually an advantage when prey availability or accessibility is low because it provides them with a greater capacity to dive (higher ADL) and to buffer changes in conditions (greater body reserves) (Kooyman, 1989; Iverson et al., 1993). However, I did not find any relation between size of the females (or body condition) and choice in foraging strategy, foraging efficiency, or rate of energy gain. This indicates that prey are likely available and accessible in both foraging environments, but that the limiting factor is the difference in prey energy density.

There is a clear energetic disadvantage to foraging on the Bering Sea shelf compared to the deep basin despite the greater distance to the foraging grounds. Optimal foraging theory (MacArthur and Pianka, 1966) states that methods of searching and obtaining food should maximize foraging efficiency to optimize fitness of the animals. However, the onshelf strategy does not seem to maximise foraging efficiency (or net energy gain) compared to the off-shelf strategy. This raises the question as to whether these foraging differences affect energy flow allocated to reproduction and thus fitness of individuals.

# 6.5.2 Conflict between time and energy currencies

Foraging efficiency of mothers is positively related to pup growth on Antarctic fur seals (Chapter 5), which means that females with greater foraging success produce bigger pups. The fact that pup mass at weaning is directly linked to pup survival in seals (Doidge

and Croxall, 1989; Baker and Fowler, 1992; Boltnev et al., 1998; McMahon et al., 2000; Hall et al., 2001; Beauplet et al., 2005) suggests that females with better foraging strategies that optimize foraging efficiency are likely to improve fitness, consistent with Optimal Foraging Theory. Applying this relationship to northern fur seals suggests that females foraging off the shelf would produce bigger pups and have greater reproductive success than females foraging on the shelf.

Maximising the rate of energy delivery to their pup rather than foraging efficiency (Costa, 1993) is another way for otariids to produce bigger pups (Gentry and Kooyman, 1986). Females foraging on the shelf perform significantly shorter foraging trips compared to females foraging in the deep basin. As foraging trip duration did not affect time spent ashore to feed pups (Antonelis et al., 1990), the females adopting the on-shelf strategy are likely to feed their pups ~ 20 - 25 % more frequently (considering a consistent 2-day difference in trip duration over the breeding season). Under this scenario, on-shelf females might not optimise foraging efficiency but rather the rate of delivery to pups to maximize their reproductive success. Rate of energy gain and time at sea were not significantly related in my study (*i.e.*, females with a greater capacity at obtaining energy while foraging did not perform shorter trips). There is disassociation between the 2 fitness-related currencies that are time and energy invested in foraging versus reproduction (Boggs, 1992; Ydenberg et al., 1992), and individuals compromised either on rate of energy gain (energy currency), or on delivery rate of milk to their pups (time currency). Both lead to a decrease in maternal investment to offspring. Northern fur seals thus seem to be conflicted by the trade-off between time away from their pups and the energy they gain foraging, while Antarctic fur seals seem to optimize both with short average foraging trips and high average FE (Table 6.1, Chapter 5).

Time and energy define the allocation of resources to growth and reproduction. Female fur seals can buffer fluctuations in these currencies by maintaining their body energy reserves when energy is in surplus, and can use them for reproduction investment or other purposes when energy intake exceeds costs (Trillmich, 1990). Sea bird parents have been shown to alternate long foraging trips that replenish their internal reserves with shorter foraging trips that maximise delivery of energy to chicks (Weimerskirch et al., 2003a). This shows the importance of body store usage to adjust and buffer energy expenditure and acquisition during foraging for all allocation purposes. Female northern fur seals with lower body condition before a foraging trip tended to spend more time at sea compared to females in better body condition (Figure 6.2). This suggests that females likely make a trade-off between maintaining their own body condition or that of their fasting pups awaiting nursing, even though some have hypothesized that females can compensate for longer periods at sea by providing greater milk energy content (see section below). This relationship was not observed on Antarctic fur seals, which means that conditions on the Kerguelen plateau were favorable enough that compromises did not have to be made (Chapter 5), unlike in the Bering Sea.

The foraging strategies employed by northern fur seals were surprisingly similar to those of Antarctic fur seals inhabiting opposite hemispheres. Antarctic fur seals had similar time budgets as northern fur seals feeding on the Bering Sea shelf (*i.e.*,time at sea, distance traveled, allocation of time to different activities). However, parameters related to energy (*i.e.*,prey captures, diet and foraging efficiency, etc.) were similar between Antarctic fur seals and northern fur seals foraging off the shelf (Table 6.1). The degree of similarity between the sub-Arctic vs sub-Antarctic species provides insights into the physiological and behavioural flexibility of otariids breeding and foraging in high-latitude environments. On one hand, both face spatial and temporal constraints during the 4-month breeding season. On the other, northern fur seals must currently compromise on one of the limiting conditions (time allocated to nursing vs energy gained) whereas Antarctic fur seals do not. Whether this explains why one population is declining whereas the other is increasing remains unclear, but provides an interesting line for further investigation.

### 6.5.3 Maternal investment

Allocation of energy to the pup is mediated through milk provisioning in fur seals. Factors such as maternal foraging trip duration, diet, mother's mass, time ashore and days postpartum affect milk composition and overall energy transfer to pups (Oftedal, 1984; Trillmich and Lechner, 1986; Arnould and Boyd, 1995b; Arnould and Hindell, 1999; Georges et al., 2001; Goebel, 2002; Banks, 2012). In my case, I measured milk composition in females in mid-lactation, and recaptured females as soon as possible after their return on land post feeding, so both days postpartum and time ashore were standardized as much as possible. Foraging trip duration and maternal diets are thus the 2 main parameters that differed between my 2 groups.

Foraging trip duration has been shown to be positively related to milk lipid and protein content in Antarctic and Australian fur seals (Arnould and Boyd, 1995a; Arnould and Hindell, 1999). In these cases, females that spend more time at sea are thought to compensate for longer fasting of their pups by producing energy richer milk. However, this relationship has not been found in northern or sub-Antarctic fur seals (Georges et al., 2001; Goebel, 2002). Another study on northern fur seal lactation even showed a negative relationship between foraging trip duration and milk lipid content, but attributed it to mammary gland involution when intervals between 2 suckling bouts are too long (Banks, 2012). In my study, I found the relationship between the lipid content of milk (post-foraging) and the duration of the foraging trip was positive. However, my results should be viewed cautiously because I only collected 5 milk samples after the females returned to land. The relationship between foraging duration and milk composition is thus unclear, although it seems that females might be able to at least partly compensate for longer trips by increasing the energy content of their milk. However, it has been suggested that the increase per additional day of absence from the pup to provide an identical milk energy yield would be physiologically unrealistic to achieve (Trillmich, 1990), and that the outcome of increased time at sea (assuming similar diets) would be a lowered growth rate of young (Gentry and Holt, 1986).

Milk composition of fur seals usually reflects what they fed on prior to the nursing bout (Goebel, 2002), which means that greater foraging efficiency should result in higher energy allocation for milk production. This is consistent with lactating females on Bogoslof Island—known to feed on high energy density prey located close to breeding grounds—having significantly greater milk lipid content than females from St. Paul Island (Banks, 2012). Assuming this difference was diet related means that the milk lipid content of my females foraging off the shelf should have been greater than the on shelf females. My results the difference was not significant (~ 50 % vs ~ 46 %), possibly due to my low sample size and reduced statistical power.

Once again, it is difficult to tease apart the impacts that difference in diets and foraging trip durations have on energy allocation to milk, as they seem to be confounding factors. Females foraging off the shelf have longer trips (that tend to increase milk lipid content) and feed on a diet that would logically provide them with a greater energy overhead available for reproduction. However, their pups would be fed less often and likely still grow slowly. On the other hand, if females foraging on the shelf gained only slightly more energy than what they spent foraging, they would feed their pups more often. However, their allocation to milk would likely involve mobilizing body reserves to produce milk of adequate fat content for fur seals (Jensen, 1996), and would thus require greater subsequent foraging efficiency to replenish these reserves. As females come ashore, body reserves are drained through suckling and the metabolic costs of fasting on land until low body reserves induces the female to return to sea to forage (Trillmich, 1990). I did not find any clear differences in body condition between females from the 2 groups, or between body condition and milk composition, but females with the lowest fat content in milk before their foraging trip tended to spend more time at sea during the following trip (n = 5).

Foraging efficiency defines allocation trade-offs of energy to milk or to body reserve and allocation of time to maternal investment or to replenishing body condition, and females might alternate between priorities to maximize both their body integrity and future survival and pup chances of survival. I did not obtain data from individual mother-pup pairs to determine reproductive success but randomly captured pups on the colony as a population index of pup growth and maternal investment of the colony compared to growth rates previously recorded. Pup mass and body condition at weaning are known to define survival of pups during their first year at sea, which links reproductive investment of females to survival of pups, and thus to population demography.

There were no significant differences between pup mass-at-age in the different available datasets (from the 1980s – Boltnev et al. (1998); from 1996 – Donohue et al. (2000); from 2011 – this study; Figure 4). However, analysis of variances on the slopes of these models showed that the growth rates of the pups in 2011 were slower than pups in the 1980s or in 1996 ( $p < 10^{-5}$  in all cases). This difference might reflect a difference in the range of age collection (*i.e.*,I collected data from age 42 to 76 d, while Boltnev et al. (1998)

collected from 0 to 140 d and Donohue et al. (2000) from 15 to 104 d). However, the data I compared my results to were taken during years when the population was not declining (Angliss and Allen, 2009). This might thus highlight a mechanistic influence of lower provisioning from mothers in recent years.

#### 6.5.4 Conclusions

Despite the limitations of my study, my results provide interesting clues into the states of the mother-pup energy transfer on St. Paul Island and highlight the importance of taking a further step towards investigating the series of events that link environmental conditions, animal behaviors, energy balance, fitness of individuals, and demographic parameters. It is interesting to note that females from an increasing population (Antarctic fur seals) can on average simultaneously maximise foraging efficiency and frequency of feeding their pups, while females from a declining population (northern fur seals) are faced with a trade-off between the two. Having a significant proportion of breeding mothers adopt a strategy that does not maximise their offspring's chances of surviving might significantly affect population recruitment. Ultimately, this could lead to northern fur seals producing a significant portion of smaller pups at weaning, with lower survival during their first year and thus a lower recruitment in the population (Baker and Fowler, 1992). If my study provides evidence of such a possibility, there is a need to investigate whether one foraging strategy yields a lower reproductive success via lower pup survival of the young compared to the other to fully understand whether this is a potential cause for the population decline. The next step would be to estimate the survival rate of juveniles over the first few years of their pelagic life at sea, and how it relates to the foraging strategies of their mothers during their first 4 months of life.

# **Chapter 7: General conclusion**

The main objective of my thesis was to determine how foraging strategies of lactating northern and Antarctic fur seal females affected their foraging efficiencies and ultimately their fitness. In this way, I sought to link environmental and prey conditions with the population dynamics of the two species. I used a quantitative approach based on energetic flows that standardized for the differences between foraging habitats. I also used state of the art technology to investigate the foraging locations and behaviours for single foraging trips of 20 mature lactating fur seals from both species, and determined how much energy they spent and how much energy they extracted from the sea. I then linked foraging strategies and efficiencies of individual mothers with the growth of their respective pups for Antarctic fur seals, and with milk samples and the mass of random pups on the colony for northern fur seals to investigate how these parameters affected reproductive success. Trade-offs in energy and time during the nursing season affect the fitness of females and ultimately the survival of their pups during their first year at sea. Ultimately, changes in survival of the young and thus of recruitment can alter the dynamics and trajectories of populations.

Studying large and elusive mammals in the marine environment is inherently difficult and resulted in some unavoidable caveats. However, my individual energetic-based approach revealed robust and quantitative connections between the environment and the population dynamics of the two species. The results have advanced understanding of how animals cope with their environment and their adaptive limits to changes or sub-optimal or changing conditions. My study emphasizes the importance of considering individual variability and individual-based studies given that individual performance and plasticity link the environment with population trends. It also shows the importance of basing studies on robust science, such as energetic-flows and thermodynamic laws, to draw mechanistic, processdriven relationships and understand and predict population changes. Ultimately, my study highlights the potential vulnerabilities of fur seal populations to current or future changes in their environments and provides information to direct spatial and temporal conservation efforts towards the most sensitive part of the population. However, my study left a number of questions unanswered and revealed new ones that need further investigation.

### 7.1 Summary of findings

Chapters 2 and 3 sought to develop, test and validate methods to predict foraging energy costs using acceleration data. In Chapter 2, I investigated foraging energy expenditure of northern and Antarctic fur seal females given their time-activity budgets at sea. To do so, I used field measurements of metabolic rates at sea from the doubly- labeled water method as a reference for energy expenditure, which, when linked to fine-scale time-activity budgets of individuals, allowed me to calculate activity-specific metabolic rates. I tested whether triaxial dynamic body acceleration was accurate at predicting energy spent at sea in freeranging animals over a full foraging trip or separated by types of activities they displayed at sea. My results show that time-activity budgets affected energy expenditure at sea of individuals. They also show that diving is the most expensive activity followed by transiting at the surface. Dynamic body acceleration (VeDBA) also needs to be separated by type of activity to accurately predict energy expenditure at sea.

I feel that Chapter 2 represents my most significant contribution to the field of marine mammal ecology as I showed how combining behavioural information from GPS, time-depth recorders and high-resolution accelerometers with metabolic measures of energy expenditure can be used to determine and partition fine-scale behaviours and their associated metabolic rates in a free-ranging animal. This approach opens new perspectives to the field of freeranging energetics to gain a greater understanding concerning how changes in foraging strategies, via time-activity budgets, can impact foraging costs. The doubly-labeled water method indirectly measure respirometry exchanges and is consequently based on more robust physiological and energetic grounds than acceleration. However, it is very expensive to perform on large animals, only provides a single measurement over the duration of the foraging trip, and is biased at the individual level. Accelerometers are a cheaper and easier tool to employ in field studies, and their strength lies in their potential to provide energetic information at the activity level. However, acceleration is a measure of body movement and not of gas exchange, and limitations to its use also remain, mostly at the full trip level. Future research should thus take into account the pros and cons of each technique depending on the focus of the study to decide on a methodology.

In Chapter 3, I investigated similar questions and employed a similar approach as in Chapter 2, but focused on the cost of transport rather than tri-axial dynamic body acceleration. I developed a method to detect flipper strokes from acceleration data and tested whether their number, rate, or amplitude could accurately predict energy expenditure at sea. I also calculated energetic cost of stroking flippers for fur seals, and the cost to travel a given distance under water. My results show that flipper strokes could only be detected when animals were diving, and when they were transiting at a fast rate. Flipper strokes were accurate at predicting energy expenditure during diving and to a lesser extent during transiting, but not as well during an entire foraging trip. My results also show that tri-axial dynamic body acceleration is a better predictor of energy spent while foraging when separated by activities than are flipper stroke metrics.

In Chapter 4, I focused on determining accurate diet composition of northern fur seals, an essential component of energy gained from foraging. I employed modern molecular methods to improve detection of prey from scats compared to more traditional hard-part based methods. My results show that despite inherent biases from both methods, DNA meta-barcoding provides a finer more complete list of prey. Concurrent measures of C and N blood stable isotopes on 20 individual females from the colony where the scats were collected, indicated that northern fur seals foraged in two different locations (on shelf versus deep basin) and had two different diets. Presence of these two diets was confirmed by a cluster analyses of prey associated within scats, but were not represented in the same proportion as was indicated by stable isotopes. This work represents the first step towards determining the energy gained by females as a function of the prey they consumed.

Chapters 5 and 6 integrated results from Chapters 2 and 4 to investigate differences in foraging strategies and their respective foraging efficiency, and discussed their impacts on reproduction success. I estimated foraging strategies (from foraging locations and time-activity budgets at sea), foraging costs (from the DLW measure of energy expenditure) and foraging benefits (from diet composition), size of prey (from scat hard-part remains), and number of prey capture attempts (from acceleration data). Chapter 5 focused on Antarctic fur seals and linked foraging parameters of mothers to the growth of their respective pups from birth to weaning. The results showed that the quality of mothers as foragers was directly

linked to the size of their pups at weaning, and thus of the probability of their pups surviving during their first year at sea. Better foragers were capable of increasing their rate of energy gain without increasing their overall energy expenditure, and spent overall less time at sea compared to less efficient females. Thus, increased fitness through higher maternal investment was achieved by maximising both foraging efficiency and their frequency of feeding their pup.

Chapter 6 focused on the northern fur seal population, and showed that the two foraging strategies described in Chapter 4 resulted in different foraging efficiencies. Both groups of females spent similar energy to forage, but females foraging in the on-shelf neritic habitat had lower foraging efficiencies as they captured bigger prey of lesser energetic quality and did so less often compared to females foraging in the off-shelf pelagic area. They however spent less time at sea overall, which meant that they returned to land to feed their pups more frequently. So unlike Antarctic fur seals, female northern fur seals compromised either their foraging efficiency or the frequency with which they fed their pups. My results indicate that females foraging for longer periods of time might compensate for greater fasting duration of their pups by increasing milk energy content, but the extent of this compensation is likely limited. There were also limited clues that the growth rate of NFS pups of the colony might be lower than in the pre-decline time. In conclusion, compromising on either foraging efficiency or on time spent nursing their pup might result in lower pup growth rates on the colony and ultimately declines in population size, even if the direct link between foraging strategies and individual pup growth rates could not be made.

# 7.2 Evaluation of research hypotheses

My general research hypothesis was that females from the declining population used foraging strategies that increased energy expended during foraging and minimise their foraging success due to prey being less accessible and of poorer quality, and therefore have lower foraging efficiencies compared to females from the increasing population. They therefore have less energy and time to allocate to reproduction, and produce pups that grow at slower rates. As mass at weaning is critical for survival during their first year at sea, this might lead to lower recruitment in the declining population. My results did not show clear inter-specific differences in foraging strategies. As species, northern and Antarctic fur seals were similar in their average trip durations, distances traveled and time-activity budgets. They allocated the same proportion of time to diving, transiting and resting during their foraging trips at sea—even though Antarctic fur seals tended to perform more dives, but of shorter duration than northern fur seals. However, I found a significant intra-specific dichotomy in foraging strategies between northern fur seals. Forty percent of the females foraged in a pelagic habitat within the deep basin of the Bering Sea, while 60% remained in the neritic habitat of the shallow shelf. Females foraging on the shelf had similar trip durations and distances traveled as Antarctic fur seals, but their diving behaviours were different. Alternatively, female northern fur seals foraging in the deep basin displayed diving behaviours very similar to those of Antarctic fur seals, but performed trips of longer duration and traveled longer distances.

Contrary to my initial hypothesis, female northern fur seals also did not spend more energy to forage compared to Antarctic fur seals. Field metabolic rates of the northern fur seals foraging on the shelf were greater than those of the females foraging off the shelf. They were also greater than the metabolic rates of the Antarctic fur seals. However, spending less time at sea resulted in similar total energy expenditure between and within species. As expected, however, my results showed that northern fur seals as a population have a lower rate of energy gain and foraged less efficiently than Antarctic fur seals. Females foraging on the Bering Sea shelf captured prey of lower energy density and less frequently than Antarctic fur seals, which significantly reduced foraging efficiency. However, female northern fur seals whose strategy was to forage in the deep basin showed similar numbers of prey capture attempts, rates of energy gain and foraging efficiency of northern fur seals was not due to an increase in foraging cost, but reflected a decrease in the foraging success.

My hypothesis that females with lower foraging efficiency would have less energy to invest into pup rearing was validated on Antarctic fur seals. Females with a greater foraging efficiency produced bigger pups at weaning, which would thus be expected to have a higher survival probability during their first year at sea. I was however unable to make the direct link and comparison between fur seal species as I could not follow pup growth on the northern fur seal colony. My results showed that females from the increasing population were capable of high foraging efficiency and high rate of milk delivery to their pups, while females from the declining population have had to compromise either on foraging efficiency or on the frequency with which they fed their pups. Females unable to optimise both of these parameters might not produce pups big enough to survive their first year at sea, which in turn could reduce recruitment and negatively affect the population trend.

Overall, my initial hypothesis was partly validated. The reason why northern fur seals have a lower foraging efficiency than Antarctic fur seals was different than I initially expected. However, there are indications that lower foraging efficiency might indeed be a limiting factor linked to their decline. The consistency in foraging costs between and within species might suggest that females from both species forage at their metabolic ceiling and that they have limited plasticity to modulate it. Fur seals are flexible in their foraging behaviours and choices of strategies at sea, but if environmental conditions change such that they do not allow a rate of energy gain sufficient to match their high and seemingly hardwired energy costs of foraging, they might be at risk of being in a negative balance and reducing their fitness.

One of my research objectives was to also test and validate new methods to determine energy expenditure at sea, and whether this could be achieved at the activity-specific level in free-ranging fur seals. My hypothesis was that accuracy of acceleration would be limited in free-ranging setting when animals spend time at the interface between two media (*i.e.*, when they are at the surface). My results showed partial validation of my hypothesis as acceleration metrics (VeDBA or flipper stroke) were poor predictors of energy expenditure over a full foraging trip. However, when separated by type of activities performed at sea, both VeDBA and to a lesser extent flipper stroke metrics accurately predicted energy spent (*i.e.*, activity-specific metabolic rates calculated from best fit models). Consequently, acceleration can be used in fur seals to predict energy expenditure at sea provided time activity budget of individuals are taken into account.

### 7.3 Caveats and limitations

My study was integrative and exhaustive, but the constraints of data collection on large remote animals and the overall protocol came with some caveats that need to be acknowledged. One of the major caveats around foraging efficiency is that all my analyses were individual-based, and yet diet compositions of individuals were estimated from population estimates. I also assumed that females foraging off the shelf had similar diet as females breeding on Bogoslof island based on stable isotopes and foraging habitat, even though they likely differ (albeit probably slightly). Stable isotopes only provide an overall trophic level of the predator, even though careful mixing models can provide greater taxonomic resolution. Different prey species can have similar isotopic signatures and yet different sizes and energy densities, which would likely change my final calculations. In addition, the Bogoslof diet I used was collected 2 years before my field season with potentially different yearly conditions. Estimating diets of wild animals is inherently difficult and cannot at this point be estimated per individual animal. I however used the best and latest methods in combination to increase the precision of the final outcome.

My selection of females for the study was likely biased towards more successful individuals. For northern fur seals, I only selected healthy-looking females with a confirmed pup at mid-lactation (*i.e.*, individuals already successful at rearing a viable pup for 2 months). The 20 Antarctic fur seal females were selected as part of a pool of 36 females whose pups were tagged at birth among the first born on the colony. More experienced—and thus more successful—females tend to pup earlier in the season (Lunn and Boyd, 1993), so my data is likely skewed towards the best individuals. In addition, as I needed to glue 3 different data loggers on the back and head of each animals, I tended to select females towards the heaviest range (although not exclusively) to minimise the impact of drag on the animals. Indeed, presence of these tags on animals affects streamlining and buoyancy and increases drag during swimming (Wilson et al., 1986; Boyd et al., 1997). Fur seals also rely exclusively on their dense fur to thermoregulate. Gluing tags on their fur disrupts the boundary layer and affects their thermal integrity. They can compensate behaviourally for the added drag (Boyd et al 1997), but ultimately, presence of data loggers increases their energy expenditure at sea (Wilson et al 1986) and will consequently lower foraging

efficiency estimates more so for smaller females, which means the error might not be consistent between all individuals.

Financial and logistic constraints allowed me to only follow females for a single foraging trip. My results thus provide only a narrow window of information into the foraging ecology of these animals. Even though there is evidence that females are consistent within a year in terms of their foraging behaviours (Call et al., 2008) and foraging efficiencies (my study), extending my conclusions beyond the timescale of my measurements should be taken cautiously. Individuals might adjust their foraging behaviours to find better conditions in subsequent trips if conditions are suboptimal during the preceding one. Better quality Antarctic fur seals seem consistently more efficient over the breeding season in their uniform foraging habitat, but it might not be the case for northern fur seals that can alternate between two different habitats and diets. It is uncertain whether females with low foraging efficiencies during the measured trip would still be in low balance in following trips-or whether females with short foraging trips will consistently make short trips. Flexibility or adaptability of females to environmental conditions likely operates at the timescale of a breeding season rather than a single foraging trip. Therefore, my short time window only allows educated inferences and conjectures regarding the consequences of foraging efficiency on fitness of individuals.

My inferences were stronger for Antarctic fur seals as I could follow growth of pups associated with the tracked females over a full nursing period. Unfortunately, this was not feasible on the northern fur seal rookery. The biggest caveat and missing piece of this puzzle was that I could not estimate from my data whether female northern fur seals with greater foraging efficiencies or females that feed their pup more frequently produced bigger pups at weaning, whether both produced small pups given their compromise on either time or energy allocated to maternal investment, or even whether this compromise affected pup growth at all. As these two parameters were confounded in Antarctic fur seals, it was difficult to estimate their respective contribution to the overall reproduction success. In addition, determining whether compromising on time or energy allocated to reproduction impacts population dynamics of northern fur seals would require comparing present pup information with pre-decline data. I could only collect morphometric measures of random pups 3 times at mid-breeding season, so the comparison was very limited—especially since I could not obtain masses at weaning. Consequently, my data highlight possible reasons for why the northern fur seal population has declined based on comparisons with Antarctic fur seals, but this would need to be explicitly uncovered for northern fur seals.

Interspecific comparisons such as I undertook is a common approach in physiological ecology to elucidate the processes that animals use to adapt to environmental conditions. Northern and Antarctic fur seals undoubtedly share many similarities despite their opposite ranges. However, comparisons between only two species that involve correlations between individual characteristics and the environment raise several logical and statistical problems. Garland and Adolph (1994) argue that concluding significant differences between two species would not be statistically meaningful as they are expected due to genetic divergence resulting from differential environmental selective pressures. The null hypothesis should thus be that species are different, and not similar. So inferring mechanistic processes in northern fur seals based on results from Antarctic fur seals should be taken cautiously. Garland and Adolph (1994) thus advise making comparisons between more than two species and between populations from the same species. In this case, differences uncovered are less a result of genetics and are more readily related to the environments in which the populations evolved.

My results are based on only two species, but they show that northern and Antarctic fur seals are strikingly similar in numerous behavioural and biological and ecological features despite having phylogenetically diverged 5 - 6 million years ago (Hoelzel, 2002). These similarities are thus more meaningful in terms of adaptation and evolution than any differences I could have found, and indicate that selective pressures might have been convergent despite their opposite hemispheric habitats. Physical, chemical and biological features are likely more important to define similarity in foraging habitats and thus behavioural responses of animals living in them rather than plain geographical location or coordinates. Both the Bering Sea and the Kerguelen plateau present very similar features, which likely explain the similarities between species (or dissimilarities between groups of the same species foraging in different environments). I also studied two populations within the northern fur seal species and two species overall (*i.e.*, three groups in total as advised by Garland and Adolph 1994), which increases the biological significance of the differences and

similarities I uncovered. Within species differences are usually expected to be less extreme than between species, but this was not the case in my study. This provides more grounds to my inter- and intra-specific approach, and to my energetic flows approach that synthesizes and standardizes different environments to a common thermodynamic framework.

I also assumed based on previous studies of different pinniped species that growth rates and mass at weaning of pups were linked to 1) their survival during the most critical period of their life and 2) to the fitness of females. If there is evidence of such links, they are seldom explicit. Consequently, conclusions in my study are based on strong assumptions, but not totally explicit relationships. Fitness is assessed over a lifetime and is a consequence of compromises between the individuals' own long term survival and their reproductive rate. Females that invest more into reproduction in a given year might jeopardize their future survival and thus opportunities to reproduce over a longer lifetime, especially if foraging conditions are challenging (Beauplet et al., 2006). Whether foraging efficiencies of mothers ultimately affect survival of pups during their first critical year needs to be further investigated. Larger body reserves at weaning likely provide a survival advantage for naïve and inexperienced young. However, it is vital to untangle whether lack of maternal investment or changes in environmental conditions that pups face during their first year at sea are responsible for a lack of recruitment in the declining population to conclude more assertively on the matter.

Finally, I used the increasing Antarctic fur seal population as a reference to understand what mechanisms could be involved in the decline of northern fur seals. However, it is important to keep in mind that population dynamics are a result of more than just birth and death rates as partially studied in my thesis. Predation, food supply, breeding site, diseases, inter- and intra-species competition for resources, environment variability and human activities all impact the abundance of individuals. Antarctic fur seals were hunted to near extinction, and it is possible that the population has not yet reached carrying capacity, and that the observed trend reflects this post-exploitation recovery. Post-exploitation recovery of Antarctic fur seals at other sites has shown population growth rates as high as 20% per year, slowing down to 5–6% close to a new equilibrium (Hucke-Gaete et al., 2004). Human activities and predation pressure are also likely heavier in the Bering Sea compared

to the Southern Ocean for example. Consequently, even if my study hypothesized that differences in population trends between northern and Antarctic fur seals are a result of the same mechanisms, it should be acknowledged that other—and probably different—factors are likely involved.

# 7.4 Future research

Diet composition is one of the most important parameters in ecological studies. Individual-based models would greatly benefit from focusing on developing new methods and technologies to determine the diets of individual animals rather than of populations. Such direct observations of prey ingestion would require technological advancements and miniaturisation of underwater cameras, which are currently limited in battery power, image resolution in dark environments, and physical memory space. It would be interesting to pair accelerometers with the cameras to trigger a still shot of prey when a PCA is initiated. On board processing of pictures could also be developed for tags that cannot be recovered or when memory space is limited. There are still some technical hurdles, but advancements in this field would greatly enhance understanding of predator-prey interactions, of individual energetics, and ecosystem dynamics.

Despite the advances in the understanding the mechanisms and processes involved in the decline of northern fur seals, and given the limitations mentioned in the previous section, future research should focus on more explicitly linking foraging strategies and efficiencies of mothers to respective growth and survival at sea of northern fur seal pups. It should expand from the work done here to uncover whether the foraging strategies chosen by females leads to different pup masses at weaning or not, and if so, which strategy produces bigger pups at the colony level. To do so, future research should build on the fact that foraging strategies of females are associated with different C and N stable isotopic signatures (Chapter 4), which will thus be transferred to their pups through nursing. Collecting mass and blood samples for isotopic signatures of pups close to weaning on a large scale would indicate which strategy their mother adopted during the breeding season—and whether she was consistent in her choice or not—and whether mothers adopting one strategy over another provide an advantage to their offspring before leaving the breeding site. Finally, survival of these known pups during their first years at sea should be assessed through mark (before they have weaned) and recapture techniques (when they return on land for the first time). This would fill in the gap that my study left in the relationship between foraging strategies, efficiencies and fitness of the animals—and would provide a better mechanistic framework to understand why the population is declining.

Following the advice of Garland and Adolph (1994), future individual-based and quantitative work to link foraging efficiency and fitness should involve comparisons on a larger scale. Comparisons should also be undertaken using more populations within the northern and Antarctic fur seal species for more robust statistical significance. Pups from islands located on the Bering Sea shelf should be compared to pups from islands within or close to the deep basin such as Bogoslof Island or the Commander islands, or to San Miguel Island in California, which are sites of increasing populations. Alternatively, Antarctic fur seal colonies are present all around the Antarctic Peninsula with different environmental conditions and diets. Other species of mammals and seabirds are also subject to the same environmental changes and are faced with similar constraints in time and space during their reproductive cycles. Increasing the number of inter-specific comparisons with species such as penguins breeding at the same site in the southern hemisphere or alcids in the northern hemisphere to study individual strategies, efficiencies and adaptations to their environments could help to understand which species are less flexible and more susceptible to environmental changes. Such interspecies comparisons would also shed light on what coping mechanisms exist and whether general adaptation and evolutionary patterns can be uncovered.

Future research should also expand the physiological and energetic framework to investigate causes of individual variations in addition to their consequences. The personalities of individual animals likely affect their foraging strategies and decisions, capacities to forage efficiently and adaptability to changes in environmental conditions. Studies on albatrosses showed that bolder personalities for example are more prone to explore and discover new foraging grounds (Patrick et al., 2013). It would also be beneficial to understand how and why alternative strategies are maintained in the population, as well as the heritability of such traits. Linking individual personalities to foraging over one or more

breeding seasons would help to understand how choices are made, whether foraging strategies evolve within or between breeding seasons and what the environmental or physiological clues or thresholds are that trigger the changes in strategies.

To conclude, my study improved current understanding of the relationship between the foraging choices of two top marine predators, their fitness, and their population dynamics in relation to environmental conditions. It addressed the issue from an integrative, robust and comparative approach. In the context of global environmental changes and population collapses such as for northern fur seals and many others, the ecological energetics framework I used addressed real-world questions at spatial and temporal scales relevant to environmental issues. It allowed the integration of physiological requirements within ecological constraints and the interpretation of organismal energy dynamics within the broader template of ecology and environmental management. In an era of unprecedented environmental challenges, my results provide a mechanistic understanding of the drivers of threatened species. They allow process-based approaches to predict and measure the energetic responses of organisms to varying environmental conditions. Ultimately, my study provides a means to translate empirical behavioural or energetic measurements into practical, real-world outcomes on which to base management decisions for wildlife conservation.

The declining population of northern fur seals has experienced significant environmental changes around their breeding grounds. These changes appear to affect individual fitness and population dynamics through the relationship I quantified between mother's foraging efficiency and pup mass at weaning, but also most likely through lower survival of naïve young of the year that have limited diving capacities. The presence of the two northern fur seal foraging strategies might indicate that they have been maintained in the population because they have similar net fitness outcomes—which would be an evolutionary stable strategy (ESS). However, neither strategy currently appears to be efficient in terms of time or energy spent and delivered. This might reflect differences in environmental conditions now compared to when the two foraging strategies evolved. Assessing the respective contributions of the two foraging strategies to the decline of northern fur seals is difficult. Either one of the foraging strategies could contribute more to the decline of the population than the other—or the difference between them might simply be averaged in terms of fitness over several generations.

In contrast, the population of Antarctic fur seals breeding on Kerguelen is currently healthy, although increases are occurring in the frequency and intensity of anomalous conditions in the Southern Ocean caused by oceanic warming that might trigger a southward shift of the Polar Front (Moore et al., 1999). Such a change would mean that dense and predictable prey patches will be located further from the colony and likely at deeper depths. Less accessible and predictable prey implies greater energetic costs to access prey for lower energy benefits of foraging per unit of time, which can impact foraging efficiency and reproductive success (Inchausti et al., 2003).

Climatic anomalies affecting food availability (Loeb et al., 1997; Waluda et al., 2004) create more challenges for lactating fur seals (Lescroël et al., 2010); Lea et al. (2006) that might not be able to absorb the extra foraging effort required without compromising either reproduction or survival. Climatic anomalies have been hypothesized to be responsible for the decrease in reproduction success in seabirds (Weimerskirch et al., 2003b) and may have more pronounced effects on less adaptable central place foragers already operating close to their metabolic ceiling such as fur seals. Consequently, my study not only provides a greater understanding of ongoing processes to potentially explain population trends of fur seals today, but also a mechanistic framework to build predictive scenarios of the effects of environmental changes on future population dynamics.
## References

- Adachi, T., Maresh, J.L., Robinson, P.W., Peterson, S.H., Costa, D.P., Naito, Y., Watanabe, Y.Y., Takahashi, A., 2014. The foraging benefits of being fat in a highly migratory marine mammal. P. Roy. Soc. Lond. B. 281: 2014-2120.
- Altschul, S.F., Gish, W., Miller, W., Myers, E.W., Lipman, D.J., 1990. Basic local alignment search tool. J. Mol. Biol. 215, 403-410.
- Anderson, P.J., Blackburn, J.E., Johnson, B.A., 1997. Declines of forage species in the Gulf of Alaska, 1972-1995, as an indicator of regime shift., Forage fishes in marine ecosystems., Alaska Sea Grant, University of Alaska Fairbanks, Fairbanks, AK, pp. 531-543.
- Anderson, P.J., Piatt, J.F., 1999. Community reorganization in the Gulf of Alaska following ocean climate regime shift. Mar. Ecol. Prog. Ser. 189, 117-123.
- Andrews, A.G., Farley, E.V., Moss, J., J.H., Murphy, J.M., Husoe, E.F., 2009. Energy density and length of juvenile pink salmon *Oncorhynchus gorbuscha* in the eastern Bering Sea from 2004 to 2007: a period of relatively warm and cool sea surface temperatures, In: North Pacific Anadromous Fish Commission (Ed.), pp. 183-189.
- Angliss, R.P., Allen, B.M., 2009. Alaska marine mammal stock assessments, 2008, U.S. Department of Commerce, NOAA Technical Memorandum, NMFS-AFSC-193, p. 259.
- Anthony, J.A., Roby, D.D., Turco, K.R., 2000. Lipid content and energy density of forage fishes from the northern Gulf of Alaska. J. Exp. Mar. Biol. Ecol. 248, 53-78.
- Antonelis, G.A., Sinclair, E.H., Ream, R.R., Robson, B.W., 1997. Inter-island variability in the diet of female northern fur seals (*Callorhinus ursinus*) in the Bering Sea. J. Zool. 242, 435-451.
- Antonelis, G.A., Stewart, B.S., Perryman, W.F., 1990. Foraging characteristics of female northern fur seals (*Callorhinus ursinus*) and California sea lions (*Zalophus californianus*). Can. J. Zool. 68, 150-158.

- Aoki, K., Watanabe, Y.Y., Crocker, D.I.E., Robinson, P.W., Biuw, M., Costa, D.P., Miyazaki, N., Fedak, M.A., Miller, P.J.O., 2011. Northern elephant seals adjust gliding and stroking patterns with changes in buoyancy: validation of at-sea metrics of body density. J. Exp. Biol. 214, 2973-2987.
- Arim, M., Naya, D.E., 2003. Pinniped diets inferred from scats: analysis of biases in prey occurrence. Can. J. Zool. 81, 67-73.
- Arnould, J.P.Y., 1995. Indices of body condition and body composition in female Antarctic fur seals (*Arctocephalus gazella*). Mar. Mamm. Sci. 11, 301-313.
- Arnould, J.P.Y., 1997. Lactation and the cost of pup-rearing in Antarctic fur seals. Mar. Mamm. Sci. 13, 516-526.
- Arnould, J.P.Y., 2002. Southern fur seals, in: Perrin, W.F. (Ed.), Encyclopedia of marine mammals, Academic Press, San Diego, pp. 1146-1151.
- Arnould, J.P.Y., Boyd, I.L., 1995a. Inter- and intra-annual variation in milk composition in Antartic fur seals (*Arctocephalus gazella*). Physiol. Zool. 68, 1164-1180.
- Arnould, J.P.Y., Boyd, I.L., 1995b. Temporal patterns of milk production in Antarctic fur seals (*Arctocephalus gazella*). J. Zool. 237, 1-12.
- Arnould, J.P.Y., Boyd, I.L., Speakman, J.R., 1996. The relationship between foraging behaviour and energy expenditure in Antarctic fur seals. J. Zool. 239, 769-782.
- Arnould, J.P.Y., Hindell, M.A., 1999. The composition of Australian fur seal (*Arctocephalus pusillus doriferus*) milk throughout lactation. Physiol. Biochem. Zool. 72, 605-612.
- Aurioles, D., Trillmich, F., 2008. Arctocephalus gazella, In: IUCN Red List of Threatened Species. http://www.iucnredlist.org/details/3590/0
- Bailleul, F., Luque, S., Dubroca, L., Arnould, J.P.Y., Guinet, C., 2005. Differences in foraging strategy and maternal behaviour between two sympatric fur seal species at the Crozet Islands. Mar. Ecol. Prog. Ser. 293, 273-282.
- Baker, J.D., Fowler, C.W., 1992. Pup weight and survival of northern fur seals, *Callorhinus ursinus*. J. Zool. 227, 231-238.

- Banks, A.S., 2012. Variability in population trends, life history characteristics, and milk composition of northern fur seals in Alaska, MSc. Thesis, University of Alaska Fairbanks, Ann Arbor, 112pp.
- Barbraud, C., Weimerskirch, H., 2001. Emperor penguins and climate change, Nature, pp. 183-186.
- Barger, C.P., Kitaysky, A.S., 2012. Isotopic segregation between sympatric seabird species increases with nutritional stress. Biol. Lett. 8, 442-445
- Beauplet, G., Barbraud, C., Chambellant, M., Guinet, C., 2005. Interannual variation in the post-weaning survival of subantarctic fur seals: influence of pup sex, growth rate, and environmental conditions. J. Anim. Ecol. 74, 1160-1172.
- Beauplet, G., Barbraud, C., Dabin, W., Kussener, C., Guinet, C., 2006. Age-specific survival and reproductive performances in fur seals: evidence of senescence and individual quality. Oikos 112, 430-441.
- Benoit-Bird, K.J., Battaile, B.C., Heppell, S.A., Hoover, B., Irons, D., Jones, N., Kuletz, K.J., Nordstrom, C.A., Paredes, R., Suryan, R.M., Waluk, C.M., Trites, A.W., 2013. Prey patch patterns predict habitat use by top marine predators with diverse foraging strategies. PLoS ONE 8, e53348.
- Bidder, O.R., Campbell, H.A., Gomez-Laich, A., Urge, P., Walker, J., Cai, Y., Gao, L., Quintana, F., Wilson, R.P., 2014. Love thy neighbour: automatic animal behavioural classification of acceleration data using the K-nearest neighbour algorithm. PLoS ONE 9, e88609.
- Bigg, M.A., 1981. Digestion rates of herring (*Clupea harengus pallasi*) and squid (*Loligo opalescens*) in northern fur seals, 24th Annual meeting of the standing scientific committee, North Pacific Fur Seal Commission, Tokyo, Japan, pp. 1-12.
- Bigg, M.A., Fawcett, I., 1985. Two biases in diet determination of northern fur seals (*Callorhinus ursinus*), in: Beddington, J.R., Beverton, R.J.H., Lavigne, D.M. (Eds.), Marine mammals and fisheries, George Allen and Unwin Ltd., Massachusetts, pp. 284-291.

- Bishop, C.M., Spivey, R.J., Hawkes, L.A., Batbayar, N., Chua, B., Frappell, P.B., Milsom,
  W.K., Natsagdorj, T., Newman, S.H., Scott, G.R., Takekawa, J.Y., Wikelski, M.,
  Butler, P.J., 2015. The roller coaster flight strategy of bar-headed geese conserves
  energy during Himalayan migrations. Science 347, 250-254.
- Boggs, C.L., 1992. Resource allocation: exploring connections between foraging and life history. Funct. Ecol. 6, 508-518.
- Boltnev, A.I., York, A.E., Antonelis, G.A., 1998. Northern fur seal young: interrelationships among birth size, growth, and survival. Can. J. Zool. 76, 843-854.
- Booth, D.T., 2009. Swimming for your life: locomotor effort and oxygen consumption during the green turtle (*Chelonia mydas*) hatchling frenzy. J. Exp. Biol. 212, 50-55.
- Bouten, C.V., Westerterp, K.R., Verduin, M., Janssen, J.D., 1994. Assessment of energy expenditure for physical activity using a triaxial accelerometer. Med. Sci. Sport. Exer. 26, 1516-1523.
- Bowen, W.D., 2000. Reconstruction of pinniped diets: accounting for complete digestion of otoliths and cephalopod beaks. Can. J. Fish. Aquat. Sci. 57, 898-905.
- Boyd, I.L., 1996. Temporal scales of foraging in a marine predator. Ecology 77, 426-434.
- Boyd, I.L., 1999. Foraging and provisioning in Antarctic fur seals: interannual variability in time-energy budgets. Behav. Ecol. 10, 198-208.
- Boyd, I.L., 2000. State-dependent fertility in pinnipeds: contrasting capital and income breeders. Funct. Ecol. 14, 623.
- Boyd, I.L., 2002. Energetics: consequences for fitness, in: Hoelzel, A.R. (Ed.), Marine mammal biology: an evolutionary approach, Blackwell Sciences Publishing, Oxford, Malden, pp. 247-277.
- Boyd, I.L., Arnbom, T., 1991. Diving behavior in relation to water temperature in the southern elephant seal: foraging implications. Polar Biol. 11, 259-266.
- Boyd, I.L., Croxall, J.P., Lunn, N.J., Reid., K., 1995a. Population demography of Antarctic fur seals: the costs of reproduction and implications for life-histories. J. Anim. Ecol. 64, 505-518.

- Boyd, I.L., Lunn, N.J., Barton, T., 1991. Time budgets and foraging characteristics of lactating Antarctic fur seals. J. Anim. Ecol. 60, 577-592.
- Boyd, I.L., Mccafferty, D.J., Walker, T.R., 1997. Variation in foraging effort by lactating Antarctic fur seals: response to simulated increased foraging costs. Behav. Ecol. Sociobiol. 40, 135-144.
- Boyd, I.L., Wanless, S., Camphuysen, C.J., 2006. Top predators in marine ecosystems : their role in monitoring and management. Cambridge University Press, Cambridge, UK ; New York.
- Boyd, I.L., Woakes, A.J., Butler, P.J., Davis, R.W., Williams, T.M., 1995b. Validation of heart rate and doubly labelled water as measures of metabolic rate during swimming in California sea lions. Funct. Ecol. 9, 151-160.
- Burnham, K.P., Anderson, D.R., 2002. Model selection and multimodel inference : a practical information-theoretic approach. 2d Ed. ed. Springer, New York.
- Burton, R.K., Koch, P.L., 1999. Isotopic tracking of foraging and long-distance migration in northeastern Pacific pinnipeds. Oecol. 119, 578-585.
- Butler, B.J., 1993. To what extent can heart rate be used as an indicator of metabolic rate in free-living marine mammals. Symp. Zool. Soc. 66, 317-332.
- Butler, P.J., Bevan, R.M., Woakes, A.J., Croxall, J.P., Boyd, I.L., 1995. The use of data loggers to determine the energetics and physiology of aquatic birds and mammals. Braz. J. Med. Biol. Res. 28, 1307-1317.
- Butler, P.J., Green, J.A., Boyd, I.L., Speakman, J.R., 2004. Measuring metabolic rate in the field: the pros and cons of the doubly labelled water and heart rate methods. Funct. Ecol. 18, 168-183.
- Butler, P.J., Woakes, A.J., Boyd, I.L., Kanatous, S., 1992. Relationship between heart rate and oxygen consumption during steady-state swimming in California sea lions. J. Exp. Biol. 170, 35-42.

- Call, K.A., Ream, R.R., Johnson, D., Sterling, J.T., Towell, R.G., 2008. Foraging route tactics and site fidelity of adult female northern fur seal (*Callorhinus ursinus*) around the Pribilof Islands. Deep-Sea Res. II 55, 1883-1896.
- Caporaso, J.G., Kuczynski, J., Stombaugh, J., Bittinger, K., Bushman, F.D., Costello, E.K.,
  Fierer, N., Peña, A.G., Goodrich, J.K., Gordon, J.I., Huttley, G.A., Kelley, S.T.,
  Knights, D., Koenig, J.E., Ley, R.E., Lozupone, C.A., McDonald, D., Muegge, B.D.,
  Pirrung, M., Reeder, J., Sevinsky, J.R., Turnbaugh, P.J., Walters, W.A., Widmann, J.,
  Yatsunenko, T., Zaneveld, J., Knight, R., 2010. QIIME allows analysis of highthroughput community sequencing data. Nature methods 7, 335-336.
- Casper, R.M., Jarman, S.N., Deagle, B.E., Gales, N.J., Hindell, M.A., 2007a. Detecting prey from DNA in predator scats: a comparison with morphological analysis, using *Arctocephalus* seals fed a known diet. J. Exp. Mar. Biol. Ecol. 347, 144-154.
- Casper, R.M., Jarman, S.N., Gales, N., Hindell, M., 2007b. Combining DNA and morphological analyses of faecal samples improves insight into trophic interactions: a case study using a generalist predator. Mar. Biol. 152, 815-825.
- CCAS, 1972. Convention for the Conservation of Antarctic Seals, http://www.ats.aq/index\_e.htm (accessed June 2015)
- Chambellant, M., Beauplet, G., Guinet, C., Georges, J.-Y., 2003. Long-term evaluation of pup growth and preweaning survival rates in subantarctic fur seals, *Arctocephalus tropicalis*, on Amsterdam Island. Can. J. Zool. 81, 1222-1232.
- Chastel, O., Weimerskirch, H., Jouventin, P., 1993. High annual variability in reproductive success and survival of an Antarctic seabird, the snow petrel *Pagodroma nivea*. Oecol. 94, 278-285.
- Cherel, Y., Bocher, P., De Broyer, C., Hobson, K.A., 2002. Food and feeding ecology of the sympatric thin-billed Pachyptila belcheri and Antarctic P. desolata prions at Iles Kerguelen, Southern Indian Ocean. Mar. Ecol. Prog. Ser. 228, 263-281.
- Cherel, Y., Hobson, K.A., Guinet, C., Vanpe, C., 2007. Stable isotopes document seasonal changes in trophic niches and winter foraging individual specialization in diving predators from the Southern Ocean. J. Anim. Ecol. 76, 826-836.

- Cherel, Y., Hobson, K.A., Weimerskirch, H., 2005. Using stable isotopes to study resource acquisition and allocation in procellariiform seabirds. Oecol. 145, 533-540.
- Clark, T.D., Sandblom, E., Hinch, S.G., Patterson, D.A., Frappell, P.B., Farrell, A.P., 2010. Simultaneous biologging of heart rate and acceleration, and their relationships with energy expenditure in free-swimming sockeye salmon (*Oncorhynchus nerka*). J. Comp. Physiol. B 180, 673-684.
- Clarke, M.R., 1962. The identification of cephalopod beaks and the relationship between beak size and total body weight. Bull. Br. Mus. Nat. Hist. 8, 312.
- Clarke, M.R., 1986. A Handbook for the identification of cephalopod beaks. Clarendon Press ;Oxford University Press, Oxford [Oxfordshire] :New York.
- Costa, C., 1993. The relationship between reproductive and foraging energetics and the evolution of the pinnipedia. Symp. Zool. Soc. Lond. 66, 293-313.
- Costa, D., Gales, N., 2000. Foraging energetics and diving behavior of lactating New Zealand sea lions, *Phocarctos hookeri*. J. Exp. Biol. 203, 3655-3665.
- Costa, D.P., 1987. Isotopic methods for quantifying material and energy intake of freeranging marine mammals., in: Huntley, A.C., Costa, D.P., Worthy, G.A.J., Castellini, M.A. (Eds.), Approaches to marine mammal energetics, Allen Press, Lawrence, KA, pp. 43-66.
- Costa, D.P., 1991. Reproductive and foraging energetics of pinnipeds: implications for life history patterns., in: Renouf, D. (Ed.), Behavior of pinnipeds, Chapman and Hall, London, pp. 300-344.
- Costa, D.P., 2007. A conceptual model of the variation in parental attendance in response to environmental fluctuation: foraging energetics of lactating sea lions and fur seals. Aquat. Cons. Mar. Freshwat. Ecosys. 17, S44-S52.
- Costa, D.P., Croxall, J.P., Duck, C.D., 1989. Foraging energetics of Antarctic fur seals in relation to changes in prey availability. Ecology 70, 596-606.

- Costa, D.P., Gentry, R.L., 1986. Free-ranging energetics of northern fur seals, in: Gentry, R.L., Kooyman, G.L. (Eds.), Fur seals: maternal strategies on land and at sea, Princeton University Press, Princeton, NJ, pp. 79-101.
- Costa, D.P., Trillmich, F., 1988. Mass changes and metabolism during the perinatal fast A comparison between Antarctic (*Arctocephalus gazella*) and Galapagos fur seals (*Arctocephalus galapagoensis*). Physiol. Zool. 61, 160-169.
- Costa, D.P., Williams, T.M., 1999. Marine mammals energetics, in: Reynolds, J.E., Rommel, S.A. (Eds.), Biology of marine mammals, Smithsonian Institution Press, Washington, D.C., pp. 176-217.
- Coward, W., Prentice, A., Murgatroyd, P., Davies, H., Cole, T., Sawyer, M., Goldberg, G., Halliday, D., Macnamara, J., 1985. Measurement of CO<sub>2</sub> and water production rates in man using <sup>2</sup>H, <sup>18</sup>O labelled H<sub>2</sub>O: comparisons between calorimeter and isotope values, in: van Es, A.J.H. (Ed.), Human energy metabolism: physical activity and energy expenditure measurements in epidemiological research based upon direct and indirect calorimetry., Instituut voor de Voeding, Stitchting, Nederlands, pp. 126-128.
- Cristescu, M.E., 2014. From barcoding single individuals to metabarcoding biological communities: towards an integrative approach to the study of global biodiversity. Trends Ecol. Evol. 29, 566-571.
- Dalton, A.J.M., Rosen, D.A.S., Trites, A.W., 2014a. Broad thermal capacity facilitates the primarily pelagic existence of northern fur seals (*Callorhinus ursinus*). Mar. Mamm. Sci. 30, 994-1013.
- Dalton, A.J.M., Rosen, D.A.S., Trites, A.W., 2014b. Season and time of day affect the ability of accelerometry and the doubly labeled water methods to measure energy expenditure in northern fur seals (*Callorhinus ursinus*). J. Exp. Mar. Biol. Ecol. 452, 125-136.
- Dassis, M., Rodríguez, D.H., Ieno, E.N., Davis, R.W., 2012. Submerged swimming and resting metabolic rates in Southern sea lions. J. Exp. Mar. Biol. Ecol. 432–433, 106-112.

- Davis, R.W., Brandon, E.A.A., Adams, T.C., Williams, T.M., Castellini, M.A., Loughlin, T.R., Calkins, D.G., 1996. Indices of reproductive effort, body condition and pup growth for Steller sea lions (*Eumetopias jubatus*) in Alaska, Steller sea lion recovery investigations in Alaska, 1992-1994, Alaska Department of Fish and Game, Wildlife Technical Bulletin 13., pp. 53-61.
- Davis, R.W., Williams, T.M., Kooyman, G.L., 1985. Swimming metabolism of yearling and adult harbor seals *Phoca vitulina*. Physiol. Zool. 58, 590-596.
- Deagle, B., Chiaradia, A., McInnes, J., Jarman, S., 2010. Pyrosequencing faecal DNA to determine diet of little penguins: is what goes in what comes out? Conserv. Genet. 11, 2039-2048.
- Deagle, B.E., Kirkwood, R., Jarman, S.N., 2009. Analysis of Australian fur seal diet by pyrosequencing prey DNA in faeces. Mol. Ecol. 18, 2022-2038.
- Deagle, B.E., Thomas, A.C., Shaffer, A.K., Trites, A.W., Jarman, S.N., 2013. Quantifying sequence proportions in a DNA-based diet study using Ion Torrent amplicon sequencing: which counts count? Mol. Ecol. Resour. 13, 620-633.
- Deagle, B.E., Tollit, D., Jarman, S.N., Hindell, M.A., Trites, A.W., Gales, N.J., 2005. Molecular scatology as a tool to study diet: analysis of prey DNA in scats from captive Steller sea lions. Mol. Ecol. 14, 1831-1842.
- Deagle, B.E., Tollit, D.J., 2007. Quantitative analysis of prey DNA in pinniped faeces: potential to estimate diet composition? Conserv. Genet. 8, 743-747.
- Doidge, D.W., Croxall, J.P., 1989. Factors affecting weaning weight in Antarctic fur seals, *Arctocephalus gazella* at South Georgia. Polar Biol. 9, 155-160.
- Donohue, M.J., Costa, D.P., Goebel, M.E., Baker, J.D., 2000. The ontogeny of metabolic rate and thermoregulatory capabilities of northern fur seal, *Callorhinus ursinus*, pups in air and water. J. Exp. Biol. 203, 1003-1016.
- Dragoo, D.E., 1991. Food habits and productivity of kittiwakes and murres at St. George Island, Alaska, MSc.thesis, University of Alaska Fairbanks, Fairbanks, Alaska.

- Dudley, P.N., Bonazza, R., Jones, T.T., Wyneken, J., Porter, W.P., 2014. Leatherbacks swimming in *silico*: modeling and verifying their momentum and heat balance using computational fluid dynamics. PLoS ONE 9, e110701.
- Edgar, R.C., 2010. Search and clustering orders of magnitude faster than BLAST. Bioinformatics 26, 2460-2461.
- Elliott, K.H., Le Vaillant, M., Kato, A., Speakman, J.R., Ropert-Coudert, Y., 2013. Accelerometry predicts daily energy expenditure in a bird with high activity levels. Biol. Lett. 9. 20120919
- Fahlman, A., Svärd, C., Rosen, D.A.S., Wilson, R.P., Trites, A.W., 2013. Activity as a proxy to estimate metabolic rate and to partition the metabolic cost of diving vs. breathing in pre- and post-fasted Steller sea lions. Aquatic Biol. 18, 175-184.
- Fahlman, A., Wilson, R., Svard, C., Rosen, D.A.S., Trites, A.W., 2008. Activity and diving metabolism correlate in Steller sea lion *Eumetopias jubatus*. Aquatic Biol. 2, 75-84.
- Fedak, M.A., Arnbom, T., Boyd, I.L., 1996. The relation between the size of southern elephant seal mothers, the growth of their pups, and the use of maternal energy, fat, and protein during lactation. Physiol. Zool. 69, 887-911.
- Fernández, D.A., Lattuca, M.E., Boy, C.C., Pérez, A.F., Ceballos, S.G., Vanella, F.A., Morriconi, E.R., Malanga, G.F., Aureliano, D.R., Rimbau, S., Calvo, J., 2009. Energy density of sub-Antarctic fishes from the Beagle Channel. Fish. Physiol. Biochem. 35, 181-188.
- Fischer, B., Taborsky, B., Dieckmann, U., 2009. Unexpected patterns of plastic energy allocation in stochastic environments. Amer. Nat. 173, E108-E120.
- Fish, F.E., Innes, S., Ronald, K., 1988. Kinematics and estimated thrust production of swimming harp and ringed seals. J. Exp. Biol. 137, 157-173.
- Forcada, J., Staniland, I.J., 2009. Antarctic fur seal Arctocephalus gazella, in: Perrin, W.F., Wursig, B.G., Thewissen, J.G.M. (Eds.), Encyclopedia of marine mammals, Academic Press, San Diego, pp. 36-42.

- France, R.L., 1995. Carbon-13 enrichment in benthic compared to planktonic algae: food web implications. Mar. Ecol. Prog. Ser. 124, 307-312.
- Froget, G., Butler, P.J., Woakes, A.J., Fahlman, A., Kuntz, G., Le Maho, Y., Handrich, Y., 2004. Heart rate and energetics of free-ranging king penguins (*Aptenodytes patagonicus*). J. Exp. Biol. 207, 3917-3926.
- Frost, K.J., Lowry, L.F., 1981. Trophic importance of some marine gadids in northern Alaska and their body-otolith size relationship. Fish. Bull. 79, 187-192.
- Gallon, S., Bailleul, F., Charrassin, J.B., Guinet, C., Bost, C.A., Handrich, Y., Hindell, M., 2013. Identifying foraging events in deep diving southern elephant seals, *Mirounga leonina*, using acceleration data loggers. Deep-Sea Res. II 88–89, 14-22.
- Garland, T., Adolph, S.C., 1994. Why not to do two-species comparative studies: limitations on inferring adaptation. Physiol. Zool. 67, 797-828.
- Gende, S.M., Quinn, T.P., Willson, M.F., Heintz, R., Scott, T.M., 2004. Magnitude and fate of salmon-derived nutrients and energy in a coastal stream ecosystem. J. Fresh. Ecol. 19, 149-160.
- Gentry, R.L., 1998. Behavior and Ecology of the Northern Fur Seal. Princeton University Press, Princeton, New Jersey.
- Gentry, R.L., 2002. Northern fur seal *Callorhinus ursinus*. Pp. 813-817. *in* W. F. Perrin, B.
  Würsig, and H. G. M. Thewissen (eds.). Encyclopedia of Marine Mammals, Academic Press, San Diego.
- Gentry, R.L., Holt, J.R., 1986. Attendance behavior of northern fur seals, in: Gentry, R.L., Kooyman, G.L. (Eds.), Fur seals: Maternal strategies on land and at sea, Princeton University Press, Princeton, NJ, pp. 41-60.
- Gentry, R.L., Kooyman, G.L. (Eds.), 1986. Fur seals- maternal strategies on land and at sea. Princeton University Press.
- Georges, J.Y., Bonadonna, F., Guinet, C., 2000. Foraging habitat and diving activity of lactating Subantarctic fur seals in relation to sea-surface temperatures at Amsterdam Island. Mar. Ecol. Prog. Ser. 196, 291-304.

- Georges, J.Y., Groscolas, R., Guinet, C., Robin, J.P., 2001. Milking strategy in subantarctic fur seals *Arctocephalus tropicalis* breeding on Amsterdam Island: Evidence from changes in milk composition. Physiol. Biochem. Zool. 74, 548-559.
- Gerlinsky, C.D., Rosen, D.A.S., Trites, A.W., 2013. High diving metabolism results in a short aerobic dive limit for Steller sea lions (*Eumetopias jubatus*). J. Comp. Physiol. B 183, 699-708.
- Gleiss, A.C., Dale, J.J., Holland, K.N., Wilson, R.P., 2010. Accelerating estimates of activity-specific metabolic rate in fishes: Testing the applicability of acceleration data-loggers. J. Exp. Mar. Biol. Ecol. 385, 85-91.
- Gleiss, A.C., Gruber, S.H., Wilson, R.P., 2009. Multi-channel data-logging: towards determination of behaviour and metabolic rate in free-swimming sharks, in: Nielsen, J.L., Arrizabalaga, H., Fragoso, Nielsen, J.L. Arrizabalaga, H., Fragoso, N., Hobday A., Lutcavage M., Sibert J. (Eds.), Tagging and tracking of marine animals with electronic devices, Springer, Fairbanks, AK, pp. 211-228.
- Gleiss, A.C., Wilson, R.P., Shepard, E.L.C., 2011. Making overall dynamic body acceleration work: on the theory of acceleration as a proxy for energy expenditure. Methods Ecol. Evol. 2, 23-33.
- Goebel, M.E., 2002. Northern fur seal lactation, attendance and reproductive success in two years of contrasting oceanography, PhD thesis, University of California, Santa Cruz, 227pp.
- Goldsworthy, S.D., Lea, M.-A., Guinet, C., 2004. Comparison of mass-transfer and isotopic dilution methods for estimating milk intake in Antarctic fur seal pups. Polar Biol. 27, 801-809.
- Gomez-Laich, A., Wilson, R.P., Gleiss, A.C., Shepard, E.L.C., Quintana, F., 2011. Use of overall dynamic body acceleration for estimating energy expenditure in cormorants: Does locomotion in different media affect relationships? J. Exp. Mar. Biol. Ecol. 399, 151-155.

- Gomez-Laich, A., Wilson, R.P., Quintana, F., Shepard, E.L.C., 2008. Identification of imperial cormorant *Phalacrocorax atriceps* behaviour using accelerometers. Endang. Species Res. 10, 29-37.
- Gomez-Laich, A., Wilson, R.P., Shepard, E.L.C., Quintana, F., 2013. Energy expenditure and food consumption of foraging Imperial cormorants in Patagonia, Argentina. Mar. Biol. 160, 1697-1707.
- Gorbatenko, K.M., Kiyashko, S.I., Lazhentsev, A.Y., Nadtochii, V.A., Savin, A.B., 2008. Benthic-pelagic trophic interactions within the fish assemblage in the western Bering Sea shelf area according to stomach content analysis and ratios of C and N stable isotopes. Russ. J. Mar. Biol. 34, 497-506.
- Goundie, E.T., Rosen, D.A.S., Trites, A.W., 2015. Low prey abundance leads to less efficient foraging behavior in Steller sea lions. J. Exp. Mar. Biol. Ecol. 470, 70-77.
- Green, J.A., Halsey, L.G., Wilson, R.P., Frappell, P.B., 2009. Estimating energy expenditure of animals using the accelerometry technique: activity, inactivity and comparison with the heart-rate technique. J. Exp. Biol. 212, 471-482.
- Grémillet, D.J.H., Plös, A.L., 1994. The use of stomach temperature records for the calculation of daily food intake in cormorants. J. Exp. Biol. 189, 105-115.
- Gudmundson, C.J., Zeppelin, T.K., Ream, R.R., 2003. Comparison of two methodologies for determining diet in northern fur seals (*Callorhinus ursinus*), National Marine Fisheries Service, Seattle, WA, 104(3):445-455.
- Gudmundson, C.J., Zeppelin, T.K., Ream, R.R., 2006. Application of two methods for determining diet of northern fur seals (*Callorhinus ursinus*). Fish. Bull. 104, 445-455.
- Guinet, C., Dubroca, L., Lea, M.A., Goldsworthy, S., Cherel, Y., Duhamel, G., Bonadonna,
   F., Donnay, J.P., 2001. Spatial distribution of foraging in female Antarctic fur seals
   Arctocephalus gazella in relation to oceanographic variables: a scale-dependent
   approach using geographic information systems. Mar. Ecol. Prog. Ser. 219, 251-264.
- Guinet, C., Vacquié-Garcia, J., Picard, B., Bessigneul, G., Lebras, Y., Dragon, A., Viviant,M., Arnould, J., Bailleul, F., 2014. Southern elephant seal foraging success in relation

to temperature and light conditions: insight into prey distribution. Mar. Ecol. Prog. Ser. 499, 285-301.

- Hall, A., McConnell, B.J., Barker, R.J., 2001. Factors affecting first-year survival in grey seals and their implications for life history strategy. J. Anim. Ecol. 70, 138-149.
- Halliday, D., Miller, A.G., 1977. Precise measurement of total body water using trace quantities of deuterium oxide. Biol. Mass Spectrom. 4, 82-87.
- Halsey, L.G., Green, J.A., Wilson, R.P., Frappell, P.B., 2009a. Accelerometry to estimate energy expenditure during activity: best practice with data loggers. Physiol. Biochem. Zool. 82, 396-404.
- Halsey, L.G., Jones, T.T., Jones, D.R., Liebsch, N., Booth, D.T., 2011a. Measuring energy expenditure in sub-adult and hatchling sea turtles via accelerometry. PLoS ONE 6, e22311.
- Halsey, L.G., Shepard, E.L.C., Hulston, C.J., Venables, M.C., White, C.R., Jeukendrup, A.E., Wilson, R.P., 2008. Acceleration versus heart rate for estimating energy expenditure and speed during locomotion in animals: tests with an easy model species, *Homo sapiens*. Zoology 111, 231-241.
- Halsey, L.G., Shepard, E.L.C., Quintana, F., Gomez Laich, A., Green, J.A., Wilson, R.P., 2009b. The relationship between oxygen consumption and body acceleration in a range of species. Comp. Biochem. Physiol. A. 152, 197-202.
- Halsey, L.G., Shepard, E.L.C., Wilson, R.P., 2011b. Assessing the development and application of the accelerometry technique for estimating energy expenditure. Comp. Biochem. Physiol. A. 158, 305-314.
- Halsey, L.G., White, C.R., 2010. Measuring energetics and behaviour using accelerometry in cane toads *Bufo marinus*. PLoS ONE 5, e10170.
- Halsey, L.G., White, C.R., Enstipp, M.R., Wilson, R.P., Butler, P.J., Martin, G.R., Grémillet,
  D., Jones, D.R., 2011c. Assessing the validity of the accelerometry technique for estimating the energy expenditure of diving double-crested cormorants *Phalacrocorax auritus*. Physiol. Biochem. Zool. 84, 230-237.

- Hanchet, S., Horn, P., Stevenson, M., 2003. Fishing in the ice: is it sustainable? Water Atmos. 11, 24–25.
- Harvey, J.T., Antonelis, G.A., 1994. Biases associated with non-lethal methods of determining the diet of northern elephant seals. Mar. Mamm. Sci. 10, 178-187.
- Harvey, J.T., Loughlin, T.R., Perez, M.A., Oxman, D.S., 2000. Relationship between fish size and otolith length for 63 species of fishes from the eastern north Pacific Ocean, p. 36.
- Hauser, D.W., Allen, C.S., Rich, H.B., Quinn, T.P., 2008. Resident harbor seals (*Phoca vitulina*) in Iliamna Lake, Alaska: summer diet and partial consumption of adult sockeye salmon (*Oncorhynchus nerka*). Aquat. Mamm. 34, 303-309.
- Hays, G.C., Marshall, G.J., Seminoff, J.A., 2007. Flipper beat frequency and amplitude changes in diving green turtles, *Chelonia mydas*. Mar. Biol. 150, 1003-1009.
- Hendry, A.P., Berg, O.K., 1999. Secondary sexual characters, energy use, senescence, and the cost of reproduction in sockeye salmon Can. J. Zool. 77, 1663-1681.
- Hind, A.T., Gurney, W.S.C., 1997. The metabolic cost of swimming in marine homeotherms.J. Exp. Biol. 200, 531-542.
- Hindell, M., Crocker, D.E., Mori, Y., Tyack, P.L., 2010. Foraging behaviour, in: Boyd, I.L., Bowen, W.D., Iverson, S.J. (Eds.), Marine mammal ecology and conservation: a handbook of techniques, Oxford University Press, Oxford, pp. 243-262.
- Hindle, A.G., Rosen, D.A.S., Trites, A.W., 2010. Swimming depth and ocean currents affect transit costs in Steller sea lions *Eumetopias jubatus*. Aquatic Biol. 10, 139-148.
- Hobson, K.A., Piatt, J.F., Pitocchelli, J., 1994. Using stable isotopes to determine seabird trophic relationships. J. Anim. Ecol., 786-798.
- Hobson, K.A., Welch, H.E., 1992. Determination of trophic relationships within a high Arctic marine food web using stable-isotope analysis. Mar. Ecol. Prog. Ser. 84, 9-18.
- Hoelzel, A.R., 2002. Marine mammal biology: an evolutionary approach. Blackwell Science, Oxford.

- Honkalehto, T., McCarthy, A., Ressler, P., Williams, K., Jones, D., 2012. Results of the acoustic-trawl survey of walleye pollock (*Theragra chalcogramma*) on the U.S. and Russian Bering sea shelf in June - August 2010 (DY1006), In: Alaska Fisheries Science Center Processed Rep., (Ed.), NOAA, Natl. Mar. Fish. Serv., Seattle WA, p. 59 pp.
- Hucke-Gaete, R., Osman, L.P., Moreno, C.A., Torres, D., 2004. Examining natural population growth from near extinction: the case of the Antarctic fur seal at the South Shetlands, Antarctica. Polar Biol. 27, 304-311.
- Hunt Jr, G.L., Eppley, Z., Burgeson, B., Squibb, R., 1980. Reproductive ecology, foods and foraging areas of seabirds nesting on the Pribilof Islands, 1975–1979, In: studies, B. (Ed.), Environmental assessment of the Alaskan continental shelf: Final report of the principal investigators, NOAA Environmental Research Lab, Boulder, Colorado, pp. 1-258.
- Inchausti, P., Guinet, C., Koudil, M., Durbec, J.P., Barbraud, C., Weimerskirch, H., Cherel, Y., Jouventin, P., 2003. Inter-annual variability in the breeding performance of seabirds in relation to oceanographic anomalies that affect the Crozet and the Kerguelen sectors of the Southern Ocean. J. Avian Biol. 34, 170-176.
- Insley, S.J., 2008. Acoustic determination of activity and flipper stroke rate in foraging northern fur seal females. Endang. Species Res. 4, 147-155.
- IPCC, 2007. Climate change 2007: The physical science basis. Cambridge University Press, Cambridge.
- Iverson, S.J., Bowen, W.D., Boness, D.J., Oftedal, O.T., 1993. The effect of maternal size and milk energy output on pup growth in grey seals (*Halichoerus grypus*). Physiol. Zool. 66, 61-88.
- Iverson, S.J., Frost, K.J., Lang, S.L.C., 2002. Fat content and fatty acid composition of forage fish and invertebrates in Prince William Sound, Alaska: factors contributing to among and within species variability. Mar. Ecol. Prog. Ser. 241, 161-181.

- Iwata, T., Sakamoto, K.Q., Takahashi, A., Edwards, E.W.J., Staniland, I.J., Trathan, P.N., Naito, Y., 2012. Using a mandible accelerometer to study fine-scale foraging behavior of free-ranging Antarctic fur seals. Mar. Mamm. Sci. 28, 345-357.
- Jensen, R.G., 1996. The lipids in human milk. Prog. Lipid Res. 35, 53-92.
- Jobling, M., Breiby, A., 1986. The use and abuse of fish otoliths in studies of feeding habits of marine piscivores. Sarsia 71, 265-274.
- Jodice, P.G.R., Roby, D.D., Turco, K.R., Suryan, R.M., Irons, D.B., Piatt, J.F., Shultz, M.T., Roseneau, D.G., Kettle, A.B., Anthony, J.A., 2006. Assessing the nutritional stress hypothesis: relative influence of diet quantity and quality on seabird productivity. Mar. Ecol. Prog. Ser. 325, 267-279.
- Jouma'a, J., Le Bras, Y., Richard, G., Vacquié-Garcia, J., Picard, B., El Ksaby, N., Guinet, C., 2015. Adjustment of diving behaviour with prey encounters and body condition in a deep diving predator: the Southern elephant seal. Funct. Ecol. In Press.
- Karnovsky, N.J., Hobson, K.A., Iverson, S.J., 2012. From lavage to lipids: estimating diets of seabirds. Mar. Ecol. Prog. Ser. 451, 263-284.
- Kembel, S.W., Wu, M., Eisen, J.A., Green, J.L., 2012. Incorporating 16S gene copy number information improves estimates of microbial diversity and abundance. PLoS Comput Biol 8, e1002743.
- Kerley, G.I.H., 1985. Pup growth in the fur seals Arctocephalus tropicalis and A. gazella on Marion Island. Journal of Zoology, London 205, 315-324.
- Kitaysky, A.S., Golubova, E.G., 2000. Climate change causes contrasting trends in reproductive performance of planktivorous and piscivorous alcids. J. Anim. Ecol. 69, 248-262.
- Kitaysky, A.S., Kitaiskaia, E.V., Piatt, J.F., Wingfield, J.C., 2006. A mechanistic link between chick diet and decline in seabirds? P. Roy. Soc. Lond. B. 273, 445-450.
- Kleiber, M., 1947. Body size and metabolic rate. Physiol. Rev. 27, 511-541.
- Kleiber, M., 1975. The fire of life: an introduction to animal energetics. Robert E. Krieger Publ. Co., New York.

- Kline, T.C., 2010. Stable carbon and nitrogen isotope variation in the northern lampfish and Neocalanus, marine survival rates of pink salmon, and meso-scale eddies in the Gulf of Alaska. Prog. Oceanogr. 87, 49-60.
- Kock, K.H., 1992. Antarctic fish and fisheries. Cambridge University Press, Cambridge ; New York.
- Kojadinovic, J., Richard, P., Le Corre, M., Cosson, R.P., Bustamante, P., 2008. Effects of lipid extraction on  $\delta^{13}$ C and  $\delta^{15}$ N values in seabird muscle, liver and feathers. Waterbirds 31, 169-178.
- Kokubun, N., Kim, J.-H., Shin, H.-C., Naito, Y., Takahashi, A., 2011. Penguin head movement detected using small accelerometers: a proxy of prey encounter rate. J. Exp. Biol. 214, 3760-3767.
- Kooijman, S.A.L.M., 2000. Dynamic energy and mass budgets in biological systems. 2nd ed. Cambridge University Press, Cambridge, UK ; New York, NY, USA.
- Kooijman, S.A.L.M., Andersen, T., Kooi, B.W., 2004. Dynamic energy busget representations of stoichiometric constraints on population dynamics. Ecology 85, 1230-1243.
- Kooyman, G.L., 1989. Diverse divers : physiology and behavior. Springer-Verlag, Berlin ; New York.
- Kowalik, Z., Stabeno, P., 1999. Trapped motion around the Pribilof Islands in the Bering Sea. J. Geophys. Res.- Oceans 104, 25667-25684.
- Kuhn, C., Frey, E., 2012. Walking like caterpillars, flying like bats—pinniped locomotion. Palaeobio. Palaeoenv. 92, 197-210.
- Kuhn, C.E., Costa, D.P., 2006. Identifying and quantifying prey consumption using stomach temperature change in pinnipeds. J. Exp. Biol. 209, 4524-4532.
- Kuhn, C.E., Ream, R.R., Sterling, J.T., Thomason, J.R., Towell, R.G., 2014. Spatial segregation and the influence of habitat on the foraging behavior of northern fur seals (*Callorhinus ursinus*). Can. J. Zool. 92, 861-873.

- Kurle, C.M., 2002. Stable-isotope ratios of blood components from captive northern fur seals (*Callorhinus ursinus*) and their diet: applications for studying the foraging ecology of wild otariids. Can. J. Zool. 80, 902-909.
- Kurle, C.M., Worthy, G.A.J., 2001. Stable isotope assessment of temporal and geographic differences in feeding ecology of northern fur seals (*Callorhinus ursinus*) and their prey. Oecol. 126, 254-265.
- Lavigne, D.M., Barchard, W., Innes, S., Øritsland, N.A., 1982. Pinniped bioenergetics, Mammals in the seas: Small cetaceans, seals, sirenians, and sea otters, FAO Fisheries Series 5, Rome, pp. 191-235.
- Le Boeuf, B.J., Crocker, D.E., Grayson, J., Gedamke, J., Webb, P.M., Blackwell, S.B., Costa, D.P., 2000. Respiration and heart rate at the surface between dives in northern elephant seals. J. Exp. Biol. 203, 3265-3274.
- Lea, M.A., Cherel, Y., Guinet, C., Nichols, P.D., 2002a. Antarctic fur seals foraging in the Polar Frontal Zone: inter-annual shifts in diet as shown from fecal and fatty acid analyses. Mar. Ecol. Prog. Ser. 245, 281–297.
- Lea, M.A., Guinet, C., Cherel, Y., Duhamel, G., Dubroca, L., Pruvost, P., Hindell, M., 2006. Impacts of climatic anomalies on provisioning strategies of a Southern Ocean predator. Mar. Ecol. Prog. Ser. 310, 77-94.
- Lea, M.A., Guinet, C., Cherel, Y., Hindell, M., Dubroca, L., Thalmann, S., 2008. Colonybased foraging segregation by Antarctic fur seals at the Kerguelen Archipelago. Mar. Ecol. Prog. Ser. 358, 273-287.
- Lea, M.A., Hindell, M., Guinet, C., Goldsworthy, S., 2002b. Variability in the diving activity of Antarctic fur seals, *Arctocephalus gazella*, at Iles Kerguelen. Polar Biol. 25, 269-279.
- Lea, M.A., Johnson, D.E., Ream, R., Sterling, J., Melin, S., Gelatt, T., 2009. Extreme weather events influence dispersal of naive northern fur seals. Biol. Lett. 5, 252-257.

- Lea, M.A., Nichols, P.D., Wilson, G., 2002c. Fatty acid composition of lipid-rich myctophids and mackerel icefish (*Champsocephalus gunnari*) - southern ocean food-web implications. Polar Biol. 25, 843-854.
- Learmonth, J.A., MacLeod, C.D., Santos, M.B., Pierce, G.J., Crick, H.Q.P., Robinson, R.A., 2006. Potential effect of climate change on marine mammals. Oceanogr. Mar. Biol. Annu. Rev. 44, 431-464.
- Lemon, W.C., 1991. Fitness consequences of foraging behaviour in the zebra finch. Nature 352, 153-155.
- Lescroël, A., Ballard, G., Toniolo, V., Barton, K.J., Wilson, P.R., Lyver, P.O.B., Ainley, D.G., 2010. Working less to gain more: when breeding quality relates to foraging efficiency. Ecology 91, 2044-2055.
- Liebsch, N., Wilson, R.P., Bornemann, H., Adelung, D., Plötz, J., 2007. Mouthing off about fish capture: Jaw movement in pinnipeds reveals the real secrets of ingestion. Deep-Sea Res. II 54, 256-269.
- Lifson, N., McClintock, R., 1966. Theory of use of the turnover rates of body water for measuring energy and material balance. J. Theor. Biol. 12, 46-74.
- Loeb, V., Siegel, V., Holm-Hansen, O., Hewitt, R., Fraser, W., Trivelplece, W., Trivelplece, S., 1997. Effects of sea-ice extent and krill or salp dominance on the Antarctic food web. Nature 387, 897-900.
- Logerwell, E., Christiansen, R.A., 2000. Energy density of Steller sea lion prey in western Alaska: species, regional, and seasonal differences. Poster presentation Alaska Marine Science Symposium, Anchorage, AK.
- Lunn, N.J., Boyd, I.L., 1993. Effects of maternal age and condition on parturition and the perinatal period of Antarctic fur seals. Journal of Zoology, London 229, 55-67.
- Lunn, N.J., Boyd, I.L., Barton, T., Croxall, J.P., 1993. Factors affecting the growth rate and mass at weaning of Antarctic fur seals at Bird Island, South Georgia. J. Mammal. 74, 908-919.

- Lunn, N.J., Boyd, I.L., Croxall, J.P., 1994. Reproductive performance of female Antarctic fur seals: the influence of age, breeding experience, environmental variation and individual quality. J. Anim. Ecol. 63, 827-840.
- Luque, S.P., Arnould, J.P.Y., Miller, E.H., Cherel, Y., Guinet, C., 2007. Foraging behaviour of sympatric Antarctic and subantarctic fur seals: does their contrasting duration of lactation make a difference? Mar. Biol. 152, 213-224.
- Luque, S.P., Guinet, C., 2007. A Maximum likelihood approach for identifying dive bouts improves accuracy, precision and objectivity. Behaviour 144, 1315-1332.
- MacArthur, R.H., Pianka, E.R., 1966. On optimal use of a patchy environment. Am. Nat. 100, 603–609.
- Mace, G.M., Balmford, A., 2000. Patterns and processes in contemporary mammalian extinction., in: Entwistle, A., Dunstone, N. (Eds.), Priorities for the conservation of mammalian diversity: has the panda had its day?, Cambridge University Press, Cambridge, U.K.; New York, NY, USA, pp. 27-52.
- Maresh, J.L., Simmons, S.E., Crocker, D.E., McDonald, B.I., Williams, T.M., Costa, D.P., 2014. Free-swimming northern elephant seals have low field metabolic rates that are sensitive to an increased cost of transport. J. Exp. Biol. 217, 1485-1495.
- Mattlin, R.H., 1981. Pup growth of the New Zealand fur seal *Arctocephalus forsteri* on the Open Bay Islands, New Zealand. Journal of Zoology, London 193, 305-314.
- McCafferty, D.J., Boyd, I.L., Walker, T.R., Taylor, R.I., 1998. Foraging responses of Antarctic fur seals to changes in the marine environment. Mar. Ecol. Prog. Ser. 166, 285-299.
- McMahon, C.R., Burton, H.R., Bester, M., 2000. Weaning mass and the future survival of juvenile southern elephant seals, *Mirounga leonina*, at Macquarie Island. Antarct. Sci. 12, 149-153.
- McPhee, J.M., Rosen, D.A.S., Andrews, R.D., Trites, A.W., 2003. Predicting metabolic rate from heart rate in juvenile Steller sea lions *Eumetopias jubatus*. J. Exp. Biol. 206, 1941-1951.

- Miller, P.J.O., Biuw, M., Watanabe, Y.Y., Thompson, D., Fedak, M.A., 2012. Sink fast and swim harder! Round-trip cost-of-transport for buoyant divers. J. Exp. Biol. 215, 3622-3630.
- Moore, J.K., Abbott, M.R., Richman, J.G., 1999. Location and dynamics of the Antarctic Polar Front from satellite sea surface temperature data. J. Geophys. Res.-Oceans 104, 3059-3073
- Nagy, K.A., 1980. CO2 production in animals: analysis of potential errors in the doubly labeled water method. Am. J. Physiol. Regul. Integr. Comp. Physiol. 238, 466-473.
- Nagy, K.A., 1983. The doubly-labeled water (<sup>3</sup>HH<sup>18</sup>O) method: a guide to its use., UCLA publications (Ed.), University of California, Los Angeles, CA,. 45pp.
- Nagy, K.A., Costa, D.P., 1980. Water flux in animals: analysis of potential errors in the tritiated water method. Am. J. Physiol. 238, R454-465.
- Naito, Y., Costa, D.P., Adachi, T., Robinson, P.W., Fowler, M., Takahashi, A., 2013. Unravelling the mysteries of a mesopelagic diet: a large apex predator specializes on small prey. Funct. Ecol. 27, 710-717.
- Napp, J.M., Kendall, A.W., Schumacher, J.D., 2000. A synthesis of biological and physical processes affecting the feeding environment of larval walleye pollock (*Theragra chalcogramma*) in the eastern Bering Sea. Fish. Oceanog. 9, 147-162.
- Nathan, R., Spiegel, O., Fortmann-Roe, S., Harel, R., Wikelski, M., Getz, W.M., 2012. Using tri-axial acceleration data to identify behavioral modes of free-ranging animals: general concepts and tools illustrated for griffon vultures. J. Exp. Biol. 215, 986-996.
- Newsome, S.D., Clementz, M.T., Koch, P.L., 2010. Using stable isotope biogeochemistry to study marine mammal ecology. Mar. Mamm. Sci. 26, 509-572.
- NMFS, 1993. Final Conservation Plan for the northern fur seal (*Callorhinus ursinus*). Prepared by the National Marine Mammal Laboratory/Alaska Fisheries Science Center, Seattle, Washington, and the Office of Protected Resources/National Marine Fisheries Service, Silver Spring, Maryland, p. 80.

- Noda, T., Okuyama, J., Koizumi, T., Arai, N., Kobayashi, M., 2012. Monitoring attitude and dynamic acceleration of free-moving aquatic animals using a gyroscope. Aquatic Biol. 16, 265-276.
- Nordstrom, C.A., Battaile, B.C., Cotté, C., Trites, A.W., 2013. Foraging habitats of lactating northern fur seals are structured by thermocline depths and submesoscale fronts in the eastern Bering Sea. Deep-Sea Res. II 88–89, 78-96.
- Oftedal, O.T., 1984. Body size and reproductive strategy as correlates of milk energy output in lactating mammals. Acta Zool. Fennica 171, 183-186.
- Olesiuk, P.F., 1993. Annual prey consumption by harbor seals (*Phoca vitulina*) in the Strait of Georgia, British Columbia. Fish. Bull. 91, 491-515.
- Olesiuk, P.F., Bigg, M.A., Ellis, G.M., Crockford, S.J., Wigen, R.J., 1990. An assessment of the feeding habits of harbour seals (*Phoca vitulina*) in the Strait of Georgia, British Columbia, based on scat analysis, Dept. of Fisheries and Oceans, pp. 1-135.
- Orlov, A., Binohlan, C., 2009. Length-weight relationships of deep-sea fishes from the western Bering Sea. J. Appl. Ichthyol. 25, 223-227.
- Orr, A.J., Harvey, J.T., 2001. Quantifying errors associated with using fecal samples to determine the diet of the California sea lion (*Zalophus californianus*). Can. J. Zool. 79, 1080-1087.
- Orr, A.J., Newsome, S.D., DeLong, R.L., 2008. Variation in stable carbon and nitrogen isotope values from multiple tissues of California sea lions (*Zalophus californianus*), Proceedings of the Seventh California Islands Symposium, Oxnard, California February.
- Osman, L.P., Moreno, C.A., Trites, A.W., 2010. Growth rates and differential investment in male and female Juan Fernandez fur seal pups. J. Mammal. 91, 1188–1196.
- Overland, J.E., Stabeno, P.J., 2004. Is the climate of the Bering Sea warming and affecting the ecosystem? EOS Trans. AGU 85.
- Paternoster, R., Brame, R., Mazerolle, P., Piquero, A., 1998. Using the correct statistical test for the equality of regression coefficients. Criminol. 36, 859-866.

- Patrick, S.C., Charmantier, A., Weimerskirch, H., 2013. Differences in boldness are repeatable and heritable in a long-lived marine predator. Ecol. Evol. 3, 4291-4299.
- Paul, A.J., Paul, J.M., 1999. Interannual and regional variations in body length, weight and energy content of age-0 Pacific herring from Prince William Sound, Alaska. J. Fish Biol. 54, 996-1001.
- Perez, M.A., 1994. Calorimetry measurements of energy value of some Alaskan fishes and squids, U.S. Department of Commerce, NOAA Technical Memorandum, NMFS-AFSC-32 p. 35.
- Perez, M.A., Bigg, M., 1981. An assessment of the feeding habits of the northern fur seal in the Eastern North Pacific ocean and Eastern Bering Sea. US Natl. Mar. Fish. Serv.., Northwest, and Alaska Fish. Cent., Natl. Mar. Mammal Lab. Seattle, WA, 146pp
- Perez, M.A., Bigg, M.A., 1986. Diet of northern fur seals, *Callorhinus ursinus*, off western North America. Fish. Bull. 84, 957-971.
- Perez, M.A., Mooney, E.E., 1986. Increased food and energy consumption of lactating northern fur seals, *Callorhinus ursinus*. Fish. Bull. 84, 371-381.
- Pomeroy, P.P., Fedak, M.A., Rothery, P., Anderson, S., 1999. Consequences of maternal size for reproductive expenditure and pupping success of grey seals at North Rona, Scotland. J. Anim. Ecol. 68, 235-253.
- Ponganis, P.J., Kooyman, G.L., Winter, I.M., Starke, L.N., 1997. Heart rate and plasma lactate responses during submerged swimming and trained diving in California sea lions, *Zalophus californianus*. J. Comp. Physiol. B 167, 9-16.
- Pyke, G.H., 1984. Optimal foraging theory: a critical review. Ann. Rev. Ecol. Syst. 15, 523-575.
- Qasem, L., Cardew, A., Wilson, A., Griffiths, I., Halsey, L.G., Shepard, E.L.C., Gleiss, A.C., Wilson, R., 2012. Tri-axial dynamic acceleration as a proxy for animal energy expenditure; should we be summing values or calculating the vector? PLoS ONE 7, e31187.

- Reid, K., Davis, D., Staniland, I.J., 2006. Spatial and temporal variability in the fish diet of Antarctic fur seal (*Arctocephalus gazella*) in the Atlantic sector of the Southern Ocean. Can. J. Zool. 84, 1025-1037.
- Reid, K., Forcada, J., 2005. Causes of offspring mortality in the Antarctic fur seal, *Arctocephalus gazella*: the interaction of density dependence and ecosystem variability. Can. J. Zool. 83, 604-609.
- Richard, G., Vacquié-Garcia, J., Jouma'a, J., Picard, B., Génin, A., Arnould, J.P.Y., Bailleul,
  F., Guinet, C., 2014. Variation in body condition during the post-moult foraging trip of southern elephant seals and its consequences on diving behaviour. J. Exp. Biol. 217, 2609-2619.
- Ritchie, M.E., 1990. Optimal foraging and fitness in Columbian ground squirrels. Oecol. 82, 56-67.
- Robson, B.W., 2001. The relationship between foraging areas and breeding sites of lactating northern fur seals *Callorhinus ursinus* in the eastern Bering Sea, MSc. Thesis, School of Aquatic and Fisheries Science, University of Washington, Seattle, WA, p. 67.
- Robson, B.W., Goebel, M.E., Baker, J.D., Ream, R.R., Loughlin, T.R., Francis, R.C., Antonelis, G.A., Costa, D.P., 2004. Separation of foraging habitat among breeding sites of a colonial marine predator, the northern fur seal (*Callorhinus ursinus*). Can. J. Zool. 82, 20-29.
- Rosen, D.A.S., Trites, A.W., 1997. Heat increment of feeding in Steller sea lions, *Eumetopias jubatus*. Comp. Biochem. Physiol. A. 118, 877-881.
- Rosen, D.A.S., Trites, A.W., 2000. Digestive efficiency and dry-matter digestibility in Steller sea lions fed herring, pollock, squid, and salmon. Can. J. Zool. 78, 234-239.
- Sakamoto, K.Q., Sato, K., Ishizuka, M., Watanuki, Y., Takahashi, A., Daunt, F., Wanless, S., 2009. Can ethograms be automatically generated using body acceleration data from free-ranging birds? PLoS ONE 4, e5379.

- Sato, K., Mitani, Y., Cameron, M.F., Siniff, D.B., Naito, Y., 2003. Factors affecting stroking patterns and body angle in diving Weddell seals under natural conditions. J. Exp. Biol. 206, 1461-1470.
- Sato, K., Watanuki, Y., Takahashi, A., Miller, P.J.O., Tanaka, H., Kawabe, R., Ponganis,
  P.J., Handrich, Y., Akamatsu, T., Watanabe, Y., Mitani, Y., Costa, D.P., Bost, C.-A.,
  Aoki, K., Amano, M., Trathan, P.N., Shapiro, A., Naito, Y., 2007. Stroke frequency,
  but not swimming speed, is related to body size in free-ranging seabirds, pinnipeds
  and cetaceans. P. Roy. Soc. Lond. B. 274, 471-477.
- Schoeller, D.A., 1988. Measurement of energy expenditure in free-living humans by using doubly labeled water. J. Nutr. 118, 1278-1289.
- Schreer, J.F., Kovacs, K.M., O'Hara Hines, R.J., 2001. Comparative diving patterns of pinnipeds and seabirds. Ecol. Monogr. 71, 137-162.
- Serreze, M.C., Walsh, J.E., Chapin, F.S., III, Osterkamp, T., Dyurgerov, M., Romanovsky, V., Oechel, W.C., Morison, J., Zhang, T., Barry, R.G., 2000. Observational evidence of recent change in the Northern high-latitude environment. Climatic Change 46, 159-207.
- Shaffer, S.A., Costa, D.P., Weimerskirch, H., 2003. Foraging effort in relation to the constraints of reproduction in free-ranging albatrosses. Funct. Ecol. 17, 66-74.
- Shepard, E.L.C., Wilson, R.P., Halsey, L.G., Quintana, F., Laich, A.G., Gleiss, A.C., Liebsch, N., Myers, A.E., Norman, B., 2008. Derivation of body motion via appropriate smoothing of acceleration data. Aquatic Biol. 4, 235-241.
- Sinclair, E.H., Loughlin, T.R., Pearcy, W., 1994. Prey selection by northern fur seals (*Callorhinus ursinus*) in the eastern Bering Sea. Fish. Bull. 92, 132-156.
- Sinclair, E.H., Vlietstra, L.S., Johnson, D.S., Zeppelin, T.K., Byrd, G.V., Springer, A.M., Ream, R.R., Hunt, G.L., 2008. Patterns in prey use among fur seals and seabirds in the Pribilof Islands. Deep-Sea Res. II 55, 1897-1918.

- Siniff, D.B., Garrott, R.A., Rotella, J.J., Fraser, W.R., Ainley, D.G., 2008. Opinion Projecting the effects of environmental change on Antarctic seals. Antarct. Sci. 20, 425-435.
- Sinnott, R.W., 1984. Virtues of the Haversine. Sky Telescope 68, 159.
- Skinner, J., Norberg, S., Andrews, R., 2009. Head striking during fish capture attempts by Steller sea lions and the potential for using head surge acceleration to predict feeding behavior. Endang. Species Res. 10, 61-69.
- Skinner, J.P., Mitani, Y., Burkanov, V.N., Andrews, R.D., 2014. Proxies of food intake and energy expenditure for estimating the time–energy budgets of lactating northern fur seals *Callorhinus ursinus*. J. Exp. Mar. Biol. Ecol. 461, 107-115.
- Sousa, T., Domingos, T., Poggiale, J.-C., Kooijman, S.A.L.M., 2010. Dynamic energy budget theory restores coherence in biology. Philos. Trans. R. Soc. Lond. B 365, 3413-3428.
- Sparling, C.E., Fedak, M.A., 2004. Metabolic rates of captive grey seals during voluntary diving. J. Exp. Biol. 207, 1615-1624.
- Sparling, C.E., Fedak, M.A., Thompson, D., 2007. Eat now, pay later? Evidence of deferred food-processing costs in diving seals. Biol. Lett. 3, 95-99.
- Sparling, C.E., Thompson, D., Fedak, M.A., Gallon, S.L., Speakman, J.R., 2008. Estimating field metabolic rates of pinnipeds: doubly labelled water gets the seal of approval. Funct. Ecol. 22, 245-254.
- Speakman, J.R., 1987. Calculation of CO<sub>2</sub> production in doubly-labeled water studies. J. Theor. Biol. 126, 101-104.
- Speakman, J.R., 1993. How should we calculate CO<sub>2</sub> production in doubly labeled water studies of animals? Funct. Ecol. 7, 746-750.
- Speakman, J.R., 1997. Doubly labelled water: theory and practice. London.
- Speakman, J.R., 2005. The role of technology in the past and future development of the doubly labelled water method. Isotopes Environ Health Stud 41, 335-343.

- Speakman, J.R., Krol, B., 2005. Comparison of different approaches for the calculation of energy expenditure using doubly labeled water in a small mammal. Physiol. Biochem. Zool. 78, 650-667.
- Speakman, J.R., Nair, K.S., Goran, M.I., 1993. Revised equations for calculating CO<sub>2</sub> production from doubly labeled water in humans. Am. J. Physiol. 264, E912-917.
- Speakman, J.R., Racey, P.A., 1987. The equilibrium concentration of O<sup>18</sup> in body water -Implications for the accuracy of the doubly-labeled water technique and a potential new method of measuring RQ in free-living animals. J. Theor. Biol. 127, 79-95.
- Staniland, I.J., 2002. Investigating the biases in the use of hard prey remains to identify diet composition using Antarctic fur seals (*Arctocephalus gazella*) in captive feeding trials. Mar. Mamm. Sci. 18, 223-243.
- Staniland, I.J., Boyd, I.L., Reid, K., 2007. An energy-distance trade-off in a central-place forager, the Antarctic fur seal (*Arctocephalus gazella*). Mar. Biol. 152, 233-241.
- Stephens, D.W., Krebs, J.R., 1986. Foraging theory. Princeton University Press, Princeton, New Jersey.
- Suzuki, I., Naito, Y., Folkow, L.P., Miyazaki, N., Blix, A.S., 2009. Validation of a device for accurate timing of feeding events in marine animals. Polar Biol. 32, 667-671.
- Thomas, A.C., Deagle, B.E., Eveson, P.J., Wilson, C.H., Trites, A.W., In prep. Quantitative DNA metabarcoding: improved estimates of species proportional biomass using correction factors derived from control material.
- Thomas, A.C., Jarman, S.N., Haman, K.H., Trites, A.W., Deagle, B.E., 2014. Improving accuracy of DNA diet estimates using food tissue control materials and an evaluation of proxies for digestion bias. Mol. Ecol. 23, 3706-3718.
- Thorarensen, H., Gallaugher, P.E., Farrell, A.P., 1996. The limitations of heart rate as a predictor of metabolic rate in fish. J. Fish Biol. 49, 226-236.
- Tierney, M., Hindell, M.A., Goldworthy, S., 2002. Energy content of mesopelagic fish from Macquarie Island. Antarct. Sci. 14, 225-230.

- Tollit, D.J., M. Wong, A.J. Winship, D.A.S. Rosen, Trites, A.W., 2003. Quantifying errors associated with using prey skeletal structures from fecal samples to determine the diet of Steller's sea lion (*Eumetopias jubatus*). Mar. Mamm. Sci. 19, 724-744.
- Tollit, D.J., S. Heaslip, T. Zepplelin, R. Joy, Call, K., Trites, A.W., 2004. A method to improve size estimates of walleye pollock (*Theragra chalcogramma*) and Atka mackerel (*Pleurogrammus monopterygius*) consumed by pinnipeds: digestion correction factors applied to bones and otoliths recovered in scats. Fish. Bull. 102, 498-508.
- Tollit, D.J., Schulze, A.D., Trites, A.W., Olesiuk, P.F., Crockford, S.J., Gelatt, T.S., Ream, R.R., Miller, K.M., 2009. Development and application of DNA techniques for validating and improving pinniped diet estimates. Ecol. Appl. 19, 889-905.
- Tollit, D.J., Steward, M.J., Thompson, P.M., Pierce, G.J., Santos, M.B., Hughes, S., 1997. Species and size differences in the digestion of otoliths and beaks: implications for estimates of pinniped diet composition. Can. J. Fish. Aquat. Sci. 54, 105-119.
- Towell, R.G., Ream, R., 2011. 2011 northern fur seal pup production estimate on Bogoslof Island, Alaska, In: National Marine Mammal Laboratory Report. (Ed.), Seattle, WA.
- Towell, R.G., Ream, R.R., York, A.E., 2006. Decline in northern fur seal (*Callorhinus ursinus*) pup production on the Pribilof Islands. Mar. Mamm. Sci. 22, 486-491.
- Tremblay, Y., Cherel, Y., 2000. Benthic and pelagic dives: a new foraging behaviour in rockhopper penguins. Mar. Ecol. Prog. Ser. 204, 257-267.
- Tremblay, Y., Shaffer, S.A., Fowler, S.L., Kuhn, C.E., McDonald, B.I., Weise, M.J., Bost, C.-A., Weimerskirch, H., Crocker, D.E., Goebel, M.E., Costa, D.P., 2006.
  Interpolation of animal tracking data in a fluid environment. J. Exp. Biol. 209, 128-140.
- Trillmich, F., 1990. The behavioral ecology of maternal effort in fur seals and sea lions. Behaviour 114, 3-20.

- Trillmich, F., Lechner, E., 1986. Milk of the Galapagos fur seal and sea lion, with a comparison of the milk of eared seals (*Otariidae*). Journal of Zoology, London 209, 271-277.
- Trillmich, F., Ono, K. (Eds.), 1991. Pinnipeds and El Niño: responses to environmental stress. Springer-Verlag, New York.
- Trites, A.W., 1989. Estimating the juvenile survival rate of male northern fur seals (*Callorhinus ursinus*). Can. J. Fish. Aquat. Sci. 46, 1428-1436.
- Trites, A.W., 1992. Reproductive synchrony and the estimation of mean date of birth from daily counts of northern fur seal pups. Mar. Mamm. Sci. 8, 44-56.
- Trites, A.W., 1993. Biased estimates of fur seal pup mass: origins and implications. Journal of Zoology, London 229, 515-525.
- Trites, A.W., Battaile, B.C., Benoit-Bird, K.J., Cooper, L.W., Grebmeier, J.M., Harding,
  A.N.A., Heppell, S.A., Irons, D.B., Jay, C.V., Kitaysky, A.S., Kuletz, K.J., Paredes,
  R., Renner, H.M., Roby, D.D., 2015. Patch Dynamics Study. NPRB BSIERP Projects
  B67 and B77 Final Report, North Pacific Research Board, p. 726 p.
- Trites, A.W., Bigg, M.A., 1996. Physical growth of northern fur seals (*Callorhinus ursinus*): seasonal fluctuations and migratory influences. J. Zool. 238, 459-482.
- Trites, A.W., Calkins, D.G., 2008. Diets of mature male and female Steller sea lions (*Eumetopias jubatus*) differ and cannot be used as proxies for each other. Aquatic Mammals 34, 25-34.
- Trites, A.W., Donnelly, C.P., 2003. The decline of Steller sea lions *Eumetopias jubatus* in Alaska: a review of the nutritional stress hypothesis. Mammal. Rev. 33, 3-28.
- Trites, A.W., Miller, A.J., Maschner, H.D.G., Alexander, M.A., Bograd, S.J., Calder, J.A., Capotondi, A., Coyle, K.O., Di Lorenzo, E., Finney, B.P., Gregr, E.J., Grosch, C.E., Hare, S.R., Hunt, G.L., Jahncke, J., Kachel, N.B., Kim, H.J., Ladd, C., Mantua, N.J., Marzban, C., Maslowski, W., Mendelssohn, R., Neilson, D.J., Okkonen, S.R., Overland, J.E., Reedy-Maschner, K.L., Royer, T.C., Schwing, F.B., Wang, J.X.L., Winship, A.J., 2007. Bottom-up forcing and the decline of Steller sea lions

(*Eumetopias jubatus*) in Alaska: assessing the ocean climate hypothesis. Fish. Oceanog. 16, 46-67.

- Van Pelt, T.I., Piatt, J.F., Lance, B.K., Roby, D.D., 1997. Proximate composition and energy density of some North Pacific forage fishes. Comparative Biochemistry and Physiology A118, 1393-1398.
- Visser, H.G., Schekkerman, H., 1999. Validation of the doubly labeled water method in growing precocial birds: the importance of assumptions concerning evaporative water loss. Physiol. Biochem. Zool. 72, 740-749.
- Viviant, M., Trites, A.W., Rosen, D.A.S., Monestiez, P., Guinet, C., 2010. Prey capture attempts can be detected in Steller sea lions and other marine predators using accelerometers. Polar Biol. 33, 713-719.
- Vollenweider, J.J., Heintz, R.A., Schaufler, L., Bradshaw, R., 2011. Seasonal cycles in whole-body proximate composition and energy content of forage fish vary with water depth. Mar. Biol. 158, 413-427.
- Volpov, B.L., Hoskins, A.J., Battaile, B., Viviant, M., Wheatley, K.E., Marshall, G.J., Abernathy, K., Arnould, J.P.Y., 2015. Identification of prey captures in Australian fur seals (*Arctocephalus pusillus doriferus*) using head-mounted accelerometers: Field validation with animal-borne video cameras. PLOS One. 10, e0128789
- von Bertalanffy, L., 1938. A quantitative theory of organic growth (inquiries on growth laws. II). Hum. Biol. 10, 181-213.
- Wada, E., Mizutani, H., Minagawa, M., 1991. The use of stable isotopes for food web analysis. Crit. Rev. Food. Sci. Nutr. 30, 363-371.
- Waite, J.N., Trumble, S.J., Burkanov, V.N., Andrews, R.D., 2012. Resource partitioning by sympatric Steller sea lions and northern fur seals as revealed by biochemical dietary analyses and satellite telemetry. J. Exp. Mar. Biol. Ecol. 416–417, 41-54.
- Walther, G.-R., Post, E., Convey, P., Menzel, A., Parmesan, C., Beebee, T.J.C., Fromentin, J.-M., Hoegh-Guldberg, O., Bairlein, F., 2002. Ecological responses to recent climate change. Nature 416, 389-395.

- Waluda, C., Trathan, P., Rodhouse, P., 2004. Synchronicity in southern hemisphere squid stocks and the influence of the Southern Oscillation and Trans Polar Index. Fish. Oceanog. 13, 255-266.
- Wanless, S., Harris, M.P., Redman, P., Speakman, J.R., 2005. Low energy values of fish as a probable cause of a major seabird breeding failure in the North Sea. Mar. Ecol. Prog. Ser. 294, 1-8.
- Ward, S., Bishop, C.M., Woakes, A.J., Butler, P.J., 2002. Heart rate and the rate of oxygen consumption of flying and walking barnacle geese (*Branta leucopsis*) and bar-headed geese (*Anser indicus*). J. Exp. Biol. 205, 3347-3356.
- Watanabe, Y., Baranov, E.A., Sato, K., Naito, Y., Miyazaki, N., 2006. Body density affects stroke patterns in Baikal seals. J. Exp. Biol. 209, 3269-3280.
- Watanabe, Y.Y., Takahashi, A., 2013. Linking animal-borne video to accelerometers reveals prey capture variability. Proc. Natl. Acad. Sci. 110, 2199-2204.
- Watson, R.T., Zinyowera, M.C., Moss, R.H., Intergovernmental Panel on Climate Change.Working Group II., 1998. The regional impacts of climate change : an assessment of vulnerability. Cambridge University Press, Cambridge, UK ; New York, NY.
- Weimerskirch, H., Ancel, A., Caloin, M., Zahariev, A., Spagiari, J., Kersten, M., Chastel, O., 2003a. Foraging efficiency and adjustment of energy expenditure in a pelagic seabird provisioning its chick. J. Anim. Ecol. 72, 500-508.
- Weimerskirch, H., Gault, A., Cherel, Y., 2005. Prey distribution and patchiness: factors in foraging success and efficiency of wandering albatrosses. Ecology 86, 2611-2622.
- Weimerskirch, H., Inchausti, P., Guinet, C., Barbraud, C., 2003b. Trends in bird and seal populations as indicators of a system shift in the Southern Ocean. Antarct. Sci. 15, 249-256.
- Whitman, L.D., 2010. Variation in the energy density of forage fishes and invertebrates from the Southeastern Bering Sea, MSc. Thesis, Department of Fisheries and Wildlife, Oregon State University, Corvallis, p. 75pp.

- Wickens, P., York, A.E., 1997. Comparative population dynamics of fur seals. Mar. Mamm. Sci. 13, 241-292.
- Williams, T.M., 1999. The evolution of cost efficient swimming in marine mammals: limits to energetic optimization. Phil. Trans. R. Soc. Lond. B 354, 193-201.
- Williams, T.M., Davis, R.W., Fuiman, L.A., Francis, J., Le Boeuf, B.L., Horning, M., Calambokidis, J., Croll, D.A., 2000. Sink or swim: Strategies for cost-efficient diving by marine mammals. Science 288, 133-136.
- Williams, T.M., Fuiman, L.A., Horning, M., Davis, R.W., 2004. The cost of foraging by a marine predator, the Weddell seal *Leptonychotes weddellii*: pricing by the stroke. J. Exp. Biol. 207, 973-982.
- Wilson, R.P., Grant, W.S., Duffy, D.C., 1986. Recording devices on free-ranging marine animals: does measurement affect foraging performance? Ecology 67, 1091-1093.
- Wilson, R.P., Shepard, E.L.C., Laich, A.G., Frere, E., Quintana, F., 2010. Pedalling downhill and freewheeling up; a penguin perspective on foraging. Aquatic Biol. 8, 193-202.
- Wilson, R.P., White, C.R., Quintana, F., Halsey, L.G., Liebsch, N., Martin, G.R., Butler, P.J., 2006. Moving towards acceleration for estimates of activity-specific metabolic rate in free-living animals: the case of the cormorant. J. Anim. Ecol. 75, 1081-1090.
- Wilson, S.M., Hinch, S.G., Eliason, E.J., Farrell, A.P., Cooke, S.J., 2013. Calibrating acoustic acceleration transmitters for estimating energy use by wild adult Pacific salmon. Comp. Biochem. Physiol. A. 164, 491-498.
- Winship, A.J., Trites, A.W., Rosen, D.A.S., 2002. A bioenergetic model for estimating the food requirements of Steller sea lions (*Eumetopias jubatus*) in Alaska, USA. Mar. Ecol. Prog. Ser. 229, 291-312.
- Wolff, G.A., 1984. Identification and estimation of size from beaks of 18 species of cephalopods from the Pacific Ocean., In: NOAA Technical Report NMFS 17 (Ed.), National Marine Fisheries Services, p. 49pp.

- Wright, S., Metcalfe, J.D., Hetherington, S., Wilson, R., 2014. Estimating activity-specific energy expenditure in a teleost fish, using accelerometer loggers. Mar. Ecol. Prog. Ser. 496, 19-32.
- Wynen, L.P., Goldsworthy, S.D., Guinet, C., Bester, M.N., Boyd, I.L., Gjertz, I., Hofmeyr, G.J.G., White, R.W.G., Slade, R., 2000. Postsealing genetic variation and population structure of two species of fur seal (*Arctocephalus gazella* and *A. tropicalis*). Mol. Ecol. 9, 299-314.
- Ydenberg, R.C., Welham, C.V.J., Schmid-Hempel, R., Schmid-Hempel, P., Beauchamp, G., 1992. Time and energy constraints and the relationships between currencies in foraging theory. Behav. Ecol. 5, 28-34.
- Ydesen, K.S., Wisniewska, D.M., Hansen, J.D., Beedholm, K., Johnson, M.L., Madsen, P.T., 2014. What a jerk: prey engulfment revealed by high-rate, super-cranial accelerometry on a harbour seal (*Phoca vitulina*). J. Exp. Biol. 217, 2239-2243.
- Yoda, K., Naito, Y., Sato, K., Takahashi, A., Nishikawa, J., Ropert-Coudert, Y., Kurita, M., Le Maho, Y., 2001. A new technique for monitoring the behaviour of free-ranging Adelie penguins. J. Exp. Biol. 204, 685-690.
- York, A.E., 1985. Juvenile survival of fur seals, In: Kozloff, P. (Ed.), Fur seal investigations, 1982, NOAA Technical Report, US Department of Commerce, pp. 34-47.
- Young, B.L., Rosen, D.A.S., Hindle, A.G., Haulena, M., Trites, A.W., 2011. Dive behaviour impacts the ability of heart rate to predict oxygen consumption in Steller sea lions (*Eumetopias jubatus*) foraging at depth. J. Exp. Biol. 214, 2267-2275.
- Zeppelin, T.K., Orr, A.J., 2010. Stable isotope and scat analyses indicate diet and habitat partitioning in northern fur seals *Callorhinus ursinus* across the eastern Pacific. Mar. Ecol. Prog. Ser. 409, 241-253.
- Zeppelin, T.K., Ream, R.R., 2006. Foraging habitats based on the diet of female northern fur seals (*Callorhinus ursinus*) on the Pribilof Islands, Alaska. J. Zool. 270, 565-576.
- Zeppelin, T.K., Tollit, D.J., Call, K.A., Orchard, T.J., Gudmundson, C.J., 2004. Sizes of walleye pollock (*Theragra chalcogramma*) and Atka mackerel (*Pleurogrammus*)

*monopterygius*) consumed by the western stock of Steller sea lions (*Eumetopias jubatus*) in Alaska from 1998 to 2000. Fish. Bull. 102, 509-521.

Zhao, L., Schell, D.M., Castellini, M.A., 2006. Dietary macronutrients influence <sup>13</sup>C and <sup>15</sup>N signatures of pinnipeds: captive feeding studies with harbor seals (*Phoca vitulina*). Comp. Biochem. Physiol. A. 143, 469-478.

## Appendices

	Equation	Parameter	Estimate (MJ/d)	SE	P value	$\mathbf{R}^2$	AIC	Estimate (MJ/d)	SE	P value	$\mathbf{R}^2$	AIC
EE at sea (MJ)		Foraging	31.51	5.82	<2.10 <sup>-5</sup>	0.70	266.1	31.44	5.66	<2.10 <sup>-5</sup>	0.70	264.1
	Speakman	Transiting	19.73	6.84	0.009			19.58	6.52	0.006		
	1993	Surf. Mov.	15.09	8.01	1.887			14.63	6.22	0.028		
		Resting	-2.97	30.90	-0.096							
		Foraging	34.50	6.25	<2.10 <sup>-5</sup>	0.70	269.8	34.46	6.08	<10 <sup>-5</sup>	0.70	267.8
	Speakman	Transiting	23.67	7.34	0.003			23.57	7.00	0.003		
	1997	Surf. Mov.	16.00	8.58	0.075			15.68	6.68	0.028		
		Resting	-1.64	33.15	0.949							
	Coward	Foraging	27.92	6.85	0.001	0.61	274.6	27.95	6.67	<10 <sup>-3</sup>	0.61	272.6
		Transiting	20.33	8.06	0.019			20.39	7.68	0.014		
	et al. 1985	Surf. Mov.	11.13	9.42	0.250			11.33	7.33	0.136		
		Resting	1.22	36.39	0.973							
REE at sea (MJ/d)		Foraging	32.39	6.76	<10 <sup>-4</sup>	0.	165.8	32.68	6.44	<10 <sup>-4</sup>	0.	163.9
	Speakman	Transiting	21.05	7.93	0.015			21.31	7.64	0.010		
	1993	Surf. Mov.	10.55	7.53	0.175			11.09	6.81	0.117		
		Resting	4.62	24.51	0.852							
		Foraging	35.35	7.16	<10 <sup>-4</sup>	0.	168.8	35.73	6.83	<10 <sup>-4</sup>	0.	166.9
	Speakman	Transiting	24.79	8.40	0.007			25.14	8.10	0.005		
	1997	Surf. Mov.	11.64	7.98	0.159			12.35	7.22	0.101		
		Resting	6.10	25.97	0.816							
		Foraging	29.04	8.00	0.001	0.	174.7	29.51	7.63	0.001	0.	172.7
	Coward	Transiting	22.70	9.39	0.024			23.12	9.06	0.018		
	et al. 1985	Surf. Mov.	5.67	8.92	0.532			6.54	8.08	0.426		
		Resting	7.44	29.04	0.800							

Table A.1: Additional models of energy expenditure per time-activity budget for energy expenditure calculated with the 3 equations that yielded less accurate results than the equation from (Speakman et al., 1993).
Model/Eq.	Parameters	Parameters Estimates SE		р	R <sup>2</sup>	AICc
	Intercept	16.160	14.170	0.268	0.89	241.2
	Int.DBA <sub>Transit</sub>	-3.10 <sup>-7</sup>	7.10 <sup>-8</sup>	< 10 <sup>-4</sup>		
0 1 1	Dive.NB	-0.041	0.010	< 10 <sup>-3</sup>		
2.11	Time <sub>Dive</sub>	0.026	0.006	< 10 <sup>-3</sup>		
	Int.DBA <sub>Rest</sub>	-2.10-6	8.10-7	0.021		
	Total.Time	0.021	0.003	< 10 <sup>-6</sup>		
	Intercept	8.238	13.75	0.556	0.89	241.9
	Int.DBA <sub>Transit</sub>	-4.10-7	7.10 <sup>-8</sup>	< 10 <sup>-4</sup>		
2 1 2	Dive.NB	-0.041	0.010	< 10 <sup>-3</sup>		
2.12	Time <sub>Dive</sub>	0.047	0.006	< 10 <sup>-6</sup>		
	Time <sub>Tramsit</sub>	0.019	0.003	< 10 <sup>-4</sup>		
	Time <sub>Surf.mov</sub>	0.024	0.004	< 10 <sup>-5</sup>		
	Intercept	10.990	14.220	0.449	0.88	242.8
	Int.DBA <sub>Transit</sub>	-4.10 <sup>-7</sup>	7.10 <sup>-8</sup>	< 10 <sup>-4</sup>		
2.13	Dive.NB	-0.041	0.001	0.001		
	Time <sub>Dive</sub>	0.025	0.007	0.002		
	Time <sub>Rest.</sub>	-0.035	0.016	0.039		
	Total.Time	0.024	0.003	< 10 <sup>-6</sup>		
2 1 4	Intercept	87.030	26.690	0.004	0.88	243.1
2.14	Int.DBA <sub>Transit</sub>	-3.10 <sup>-7</sup>	7.10 <sup>-8</sup>	< 10 <sup>-4</sup>		
	Dive.Rate	-5.882	1.533	0.001		
	Time <sub>Dive</sub>	0.027	0.006	0.002		
	Int.DBA <sub>Rest</sub>	-2.10 <sup>-6</sup>	8.10-7	0.039		
	Total.Time	0.014	0.003	< 10 <sup>-4</sup>		
	Intercept	2.314	0.629	0.001	0.86	73.0
	Int.DBA <sub>Transit</sub> (/kg)	-3.10 <sup>-7</sup>	8.10-8	0.001		
2.15	Dive.rate	-0.178	0.044	< 10 <sup>-3</sup>		
	Time <sub>Dive</sub>	8. 10 <sup>-4</sup>	1. 10-4	< 10 <sup>-4</sup>		
	Int.DBA <sub>Total</sub> (/kg)	-2.10 <sup>-6</sup>	5.10-7	< 10 <sup>-3</sup>		

Table A.2: Details of the best models predicting energy expenditure (in MJ or in MJ/kg).

Table A.3: Frequency of occurrence (FO) and split-sample frequency of occurrence (SSFO) of different prey item in northern fur seals diets measured either by morphological identification of hard-part remains or by DNA sequences analyses in scats collected on Reef rookery (St. Paul Island, Bering Sea, Alaska) during the breeding season 2011.The mean percent of DNA sequences specific to each prey in the scats (Perc. seq.) is also indicated.

		Hard-parts		DNA			Combined	
		<b>_</b>				Perc.	Perc.	
-		FO	SSFO	FO	SSFO	seq.	FO	SSFO
Group	Species/taxon name	(%)	(%)	(%)	(%)	(%)	(%)	(%)
Gadid	Walleye pollock	75.00	62.75	96.94	51.11	70.42	96.94	46.14
	Pacific cod	8.33	7.00	18.37	4.70	2.99	22.45	6.06
	Pacific hake			21.43	5.08	2.54	21.43	4.89
Cephalopod	Unidentified squid/Octopus	20.00	12.28				12.24	2.97
	Boreopacific armhook squid			2.04	0.51	0.07	2.04	0.51
Salmon	All salmon	15.00	6.03	52.04	13.47	6.94	3.06	0.71
	Chum salmon			27.55	7.70	5.21	27.55	7.11
	Pink salmon			10.20	2.95	0.82	10.20	2.60
	Chinook salmon			7.14	1.40	0.51	7.14	1.37
	Coho salmon			5.10	0.92	0.33	5.10	0.92
	Sockeye salmon			2.04	0.49	0.07	2.04	0.49
Hexagrammid	Atka mackerel	10.00	2.92	5.10	0.78	1.15	8.16	1.63
	Kelp greenling	3.33	0.72	15.31	3.14	3.13	15.31	2.89
	Whitespotted greenling			1.02	0.26	0.03	1.02	0.20
	Lingcod			1.02	0.17	0.07	1.02	0.15
Mesopelagic	Northern smoothtongue	5.00	2.42	4.08	0.84	1.52	5.10	1.11
	Northern lampfish			1.02	0.15	0.02	1.02	0.13
	Northern lanternfish			1.02	0.15	0.02	1.02	0.13
Forage	Pacific herring			30.61	9.08	4.87	30.61	8.46
	Pacific sand lance	1.67	0.83	6.12	1.51	0.50	6.12	1.33
	Pacific sandfish	1.67	0.42	2.04	0.51	0.80	2.04	0.46
	Capelin	1.67	0.28				1.02	0.17
Flatfish	Pacific halibut			15.31	3.30	1.50	15.31	3.18
	Arrowtooth flounder	1.67	0.28	9.18	2.04	0.75	9.18	1.84
	Dover sole			1.02	0.17	0.07	1.02	0.15
Other	Rockfish	1.67	0.33				1.02	0.20
	Pacific staghorn sSculpin			5.10	1.17	0.53	5.10	1.17
	Shiner surfperch			6.12	1.02	0.56	6.12	0.97
	Crescent gunnel			2.04	0.71	1.51	2.04	0.71
	Sablefish			1.02	0.15	0.04	1.02	0.13
	Non-fish	3.33	2.50				2.04	0.51
	Worm	3.33	1.25				2.04	0.71

Sampling			Foraging					C/N
time	Tissue	ID	location	δ <sup>13</sup> C	$\delta^{15}N$	% <sup>13</sup> C	% <sup>15</sup> N	ratio
Before	Plasma	1	on-shelf	-18.73	16.46	45.80	12.40	3.69
Before	Plasma	5	on-shelf	-18.81	16.79	45.48	12.05	3.77
Before	Plasma	6	on-shelf	-19.69	14.23	46.14	12.21	3.78
Before	Plasma	7	on-shelf	-18.37	18.16	45.46	11.94	3.81
Before	Plasma	8	on-shelf	-18.94	17.35	46.06	12.03	3.83
Before	Plasma	9	on-shelf	-18.54	17.56	45.08	12.04	3.74
Before	Plasma	12	on-shelf	-19.37	16.56	43.23	11.37	3.80
Before	Plasma	13	on-shelf	-18.47	17.41	46.36	12.17	3.81
Before	Plasma	14	on-shelf	-19.38	14.96	46.18	12.40	3.73
Before	Plasma	17	on-shelf	-18.62	16.81	45.91	12.32	3.73
Before	Plasma	18	on-shelf	-18.33	17.52	43.87	12.19	3.60
Before	Plasma	19	on-shelf	-18.74	17.43	44.01	11.52	3.82
Before	Plasma	2	off-shelf	-19.69	13.89	45.33	12.15	3.73
Before	Plasma	3	off-shelf	-19.65	15.03	46.13	12.13	3.80
Before	Plasma	4	off-shelf	-19.54	14.60	45.97	12.07	3.81
Before	Plasma	10	off-shelf	-19.13	15.91	45.89	12.48	3.68
Before	Plasma	11	off-shelf	-19.78	14.90	44.60	11.43	3.90
Before	Plasma	15	off-shelf	-19.32	14.90	46.93	12.80	3.67
Before	Plasma	16	off-shelf	-19.44	15.05	47.06	12.78	3.68
Before	Plasma	20	off-shelf	-19.56	15.13	44.15	11.58	3.81
Before	RBC	1	on-shelf	-19.34	15.36	56.92	17.05	3.34
Before	RBC	5	on-shelf	-19.33	15.11	50.64	15.09	3.36
Before	RBC	6	on-shelf	-19.94	12.08	50.73	15.24	3.33
Before	RBC	7	on-shelf	-18.78	16.25	48.03	14.11	3.40
Before	RBC	8	on-shelf	-19.16	15.73	50.41	15.06	3.35
Before	RBC	9	on-shelf	-18.92	15.34	50.58	15.10	3.35
Before	RBC	12	on-shelf	-19.54	14.38	52.00	15.56	3.34
Before	RBC	13	on-shelf	-18.88	15.89	51.15	15.23	3.36
Before	RBC	14	on-shelf	-19.70	14.87	50.18	14.93	3.36
Before	RBC	17	on-shelf	-19.24	13.77	49.91	14.95	3.34
Before	RBC	18	on-shelf	-19.12	16.01	49.86	14.86	3.36
Before	RBC	19	on-shelf	-19.52	14.95	47.12	13.98	3.37
Before	RBC	2	off-shelf	-20.01	11.70	50.90	15.12	3.37
Before	RBC	3	off-shelf	-19.60	14.46	51.45	15.32	3.36
Before	RBC	4	off-shelf	-19.90	12.47	51.10	15.23	3.36
Before	RBC	10	off-shelf	-19.70	15.55	50.17	15.01	3.34
Before	RBC	11	off-shelf	-19.89	12.90	50.99	15.05	3.39

Table A.4: Stable carbon and nitrogen isotope values in plasma and red blood cells of 20 lactating northern fur seals breeding on St. Paul Island, Bering Sea.Blood samples were taken either before or after a single foraging trip at sea.

Sampling			Foraging					C/N
time	Tissue	ID	location	δ <sup>13</sup> C	$\delta^{15}N$	% <sup>13</sup> C	$\%^{15}$ N	ratio
Before	RBC	15	off-shelf	-19.28	15.10	49.96	14.92	3.35
Before	RBC	16	off-shelf	-19.46	14.95	50.06	14.98	3.34
Before	RBC	20	off-shelf	-19.53	15.11	46.20	13.69	3.37
After	Plasma	1	on-shelf	-18.85	16.49	45.53	12.24	3.72
After	Plasma	5	on-shelf	-18.55	16.85	46.16	12.44	3.71
After	Plasma	6	on-shelf	-19.72	15.04	47.61	12.43	3.83
After	Plasma	7	on-shelf	-18.20	18.01	46.17	12.13	3.81
After	Plasma	8	on-shelf	-18.85	17.78	45.47	11.84	3.84
After	Plasma	9	on-shelf	-19.19	17.08	45.20	11.95	3.78
After	Plasma	12	on-shelf	-19.02	16.89	47.08	12.20	3.86
After	Plasma	13	on-shelf	-18.74	17.62	43.22	11.36	3.80
After	Plasma	14	on-shelf	-19.00	16.76	45.51	12.18	3.74
After	Plasma	17	on-shelf	-18.82	17.00	43.67	11.28	3.87
After	Plasma	18	on-shelf	-18.32	18.58	45.38	12.16	3.73
After	Plasma	19	on-shelf	-18.93	17.71	44.20	11.30	3.91
After	Plasma	2	off-shelf	-19.79	14.04	46.04	12.01	3.83
After	Plasma	3	off-shelf	-19.60	15.17	38.27	10.00	3.83
After	Plasma	4	off-shelf	-19.58	14.48	43.69	11.49	3.80
After	Plasma	10	off-shelf	-19.63	14.98	47.07	12.61	3.73
After	Plasma	11	off-shelf	-19.64	14.76	46.90	12.33	3.80
After	Plasma	15	off-shelf	-19.49	14.58	47.60	12.70	3.75
After	Plasma	16	off-shelf	-19.28	15.58	45.29	11.92	3.80
After	Plasma	20	off-shelf	-19.51	15.33	44.15	11.72	3.77

Table A.5 Relative proportion (%), average prey mass (in g), prey energy density (ED in kJ/g), energy content (in kJ) of prey species in the diets of female northern fur seal breeding on St. Paul, and of females breeding on Bogoslof Island. Percent in diet (% SSFO) were obtained from morphological identification of hard part remains in our study and from collection of samples of Bogoslof Island rookeries in 2009 (Trites et al. 2015), and mass was calculated from size of hard part remains.

	Prey group	Prey species	% in diet	Mass (g)	ED (kJ/g)	EC (kJ)
	Gadid	Pollock	62.75	16.48 ± 1.11	$3.57 \pm 0.30$	$58.59 \pm 0.20$
		Pacific cod	7	$6.76 \pm 1.14$	$2.94 \pm 0.12$	$19.72 \pm 0.11$
	Caphalopod	Cephalopod	12.28	$3.74 \pm 0.29$	$4.76 \pm 0.11$	$17.79 \pm 0.04$
	Salmon	Salmon	6.03	$797.78 \pm 53.65$	$5.53 \pm 0.30$	$4425.00\pm9.80$
	Hexagrammid	Atka mackerel	2.92	$18.69 \pm 2.37$	$4.02\pm0.08$	$75.19 \pm 0.29$
FS		Kelp Greenling	0.72	NA	3.45	NA
LF N	Mesopelagic	Northern smoothtongue	2.42	$3.87 \pm 0.25$	$5.67 \pm 0.25$	$21.91 \pm 0.05$
HE	Forage	Sand lance	0.83	$12.76\pm0.05$	$5.06 \pm 0.12$	$64.55 \pm 0.05$
S-Z		Sandfish	0.42	$8.01 \pm 1.04$	$3.55 \pm 0.13$	$28.52 \pm 0.12$
0		Capelin	0.28	$8.49 \pm 0.31$	$4.35 \pm 0.35$	$36.95 \pm 0.11$
	Flatfish	Arrowtooth flounder	0.28	$3.78 \pm 0.01$	$5.14 \pm 0.75$	$19.45 \pm 0.08$
	Other	Rockfish sp.	0.33	$3.78 \pm 0.06$	2.97	NA
		Non-fish	2.5	NA	NA	NA
		Worm	1.25	NA	NA	NA
	Gadid	Pollock	0.86	$16.48 \pm 11.45$	$3.57\pm0.30$	$58.59 \pm 0.20$
	Caphalopod	Octopus sp.	42.70	$3.74\pm0.29$	$4.76 \pm 0.11$	$17.79 \pm 0.04$
	Salmon	Salmon	3.87	$797.78 \pm 8.92$	$5.53 \pm 0.30$	$4425.00\pm9.80$
	Hexagrammid	Atka mackerel	0.60	$18.69 \pm 2.37$	$4.02\pm0.08$	$75.19 \pm 0.29$
E	Mesopelagic	Northern lampfish	2.75	$2.02 \pm 1.01$	$8.98 \pm 1.74$	$17.84 \pm 0.30$
ELF		Northern smoothtongue	46.22	$3.87 \pm 0.25$	$5.67 \pm 0.25$	$21.91 \pm 0.05$
HS	Forage	Pacific herring	0.52	$23.49 \pm 5.75$	$6.3 \pm 0.22$	$147.76 \pm 1.14$
E E		Sand lance	1.03	$12.76\pm0.05$	$5.06 \pm 0.12$	$64.55 \pm 0.05$
0	Other	Sablefish	0.86	NA	NA	NA
		Polychaete unident.	0.34	NA	NA	NA
		Threespine stickleback	0.26	NA	NA	NA

## Equations to estimate body length and mass of fish and squid

The following equations used to convert the lengths of otolith and squid beaks into body length and mass for nine species consumed by northern fur seals:

*Walleye pollock*.  $FL = 0.50 \text{ OL}^2 + 15.74 \text{ OL} + 13.3$  (Zeppelin et al., 2004) and BM = 0.0077 ×  $FL^{2.906}$  (Frost and Lowry, 1981). Walleye pollock were separated into age classes using the cutting length of 100mm below which they were considered of age-0+ and above which they were of age-1+ (Whitman, 2010; Honkalehto et al., 2012).

**Salmon.** BM =  $0.0103 \times \text{FL}^{3.092}$  (Harvey et al., 2000) . *Simulations*:  $31.75 \pm 0.14$  cm and 797.78 ± 8.92 g (from 55% of fish between 16-24 cm, and 45% between 35 - 59cm).

*Atka mackerel*.  $FL = 8.40 \text{ OL} - 4.99 \text{ and } BM = 0.0034 \times FL^{3.401}$  (Harvey et al., 2000).

*Northern smoothtongue*.\_BM =  $0.0106 \times FL^{2.85}$  (Orlov and Binohlan, 2009).

*Capelin*.  $FL = 3.45 \text{ OL} + 3.62 \text{ and } BM = 0.0054 \times FL^{3.160}$  (Harvey et al., 2000).

**Pacific herring**. FL =  $5.24 \text{ OL} - 1.85 \text{ and BM} = 0.0044 \times \text{FL}^{3.398}$  (Harvey et al., 2000).

*Arrowtooth flounder*.  $FL = 4.75 \text{ OL} - 2.96 \text{ and } BM = 0.0093 \times FL^{2.999}$  (Harvey et al., 2000). *Simulations*: 6.00 ± 0.005 cm and 3.78 ± 0.006g (from 100% of fish between 5 - 7 cm)

**Pacific sand lance.**  $FL = 4.06 \text{ OL} - 2.01 \text{ and } BM = 0.0063 \times FL^{2.790}$  (Harvey et al., 2000). Simulations:  $15.00 \pm 0.02$  cm and  $12.76 \pm 0.05g$  (from 100% fish between 11 - 19 cm).

*Squid*. Measures of lower beak rostrum length (LRL) were converted into squid mass using the equations  $\ln(BM) = 2.52 + 1.99 \times \ln(LRL)$  (Clarke, 1962).